



## **Swansea University E-Theses**

## Investigation of oily wastewater treatment processes.

Okam, Ifeyinwa Charlotte

How to cite:

Okam, Ifeyinwa Charlotte (2008) *Investigation of oily wastewater treatment processes.*. thesis, Swansea University. http://cronfa.swan.ac.uk/Record/cronfa42786

#### Use policy:

This item is brought to you by Swansea University. Any person downloading material is agreeing to abide by the terms of the repository licence: copies of full text items may be used or reproduced in any format or medium, without prior permission for personal research or study, educational or non-commercial purposes only. The copyright for any work remains with the original author unless otherwise specified. The full-text must not be sold in any format or medium without the formal permission of the copyright holder. Permission for multiple reproductions should be obtained from the original author.

Authors are personally responsible for adhering to copyright and publisher restrictions when uploading content to the repository.

Please link to the metadata record in the Swansea University repository, Cronfa (link given in the citation reference above.)

http://www.swansea.ac.uk/library/researchsupport/ris-support/



by

Ifeyinwa Charlotte Okam April 2008

Thesis submitted to the University of Wales in fulfilment of the requirements

for the

**Degree of Master of Philosophy** 

Department of Chemical Engineering School of Engineering

> Swansea University, Singleton Park, Swansea SA2 8PP



ProQuest Number: 10807562

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10807562

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

### Acknowledgments

First of all, I would like to thank and glorify the Almighty God. God of Abraham, God of Jacob, and God of Isaac. I thank Him for taking care and keeping my family and me during the course of my research. I thank Him for each time my faith was tested - He also showed He is a faithful God. I thank Him for our Lord Jesus Christ and for his word which encouraged me all the way.

My thanks also go to Dr R.W. Lovitt, my supervisor. I thank him for always being approachable and for his support and patience during the course of my study. I am glad you were my supervisor.

I would also like to thank my loving husband for his support during my research even to the extent of getting me a recorder so I could record conversations with my supervisor and listen to them again. I thank him also for the times I thought I would never get to the end of this research. I thank him for his faith in me and also for all the materials and help he rendered during this study.

My thanks go to my colleagues in the PhD office, all the help with printers and sometimes books and cards too. Special thanks to Kanika, thank God you were there with me.

I would also like to thank those I have met here in Swansea. It is lovely meeting people from different backgrounds, it makes it more colourful. Thank you also for one thing I must have learnt from our associations. I am confident those things have helped pattern my life.

I also like to thank my friends who were on my back for me to get my work finished. I thank you because your pestering ensured I must finish.

I would also like to thank the staff of University of Wales, Swansea. They are always so helpful, always going the extra mile.

Last but not the least, I would like to thank my family: my parents, my brothers and sisters, nephews and nieces, uncles and aunts, cousins and all my relations. Their prayers, love, and support went a long way for me. I thank God for them all. They are all a blessing to me.

## Dedication

## In loving memory of my great mother, Mrs Livinah Okam. I love you mama. May 29, 2007

This work is dedicated to those who have dreams and who hear a voice which says they can't achieve what they want to but they go on to achieve their dreams despite all odds.

## **Table of Contents**

Table of Co	ntents	i
List of Tabl	es	iii
List of Figu	res	iv
Summary o	f Abbreviations	. vii
Nomenclatu	Ire	viii
Abstract		xi
Chapter 1:	Introduction	1
1.1	INTRODUCTION	1
1.2	BACKGROUND	1
1.3	THE RESEARCH PROBLEM	2
1.4	RESEARCH METHODOLOGY	3
1.5	JUSTIFICATION FOR RESEARCH	4
1.6	RESEARCH PRESENTATION	4
1.7	SCOPE OF THE RESEARCH AND KEY ASSUMPTIONS	5
Chapter 2:	Literature Review	6
2.1	INTRODUCTION	6
2.2	<b>RESEARCH JUSTIFICATION</b>	6
2.2.1	Sources of oily wastewater and quantity	6
2.2.2	Review of only wastewater separation and treatment units Biological reactors and the Immobilized Call Biogeneter (ICB)	12
2.2.3	Aims/Objectives	12
2.3	THEORETICAL ASPECTS USED IN THE RESEARCH	17
2.3.1	Oily wastewater	17
2.3.2	General review of Immobilized Cell Bioreactors	19
2.3.3	Oil water separation design theory	25
2.3.4	General review of horizontal flow separators	28
2.3.5	Cost modelling	35
2.3.7	Theory of Process Integration	38
Chapter 3:	Materials, Equipment and Methods	39
3.1	INTRODUCTION	39
3.2	Materials	39
3.2.1	Oxygen probe and meter	40
3.2.2	Rotameter	41
3.3	EXPERIMENTAL SET-UP	41
3.4	Method	42
3.4.1	Determination of $k_{La}$	42
3.5	CALCULATION PROCEDURE FOR GENERATING OIL DROPLET DISTRIBUTION	43
3.5.1	Influent oil characteristics	43
3.5.2	Droplet size distribution	46
3.3.3	Calculation of oil volume	46

3.6 3.6.1	CALCULATION PROCEDURE FOR DESIGN OF GRAVITY SEPARATORS Predicting separator performance	47 49
3.7 3.7.1	CALCULATION PROCEDURE FOR DESIGN OF CPI Predicting CPI performance	51 53
3.8 3.8.1	BIOREACTOR DESIGN PROCEDURE Power requirements of the blower	55 55
Chapter 4:	Results and discussion - Aeration in various ICB models	58
4.1	INTRODUCTION	58
4.2	RESULTS	58
4.3	DISCUSSION	61
4.3.1	Comparing aeration in various ICB models	61
4.4	Conclusions	64
Chapter 5:	Results and Discussion – Optimizing Oil Recovery	65
5.1	INTRODUCTION	65
5.2	INTEGRATED PROCESS OF HORIZONTAL FLOW SEPARATOR AND ICB	65
5.2.1	Effect of varying influent oil concentration	66
5.2.2	Effect of varying separator horizontal velocity	71
5.2.3	Effect of varying separator volume	76
5.2.4	Effect of varying average droplet size	81
5.2.5	Effect of varying wastewater viscosity	80 01
527	Discussion of results and analysis	96
5.3	INTEGRATED THE ATMENT BROCESS LISING CPI AND ICB	107
531	Effect of varying intercentor plate width	107
5.3.2	Effect of varying interceptor plate width	112
5.3.3	Effect of varying interceptor plate spacing	117
5.3.4	Effect of varying interceptor plate length	122
5.3.5	Effect of varying number of plates in interceptor	126
5.3.6	Discussion of results and analysis	132
Chapter 6:	Conclusions and Recommendations	139
6.1	Conclusions	139
6.2	RECOMMENDATIONS	141
Bibliograph	ly	143

### Appendices

I Design and cost data for oil-water separators

### II Experimental results

- III Theoretical results: Integration of a horizontal flow separator and an ICB
- IV Theoretical results: Integration of a CPI and an ICB

# List of Tables

Table 2.1: Comparison of fixed film and membrane technologies (Schultz 2005)	13
Table 2.2: Oil-water system classification	17
Table 3.1: Physical data of plastic Pall rings	40
Table 3.2: Oil droplet size distribution	44
Table 5.1: Effect of separator size and droplet size on separator efficiency	104
Table 5.2: Effect of viscosity and droplet size on separator efficiency	104
Table 5.3: Effect of horizontal velocity and droplet size on separator efficiency	104
Table 5.4: Effect of various factors on interceptor volume	135
Table 5.5: Effect of various factors on plate interceptor efficiency	135

# List of Figures

Figure 1.1: Integrated system of gravity separation and ICB	3
Figure 2.1: Velocity components of an oil droplet in a flowing stream	26
Figure 2.2: Trajectory of an oil droplet in a moving stream	26
Figure 2.3: A typical horizontal flow gravity separator	29
Figure 2.4: A cross flow Corrugated Plate Interceptor	34
Figure 2.5: Effective (projected) area of an inclined plate	35
Figure 3.1: The experimental set-up for mass transfer measurements	42
Figure 3.2: Calculation procedure for generation of oil droplet distribution	45
Figure 3.3: Calculation procedure for design of gravity separators	48
Figure 3.4: Calculation procedure for oil droplet removal	50
Figure 3.5. Calculation procedure for design of plate interceptors	52
Figure 3.6: Calculation procedure for oil droplet removal in a CPI	54
Figure 3.7: Bioreactor design calculation procedure	56
Figure 3.8: Blower size calculation procedure	57
Figure 4.1. Oxygen MTC vs. aeration rate in ICB model with 16 mm Pall rings	59
Figure 4.2: Oxygen MTC vs. aeration rate in ICB model with 25 mm Pall rings	59
Figure 4.3: Oxygen MTC vs. aeration rate in ICB model with 38 mm Pall rings	60
Figure 4.4: Oxygen MTC vs. aeration rate in ICB model with 50 mm Pall rings	60
Figure 4.5: Internal structure of the 50 mm support particle	62
Figure 4.6: The various sizes of packing (mm): 50, 38, 25, 16 (from left to right)	63
Figure 5.1: Effect of oil concentration on cost of separation	67
Figure 5.2: Effect of oil concentration on oil recovery	67
Figure 5.3: Effect of oil concentration on revenue generated from recovered oil	68
Figure 5.4: Effect of oil concentration on separator efficiency	68
Figure 5.5: Effect of oil concentration on oxygen demand	69
Figure 5.6: Effect of oil concentration on bioreactor volume	69
Figure 5.7: Effect of oil concentration on cost of bioprocess	70
Figure 5.8: Effect of oil concentration on cost of integrated process	70
Figure 5.9: Effect of horizontal velocity on cost of separation	72
Figure 5.10: Effect of horizontal velocity on oil recovery	72
Figure 5.11: Effect of horizontal velocity on revenue generated from recovered oil	73
Figure 5.12: Effect of horizontal velocity on separator efficiency	73
Figure 5.12: Effect of horizontal velocity on separator efficiency	74
Figure 5.14: Effect of horizontal velocity on bioreactor volume	74
Figure 5.15: Effect of horizontal velocity on cost of hioprocess	75
Figure 5.16: Effect of horizontal velocity on cost of integrated process	75
Figure 5.17: Effect of separator volume on cost of separation	77
Figure 5.18: Effect of separator volume on oil recovery	77
Figure 5.10. Effect of separator volume on revenue generated from recovered oil	78
Figure 5.10: Effect of separator volume on separator efficiency	78
Figure 5.20. Effect of separator volume on ovvgen demand	70
Figure 5.22: Effect of constants volume on bioreastor volume	70
Figure 5.22. Effect of separator volume on cost of bioprocess	19 QA
Figure 5.2.2. Effect of separator volume on cost of integrated process	00 QA
Figure 5.25: Effect of average droplet size on cost of separation	00 27
Figure 5.26. Effect of average droplet size on cil recovery	02 20
rigure 5.20. Effect of average diopter size of on recovery	02

Figure 5.27: Effect of average droplet size on revenue generated from recovered oil	83
Figure 5.28: Effect of average droplet size on separator efficiency	83
Figure 5.29: Effect of average droplet size on oxygen demand	84
Figure 5.30: Effect of average droplet size on bioreactor volume	84
Figure 5.31: Effect of average droplet size on cost of bioprocess	85
Figure 5.32: Effect of average droplet size on cost of integrated process	85
Figure 5.33: Effect of wastewater viscosity on cost of separation	87
Figure 5.34: Effect of wastewater viscosity on oil recovery	87
Figure 5.35: Effect of wastewater viscosity on revenue generated from recovered oil	88
Figure 5.36: Effect of wastewater viscosity on separator efficiency	88
Figure 5.37: Effect of wastewater viscosity on oxygen demand	89
Figure 5.38: Effect of wastewater viscosity on bioreactor volume	89
Figure 5.39: Effect of wastewater viscosity on cost of bioprocess	90
Figure 5.40: Effect of wastewater viscosity on cost of integrated process	90
Figure 5.41: Effect of density difference on cost of separation	92
Figure 5.42: Effect of density difference on oil recovery	92
Figure 5.43: Effect of density difference on revenue generated from recovered oil	93
Figure 5 44: Effect of density difference on separator efficiency	93
Figure 5.45. Effect of density difference on oxygen demand	94
Figure 5.46: Effect of density difference on bioreactor volume	94
Figure 5.47: Effect of density difference on cost of bioprocess	95
Figure 5.48: Effect of density difference on cost of integrated process	95
Figure 5.49: Comparing variables in Stokes' Law	103
Figure 5.50: Effect of plate width on oil recovery	108
Figure 5.50: Effect of plate width on revenue generated from recovered oil	100
Figure 5.52: Effect of plate width on cost of separation	109
Figure 5.52: Effect of plate width on intercentor volume	110
Figure 5.55: Effect of plate width on interceptor efficiency	110
Figure 5.55: Effect of plate width on oxygen demand	111
Figure 5.56: Effect of plate width on bioreactor volume	111
Figure 5.50: Effect of plate width on cost of integrated process	112
Figure 5.57: Effect of plate angle on oil recovery	112
Figure 5.50: Effect of plate angle on revenue generated from recovered oil	117
Figure 5.60: Effect of plate angle on cost of separation	114
Figure 5.60. Effect of plate angle on intercenter volume	114
Figure 5.61. Effect of plate angle on interceptor officiency	115
Figure 5.62. Effect of plate angle on interceptor efficiency	115
Figure 5.64. Effect of plate angle on bigroaster volume	110
Figure 5.64. Effect of plate angle on oloreactor volume	110
Figure 5.65: Effect of plate angle on cost of integrated process	11/
Figure 5.60: Effect of plate spacing on intercovery	110
Figure 5.67. Effect of plate spacing on interceptor volume	119
Figure 5.68: Effect of plate spacing on cost of separation	119
Figure 5.69: Effect of plate spacing on interceptor efficiency	120
Figure 5.70: Effect of plate spacing on oxygen demand	120
Figure 5./1: Effect of plate spacing on bioreactor volume	121
Figure 5./2: Effect of plate spacing on cost of integrated process	121
Figure 5.73: Effect of plate length on oil recovery	123
Figure 5./4: Effect of plate length on revenue generated from recovered oil	123
Figure 5.75: Effect of plate length on interceptor volume	124

Figure 5.76: Effect of plate length on interceptor efficiency	124
Figure 5.77: Effect of plate length on oxygen demand	125
Figure 5.78: Effect of plate length on bioreactor volume	125
Figure 5.79: Effect of plate length on cost of integrated process	126
Figure 5.80: Effect of number of plates on oil recovery	128
Figure 5.81: Effect of number of plates on revenue generated from recovered oil	128
Figure 5.82: Effect of number of plates on interceptor volume	129
Figure 5.83: Effect of number of plates on interceptor efficiency	129
Figure 5.84: Effect of number of plates on cost of separation	130
Figure 5.85: Effect of number of plates on oxygen demand	130
Figure 5.86: Effect of number of plates on bioreactor volume	131
Figure 5.87: Effect of number of plates on cost of integrated process	131

ι.

## Summary of Abbreviations

API	American	Petroleum	Institute

- BOD Biological Oxygen Demand
- COD Chemical Oxygen Demand
- CPI Corrugated Plate Interceptor
- DO Dissolved Oxygen
- ICB Immobilized Cell Bioreactor
- MBR Membrane Bioreactor
- MTC Mass Transfer Coefficient
- O&M Operation and Maintenance
- PFCs Perfluorocarbons
- UF Ultrafiltration

## Nomenclature

- a Specific interfacial area  $(m^2/m^3)$
- A Interfacial area normal to mass transfer  $(m^2)$
- $A_P$  Projected plate area of a plate (m<sup>2</sup>)
- $A_s$  Surface area required for separation (m<sup>2</sup>)
- *BP* Brake Power (kW)
- C Concentration of oxygen in the bulk liquid phase (% saturation)
- $C_c$  Cost constant
- $C^*$  The saturation value of DO at the gas-liquid interface (% saturation)
- $C_e$  Purchased cost (£)
- $C_L$  Dissolved oxygen concentration in liquid (mg/l)
- $C_s$  Saturation dissolved oxygen concentration (mg/l)
- d Oil droplet diameter (m)
- D Diffusion coefficient of the oxygen in liquid (m<sup>2</sup>/s)
- $F_{\rm B}$  Buoyancy force in Newton (N)
- $F_d$  Drag force in Newton (N)
- $F_{dp}$  Design factor for corrugated plate interceptor (%)
- g Acceleration due to gravity  $(m/s^2)$
- $h_{\rm o}$  Height travelled by any droplet (m)
- $H_s$  Separator depth (m)
- *V<sub>ICB</sub>* Volume of Immobilized Cell Bioreactor (1)
- $k_L$  Mass transfer coefficient in the liquid phase (m/s)
- $k_{La}$  Volumetric liquid phase mass transfer coefficient (s<sup>-1</sup>)
- $l_{dp}$  Distance between plates (m)
- $L_P$  Plate length (m)

- $L_s$  Separator length (m)
- *m* Mass of oxygen (mg)
- *n* Number of droplets in a size range or interval
- $N_P$  Number of plates required to achieve separation
- OD Oxygen demand (mg/s)
- $O_R$  Overflow rate (velocity of rise of the waste stream) (m/s)
- OT Oxygen actually transferred/air volume (kg/m<sup>3</sup>)
- OTR Oxygen transfer rate (mg/l.s)
- *P* Blower discharge pressure (psig)
- Q Volumetric flow rate (m<sup>3</sup>/s)
- $Q_A$  Volumetric flow rate of air (m<sup>3</sup>/s)
- $R_1$  Ratio of horizontal flow velocity  $(v_h)$  to rise velocity of critical droplet  $(v_l)$
- $R_2$  Ratio of separator depth ( $H_s$ ) to width ( $w_s$ )
- s Average frequency at which any particular vertical element is mixed or rate of surface renewal (s<sup>-1</sup>)
- *S* Characteristic size parameter in units given in Appendix I
- $t_t$  Rise time of critical droplet (s)
- $t_w$  Residence time of oil water mixture in separator (s)
- $v_h$  Horizontal velocity (m/s)
- $V_{ICB}$  Volume of Immobilized Cell Bioreactor (m<sup>3</sup>)
- $v_o$  Terminal velocity of oil droplet or oil droplet velocity (m/s)
- $V_{oil}$  Volume of oil in a droplet (m<sup>3</sup>)
- $v_p$  Flow velocity along plates (m/s)
- $v_t$  Rise rate of the design droplet or critical droplet (m/s)
- $V_t$  Total volume of oil contributed by a droplet of a particular size (m<sup>3</sup>)
- $w_P$  Plate width (m)

- $w_s$  Separator width (m)
- x Droplet size of interest  $(\mu m)$
- $x_m$  Mean droplet size ( $\mu$ m)
- *X* Fraction of droplets of a particular size recovered
- Y Fraction of droplets of a particle size not recovered or captured
- $\theta$  The time of exposure of the liquid element (s)
- $\alpha$  Angle of inclination of the plates to the horizontal (degrees)
- $\delta$  Film thickness (m)
- $\mu$  Dynamic viscosity of the continuous phase (Pa.s)
- $\mu_{w}$  Water viscosity (Pa.s)
- $\eta$  Mechanical efficiency (%)
- $\rho_o$  Density of the oil (kg/m<sup>3</sup>)
- $\rho_w$  Density of water (kg/m<sup>3</sup>)
- $\sigma$  Standard deviation from the mean

## Abstract

With the ever-increasing price of oil and increasingly stringent environmental regulations, effective treatment of oily wastewater has become very important. Industry currently seeks to find cost-effective solutions to meeting discharge limits by integration of oil recovery units with biological treatment.

Two integrated processes were investigated so as to minimize the cost of treatment: (1) integration of a horizontal flow separator and an Immobilized Cell Bioreactor (ICB), and (2) integration of a Corrugated Plate Interceptor (CPI) and an ICB.

Theoretical models of each process were developed to investigate and compare the factors that affect oil recovery and the integrated process. These models were used to generate oil droplet size distributions, size and predict the performance of the separators, size the bioreactor, and estimate the cost of the integrated process. Experimental work was undertaken to investigate aeration in a laboratory-scale model of an ICB using support particles (Pall rings) of various sizes (16 mm, 25 mm, 38 mm, and 50 mm).

For the first process reducing the horizontal flow velocities best improved the separation efficiency and hence overall cost of treatment. A 33% change in droplet size resulted in an increase in separator efficiency of 55%, whilst a 33% change in horizontal velocity resulted in an increase of 68%. For the second process the effect of number of plates and plate dimensions, was investigated. For a 25% increase in design variable, the number of plates increased separator volume by 2%, plate length by 43% and plate width by 17%. Increasing the number of plates to improve interceptor efficiency resulted in the most cost-effective overall process.

The experimental analysis found that an intermediate size of support particle  $(1\frac{1}{2})$  resulted in the best volumetric mass transfer coefficient  $(k_{La})$ . Aeration rate in the ICB column was varied between 9 and 56 ml/s. Using 38 mm Pall rings,  $k_{La}$  increased from 0.0030 to 0.0097 s<sup>-1</sup>; using 50 mm Pall rings,  $k_{La}$  increased from 0.0015 to 0.0058 s<sup>-1</sup>.

This work concludes that enhancing oil recovery is a matter of separator design, rather than manipulation of wastewater characteristics as achieved by the use of chemicals. Effective aeration, and therefore size, of an ICB is dependent on the characteristics of the support particles used. The solution to cost-effective integration lies in the proper design of the units involved.

## Chapter 1: Introduction

## 1.1 Introduction

This chapter introduces the research and its primary motivations. It presents the research problem, justifies why the research is important, and presents the scope of the research.

### 1.2 Background

With the increasing value of oil and increasingly stringent environmental regulations, finding cost effective solutions for oily wastewater treatment has become very important.

An estimated US\$40 billion is spent worldwide in treatment of oily wastewater produced in the upstream oil and gas industry (<u>http://www.torrcanada.com</u>, 2007); the large capital outlay, increasing value of oil and environmental regulation necessitates the need for process integration in oily wastewater treatment. Recovered oil could be put back into the process, thereby reducing energy cost, or can be sold. Using the recovered oil for any of these purposes ultimately reduces the overall cost of treatment. Economic factors aside, the treated water needs to be suitable for discharge to the environment.

Several technologies have been developed for oily wastewater treatment such as horizontal flow separators, Corrugated Plate Interceptors (CPI), Dissolved Air Flotation (DAF) units, hydrocyclones, bioreactors, and the membrane processes, which are the most recent advancement in oily wastewater treatment (Cheryan and Rajagopalan 1998, Shui Li *et al.* 2006). To meet current discharge limits it has become necessary to integrate two or more of these units since all treatment technologies are limited in application (Tin and Cheng 1994, Mueller *et al.* 1997).

In the industry, the practice is to integrate oil recovery units and biological treatment units. This reduces the cost of treatment by recovering oil which can be sold to offset the cost of treatment associated with aerobically operated biological units.

Reducing the cost of aeration is paramount because it is the major cost in operating aerobic bioreactors (Atkinson 1986). The membrane bioreactor, another recent advancement in oily wastewater technology, was developed to reduce the cost of aeration but this treatment technology has not attained general commercial success due to the high cost of replacing membranes. The Immobilized Cell Bioreactor (ICB) is a relatively new

technology in oily wastewater treatment. In this technology, cells are immobilized onto support particles; these support particles serve two purposes: they ensure cells are not washed away, therefore improving the bioreactor effectiveness and reducing cost of operation, and they also enhance oxygen transfer and further reduce cost due to aeration. Application of this technology in oily wastewater treatment is becoming popular (Park *et al.* 1998, Zhao *et al.* 2006 and Martin *et al.* 2000). The performance of the ICB unit in terms of oxygen transfer then depends on the physical components in the ICB unit such as spargers, sparger design, sparger location, and support particle type and size. To further reduce the cost of aeration, investigation of aeration in the various designs of ICB unit becomes necessary.

The two most widely used oil recovery units in the industry are the horizontal flow separator and the Corrugated Plate Interceptor. To optimize oil recovery and reduce the cost of an integrated process, it becomes necessary to investigate the effectiveness of these units to recover oil.

This project therefore sets out to investigate the treatment of oily wastewater by the effective integration of oil recovery units and biological treatment so as to maximize the cost effectiveness of the overall process.

### 1.3 The Research problem

Figure 1.1 outlines a simple integrated process. The aerobic treatment is intimately linked to the use of oil water separation process(es) to recover oil from the system. As more oil is recovered from the system, the less intense the aerobic treatment has to be. In fact the cost of the system can be significantly improved by a good oil-water separation as recovered oil can be sold; also the bioreactor becomes smaller and the output cleaner.

However, the ICB unit downstream of the oil recovery units contributes to the cost of the process. The cost of biological treatment with the ICB unit varies with oil concentration. Increasing oil concentration implies increasing cost of treatment for an aerobically operated ICB unit. Approximately 50% of energy cost in the bioprocess goes to aeration. This implies aeration is a key factor in improving cost of biological treatment (eutchinst.com).

Therefore to improve the efficiency and cost of the integrated process, the effectiveness of the oil recovery units to recover oil will be investigated and aeration in ICB units will be investigated.

This project therefore sets out to investigate two integrated processes:

- 1. The integration of a horizontal flow separator and an ICB.
- 2. The integration of a Corrugated Plate Interceptor (CPI) and an ICB.



Figure 1.1: Integrated system of gravity separation and ICB

### **1.4 Research Methodology**

Two methods were used in this research to obtain results: theoretical and experimental.

The theoretical aspect involved creating scenarios to investigate the effectiveness of the oil recovery units and their effect on the integrated process. It also involved developing spreadsheet models to generate theoretical oil droplet distributions in various oily wastewater streams, to design separators and predict their performance, and to size the Immobilized Cell Bioreactors required to treat the remaining oil in the effluent of the oil recovery units. Models were also developed in Excel to calculate the blower power required to meet the various oxygen demands of the ICB unit.

The experimental aspect involved a practical investigation of aeration in a laboratory scale Immobilized Cell Bioreactor (ICB) unit. Different ICB unit designs were investigated by using various sizes of Plastic Pall rings (support particles). The mass

transfer characteristics of the various sizes of support particle were investigated to predict oxygen mass transfer in the various designs of ICB. Data obtained from the experimental investigation was used as part of the theoretical investigation (refer to Chapter 3).

For the costing of the various units, cost estimation models were employed (refer to Chapter 2). Calculation procedures for the various models developed are presented in Chapter 3.

### 1.5 Justification for research

- Considering the large number of industries utilizing oil and oil related products, there is a vast amount of revenue that can be generated if oil treatment units can be effectively optimized to recover more oil and treat unrecovered oil at minimum cost.
- A study of this nature may necessitate a change in the current practice of using chemicals to enhance separator effectiveness, a practice which adds cost to treatment and affects the environment.
- To serve as a guide for profitable integration of units for oil water separation and treatment and hence reduce cost of treatment.
- Most reported works (Moosai and Dawe 2003, Al-Shamrani et al. 2002, Lundh et al. 2002) on oily wastewater treatment methods have focused on investigating the factors that affect the effectiveness of each treatment or separation method studied. A smaller proportion has reported the benefits of integrating units (Karakulski et al. 1995, Zhong et al. 2003, Gryta et al. 2001) but no study has yet outlined the window for cost-effective integration of oily wastewater separation and treatment processes.

### 1.6 Research presentation

This thesis is presented in six chapters. The literature review in Chapter 2 justifies the investigation and provides the background information for the remaining chapters. The first section of Chapter 3 describes the materials and methods used for the experimental and aspect of the research and the second section explains the spreadsheet calculation procedures used for the theoretical aspect of the research. Chapter 4 presents and discusses the results from the experimental investigation of aeration in various models of

ICB units. Chapter 5 presents the results from the theoretical investigations; it reports the effectiveness of the oil recovery units in various scenarios and the effects on the integrated process. Chapter 6 reports the conclusions and recommendations for further work.

## **1.7** Scope of the research and key assumptions

The scope of this research is limited to the study of the two most common oil recovery units: the horizontal flow separator and the Corrugated Plate Interceptor. Dissolved Air Flotation, Induced Gas Flotation, and Hydrocyclones are other oil water separation technologies that were not studied because they are not widely used in the industry.

A theoretical waste water stream was used in this research. Though an actual oily wastewater stream could have been characterized to use as a sample, this was not done mostly because the method of using a microscope to count the number of droplets with a particular size is prone to errors. Nevertheless, an actual oily wastewater stream may have the same distribution as a theoretical stream.

Oil droplets are not solid particles but for this analysis they were assumed to be spherically shaped so as to determine the volume of oil contained in each oil droplet. The cost of treating sludge was not taken into account during costing because actual oily waste streams were not used. However, it is important to note that the amount of sludge generated would vary with the source of the oily waste stream.

## Chapter 2: Literature Review

## 2.1 Introduction

This chapter is divided into two broad sections. The first section (2.2) justifies the investigation and the second section (2.3) provides the background information for the rest of the thesis.

## 2.2 Research justification

#### 2.2.1 Sources of oily wastewater and quantity

The volume of oily wastewater (oil-water mixture) generated continuously around the world is very large. This is due to the extensive and continuous utilization of oil and oil related products in most industries.

Major industrial sources of oily wastewater (Smith 1989, Villas-Bôas and Barreto 1996, Warhurst and Bridge 1996, Busca 2004) include oil fields, petroleum refineries, metal manufacturing and machining, minerals and mining. Oil fields and petroleum refineries produce large quantities of oil-water mixtures. In the metal industries, two major sources of oily water mixtures (oily wastewater) are steel manufacture and metal working (Wolfe 1992). Metal manufacturing and machining plants generally utilize machining coolants in their metal cutting, forming, rolling, and finishing operations (Mysore *et al.* 2005). These coolants often consist of chemically stabilized emulsions of oil in water with an oil concentration of 2-5%. Weintraub *et al.* 1983 reported that wastewater discharged from industrial washers may have an emulsified oil concentration of 300-7000 mg/l and free floating oil content of 30,000 mg/l. Nemerow 1978 reported that the concentration of oil in effluents from different industrial sources is found to vary widely from several mg/l to as high as 40,000 mg/l.

These large volumes of oily wastewater call for proper treatment before disposal. Also the amount of revenue which can be generated, or expense saved, from recovering these oils calls for a cost effective method of treatment which not only improves the efficiency of the units but also improves the cost of the process, decreases the number of stages involved and has a small footprint.

#### 2.2.2 Review of oily wastewater separation and treatment units

Various technologies have been developed for oil recovery from oily wastewater, including membrane processes, flotation processes, and hydrocyclones.

The membrane process is the most recent advancement in oil recovery technologies and is a pressure driven process which relies on the pore size of the membrane to separate the oil-water stream into the two streams: the concentrate and the permeate. The concentrate contains the oil phase while the permeate contains the purified water. The flotation process is a concept widely used in the mineral processing industries (Kitchener 1985) but has also been applied for oily wastewater separation. The philosophy behind the flotation processes in the oily wastewater separation application is to increase the density difference between the oil and water in order to increase the rise rate of the oil droplets. The increase in density difference is achieved by attaching gas bubbles to the oil droplets. The hydrocyclone is a unit which operates on the principle that if the natural gravitational field is increased, separation is enhanced and the time required for separation is reduced.

However, these units have not gained general commercial success in the industry. Even though the membrane process is the most effective and has a small footprint, it has not gained general commercial success due to the high cost of replacing membranes (Cheryan and Rajagopalan 1998, Shui Li et al. 2006). The membranes form a synthetic barrier which allows the transport of water (the permeate) through the membrane whilst oil (the concentrate) accumulates on the membrane surface. As a result of oil droplets accumulating on the membrane, the permeating flux decreases with time as more oil accumulates on the membrane surface causing a relatively low flux limit (Li, Yu Shui et al. 2006). This phenomenon, which is known as concentration polarization and membrane fouling, has a devastating effect on the efficiency of the membrane process and limits the application of the membrane process in the industry due to the high cost of replacing membranes. Current research on membrane technology is based on tackling fouling (Li, Yu Shui et al. 2006, Li et al. 2006, and Wang et al. 2006). Several approaches to mitigate this problem have been attempted; Li et al. 2006 developed membranes to tackle fouling, Kim et al. 2007 reported reduced fouling when an electric field was used in the membrane process. Zhong et al. 2003 reported reduced fouling when membranes were coupled with flocculation. Srijaroonrat et al. 1999 and Ma et al. 2000 reported reduced membrane fouling with back pulsing. Kobayashi, et al. 2003

investigated the use of ultrasound for cleaning fouled membranes to increase membrane life. Karakulski *et al.* 1995 reported increased membrane efficiency by coupling membranes. At present there is no outstanding solution to mitigating the effects of membrane fouling, hence the limited use of membranes.

Much effort has been made to improve the efficiency of the flotation process. This has resulted in the development of various techniques to introduce gas to the oil wastewater systems. The most common techniques are Induced Gas Flotation (IGF) and Dissolved Air Flotation (DAF). The major difference between these flotation techniques is the manner in which the gas or air is introduced. However, the flotation process is not common in the industry for many reasons. Smaller air bubbles are required to improve the efficiency of the flotation process, but since smaller bubbles have a low rate of rise, they necessitate the construction of a large tank, and the industry is interested in maintaining a small footprint. Other issues which may have limited the application of flotation processes are the many factors which affect its effectiveness. Such factors are the ability of the gas bubbles to collect oil droplets from the water, the size of the gas bubbles (Moosai and Dawe 2003), the saturation pressure of the unit, the use of chemicals upstream of the flotation process (Al-Shamrani et al. 2002), contact zone configuration in dissolved air flotation (Lundh et al. 2002), type of valve (Bratby and Marais 1975), fluid dynamics (Maddock and Timlinson 1980), nozzle geometry (Takahashi et al. 1979), and concentration of air bubbles (Al-Shamrani et al. 2002). The process seems to be too complicated and its operation remains a black box.

Also there seem to be conflicting reports on how to improve the effectiveness of these processes. Al-Shamrani *et al.* 2002 reported improved performance of dissolved air flotation when integrated with chemicals; Zouboulis and Avranas 2000 reported that not all chemicals (coagulants) are effective in the dissolved air flotation process.

Another factor which may have limited the use of the flotation process may be the requirement for additional units upstream and downstream of the DAF units since the flotation process cannot recover droplets smaller than 20 microns or larger than 150 microns (Moosai and Dawe 2003).

The major limitation of the hydrocyclone unit in the industry is that it requires considerable pumping power, which involves significant cost (Tin and Cheng 1994). Hashmi *et al.* 2004 reported the various factors which affect the efficiency of hydro-

cyclones as orifice size, feed pressure, oil concentration, and conditioning of the water. They concluded that hydrocyclones cannot operate effectively with high concentrations of oil. The inability to handle large oil concentrations may also have resulted in hydrocyclones not being common in the industry. The technology is reported to be efficient in recovering droplets in the size range of 10-30 microns (Tin and Cheng 1994).

The most commonly used oil recovery units are the horizontal flow separator and the Corrugated Plate Interceptor.

The horizontal flow separator is the oldest technology developed and applied in the separation of oil from water; it is more effective in recovering bigger sized oil droplets than smaller sized oil droplets. In the past, the philosophy of design was to oversize these separation units, accepting a large footprint in order to ensure capture of the smaller oil droplets. This practice was adequate until rapid industrial developments and new environmental regulations emerged; the horizontal flow separator then became incapable of meeting more stringent discharge limits whilst maintaining an acceptable footprint (Tin and Cheng 1994; Mueller *et al.* 1997).

The difficulty associated with recovery of smaller sized oil droplets led to the use of chemicals in gravity separators to enhance oil recovery. This was desirable because it meant there was no need to increase separator size in order to improve separator efficiency. Other technologies, such as the Corrugated Plate Interceptor, were also developed due to the inability of horizontal flow separators to recover small oil droplets (Li, Yu Shui *et al.* 2006).

Smaller sized oil droplets pose problems to the industry. Smaller sized droplets are formed as a result of turbulent mixing of the oil and water phases, resulting in the oil droplets being uniformly dispersed in the water (Hong *et al.* 2003). When this mixing is done in the absence of chemicals the resulting mixture is called an unstable emulsion; it is unstable because oil has a natural repulsion with water and will break down through a slow process.

Unstable emulsions break down by the following mechanisms:

- Creaming,
- Flocculation, and
- Agglomeration.

Creaming is a gravitational separation process where the lighter phase oil droplets rise through the system with time; this rise is initiated by the density difference between the continuous and dispersed phases. The rate of creaming is governed by the size of the oil droplets. Creaming in itself does not destabilize an emulsion, but the high concentration of oil droplets in the creamed layer is what promotes interactions that lead to flocculation, aggregation, or coalescence (Capek 2004) and hence the total breakdown of an emulsion.

Flocculation occurs as a result of agitation or any other motion or chemical which causes two or more oil droplets to lump together. In flocculation each oil droplet retains its identity but looses its kinetic independence since the aggregate moves as a unit.

Agglomeration occurs as a result of flocculation. In this mechanism droplets loose their individual identity to form larger droplets which have faster rate of rise.

The problem posed by this unstable mixture is that it takes a long time for the oil droplets to rise to the surface. Consequently, the industry has resorted to the use of chemicals to enhance droplet rise rates.

Smaller sized droplets formed as a result of turbulent mixing in the presence of surfactants are called stabilized emulsions (Capek 2004). Surfactants stabilize the oil droplets by developing surfaces charges on them; this lowers the interfacial tension between water and oil and prevents the droplets from coalescing (Capek 2004). The overall effect is a very slow droplet rising time, such that a horizontal flow separator is unable to recover oil from these emulsions.

To recover these droplets, chemicals called coagulants and flocculants are employed (Beisinger *et al.* 1974, Gardner 1972, Lash and Kominek 1975). Coagulants act to overcome the effect of surfactants by destroying the protective action of the surfactants, i.e. by overcoming the repulsive effects of the electrical double layers to allow the smaller sized oil droplets to form larger droplets through coalescence; flocculants act to promote agglomeration of the destabilized droplets, therefore increasing the rise rates. This method of recovering oil is the norm in the industry but it poses additional problems to the environment and additional cost to the oily wastewater producers. It produces hazardous sludge which must be treated and requires further treatment of the effluent stream which all adds to the cost of treatment and the number of stages involved in treatment.

Heating of oil-water mixtures to reduce the water viscosity and hence increase the rise rate of oil droplets is another method used to reduce separator size. This thermal process involves considerable cost due to the high cost of energy (Gryta *et al.* 2001, Cheryan and Rajagopalan 1998).

The Corrugated Plate Interceptor was developed by the Royal Dutch Shell group in the 1950's (Tin and Cheng 1994). The design of this interceptor was based on two concepts:

- 1. If the distance the smaller droplets need to rise is reduced then the long residence time or large tank required by a horizontal flow separator can be avoided.
- 2. If the smaller droplets can increase in size then the rise rates will be increased (This is discussed in Section 2.3.1).

These conditions were achieved by introducing plates into the separation chamber (Iggleden 1978). The plates reduced the distance for a droplet to rise and also acted as a surface for droplets to coalesce on. This concept led to development of several types of plate separators (the various types of plate separators are discussed in Appendix I), all aimed at increasing efficiency. The physical process which takes place in Corrugated Plate Interceptors has been described by Rommel *et al.* 1992. They explained that when a dispersion is fed to a plate interceptor, the lighter phase ascends to the upper plate and forms a trickling film which flows along the plates following the hydrostatic pressure to the principal interface (that is between the lighter phase and heavy phase). Subsequent droplets coalesce on the trickling film. At this point of coalescence any surfactants present are swept away by the trickling film. This phenomenon eliminates the use of chemicals to improve recovery.

However, the horizontal flow separator and CPI are limited in application. They cannot effectively recover smaller sized oil droplets (less than 60 microns). The practice in the industry is to integrate these units with a biological process. This necessary is to meet environmental standards, but the cost of integration is high. Also, with the increasing value of oil, the industry seeks methods to improve oil recovery and reduce the cost of integrated processes.

To address the need to improve oil recovery and to reduce the cost of treatment, this project investigates the effectiveness of these oil separation units in various scenarios.

#### 2.2.3 Biological reactors and the Immobilized Cell Bioreactor (ICB)

Biological reactors are most effective in treating unrecovered smaller sized oil droplets. The biological process actually entails the transformation of oil and other dissolved organics into biomass and gases ( $CO_2$ ,  $CH_4$ ,  $N_2$ , and  $SO_2$ ) (Low and Chase 1999) but its efficiency depends on a number of factors.

One of the critical factors which affect the performance of any bioreactor is the concentration of microbes in the reactor. Different configurations of bioreactors have since been developed, all in an effort to improve the concentration of microbes and hence efficiency. The initial configuration involved suspending the microbes freely in the Many problems were associated with this configuration (conventional reactor. bioreactor) such as loss of microbes, the need to operate within a certain range of flow rates or dilution rates, short retention time and high sensitivity to fluctuations in flow, concentration and temperature, high cost of operation (Gray 1989), and high sludge production. The treatment of sludge accounts for 60% of the total cost of treatment These limitations led to the development of various plants (Wei et al. 2003). configurations for bioreactors, for example the fixed film technologies and membrane Table 2.1 compares the various bioreactor configurations; the major bioreactor. advantage of the submerged biological contactor over the rotating biological contactor is its compact design.

The membrane bioreactor (MBR) is the most recent innovation in bio-processing. It is an integrated system of membranes and bioreactor. Water passes through the membranes whilst oil is retained as the concentrate; any oil passing into the permeate is oxidized by microbes. The MBR is currently the only single treatment method which recovers valuable oil as well as degrading what remains to meet discharge limits; but it has some major disadvantages: membrane fouling and the high cost of membranes (Zhi-Guo *et al.* 2005, Wang *et al.* 2006). Whilst the biological and chemical oxygen demands of the process are reduced by the initial stage of oil recovery, high aeration rates are still needed to suppress fouling. Le-Clech *et al.* 2003 reviewed the literature and listed the various factors that suppress the onset of fouling in MBRs as increasing aeration rate and decreasing channel width. The problems with MBRs can then be said to be two-fold: first, the high cost associated with membrane replacement, and second, the high cost

associated with aeration. These issues associated with membrane fouling have restricted the commercial utilization of MBRs (Zhi-Guo *et al.* 2005 and Li *et al.* 2005).

Evaluation parameter	Trickling	Rotating	Submerged	Membrane
	niter	Diological	Diological	Dioreactor
		contactor	contactor	
Effective bioassay/toxicity control				$\checkmark$
Effective BOD removal efficiency				✓
Effective COD removal efficiency				$\checkmark$
Low O&M		~	<ul> <li>✓</li> </ul>	~
Low sludge production	<ul> <li>✓</li> </ul>	✓	<ul> <li>✓</li> </ul>	$\checkmark$
Low sludge disposal cost	✓	✓	$\checkmark$	$\checkmark$
Minimum operator attention		✓	$\checkmark$	
Quick upset recovery	<ul> <li>✓</li> </ul>	✓	<ul> <li>✓</li> </ul>	
Energy efficient	1	✓	✓	
Minimum space requirements				$\checkmark$

Table 2.1: Comparison of fixed film and membrane technologies (Schultz 2005)

The Immobilized Cell Bioreactor (ICB) is another variation of the fixed film technologies. It offers a larger surface area for microbes compared to the other fixed film technologies and hence is more efficient; due to the presence of support particles it is also well mixed with respect to aeration - this allows for better mass transfer and control of biofilm thickness. Other advantages of the ICB are:

- Smaller reactor size due to immobilization of cells, which allows for higher degradation rates.
- Better protection against toxic or inhibitory compounds than free-cell systems due to the protective effect of concentration gradient that occurs within the biofilm.
- Operation at higher dilution rates without fear of washout and a much lower operating cost compared to the membrane bioreactor.

Immobilized Cell Bioreactors (ICB) are gaining ground in oily wastewater treatment (Park *et al.* 1998, Zhao *et al.* 2006 and Martin *et al.* 2000) but oxygen transfer remains crucial to their performance (Atkinson 1986). Many efforts have gone into overcoming the problem of oxygen transfer. Holst *et al.* (1982) reported generating oxygen in-situ by decomposing hydrogen peroxide using the high catalytic activity present in a micro-organism called *Glucunobacter oxydans*. Adlercreutz *et al.* 1982, Brodelius *et al.* 1981 and Szwajcer *et al.* 1982 reported generating oxygen by co-immobilizing oxygen producing micro-organisms and oxygen consuming organisms and co-immobilizing

hydrogen peroxide degrading agent, such as activated carbon, and microbial cells. Elibol and Mavituna 1995 reported generating oxygen in immobilized cells using perfluorocarbons (PFCs) or silicon compounds. The problems associated with these methods are:

- Inherent toxicity associated with the carriers.
- Carriers are expensive e.g. Perfluorocarbons (Elibol and Mavituna 1995).
- There is no inherent mixing which is necessary for transfer of other nutrients.

All of these efforts directed to overcoming the problem of oxygen transfer are due to the high cost of aeration. If aeration problems can be eliminated then operation of bioreactors becomes less expensive and the cost of an integrated process involving an aerobically operated bioreactor becomes greatly reduced and cost of treatment improved. However, this has not been achieved so there is room for other solutions; one solution is to investigate other ways of reducing the cost of aeration.

This project therefore investigates aeration in various ICB units so as to reduce the cost of aeration.

A review of literature on oily wastewater treatment reveals that much study on oily wastewater has centred on developing models for design and for predicting separator performance. Nassif and Hansard (2003) developed a new approach to the design of oil water separators; they claimed Stokes' Law does not adequately describe the rise rate of a smaller sized droplet. Nakhla *et al.* 2006 developed a kinetic model for aerobic degradation of oil and Mohammadi *et al.* 2006 developed a model to predict membrane fouling. Though these studies have provided better understanding of these technologies they have failed to identify critical factors that optimize each unit so as to optimize the unit or an integrated process.

Other studies have been centred on integrated processes but these studies have only reported the benefits of integration. Zhong *et al.* 2003 reported improved performance of micro-filtration when coupled with chemical flocculation as a pre-treatment. Gryta *et al.* 2000 reported that the permeate obtained from an ultrafiltration (UF) process generally contains less than 5 ppm of oil, but when coupled with membrane distillation complete recovery of oil from water was achieved. Karakulski *et al.* 1995 investigated purification of oily wastewater using ultrafiltration and reported a permeate of acceptable quality was

obtained for direct sewage discharge but reported further treatment (such as reverse osmosis) will permit the re-use of the water as process water. Although these studies are commendable they fail to provide a guide to the integration of oil recovery units and biological treatment units so as to minimize the overall cost of treatment.

### 2.2.4 Aims/Objectives

The industry is currently interested in maximizing treatment at minimum cost.

This project therefore sets out to investigate the treatment of oily wastewater by the effective integration of oil-water separation and biological treatment so as to maximize the cost effectiveness of the treatment. It is hoped this study will serve as a guide to profitable integration of these units.

The objectives of this study are therefore as follows:

- To investigate aeration in various ICB laboratory models.
- To develop models to design and size units in the integrated processes and predict their performances.
- To investigate the effectiveness of oil recovery units to recover oil in various scenarios and the effect on the cost of integrated processes.
- To investigate regions of economic operation and economic designs.

## 2.3 Theoretical aspects used in the research

This section introduces the theoretical concepts used in the research.

#### 2.3.1 Oily wastewater

Oily wastewater is defined by its oil concentration, oil density, water temperature, average droplet size, and viscosity. All of these wastewater characteristics affect the performance of separation or treatment technologies in various ways. For example, droplet size affects the performance of gravity separators, flotation processes, membranes, and bioreactors. Specifically, large oil droplets coat and kill microbes in bioreactors and therefore reduce efficiency and increase costs.

The state in which oil exists in water governs the choice of treatment technology. Large oil droplets are most economically removed by gravity separation technologies while smaller droplets are best removed by biological treatment or membranes. Generally, oily wastewater is classified according to the states in which oil may exist in water; these states are: as free-floating droplets, unstable or dispersed emulsions, and stable emulsions (Rhee *et al.* 1987). Table 2.2 outlines oil-water classifications.

	Oil droplet size range (µm)
Free floating oil	>150
Unstable emulsions	20-150
Stable emulsions	<20

Table 2.2: Oil-water system classification

While the droplets are not actually particles the convention is to treat each discrete oil droplet suspended in water as a spherical droplet. Oily wastewater is modelled depending on the droplet diameters (Table 2.2); sampling techniques such as the dynamic light scattering method (Poulsen *et al.* 2007) and video recording techniques (Angeli and Hewitt 2000) are used in determining the number of droplets of a particular size in a given volume of oil water mixture. A wide variety of mathematical and empirical distribution functions, such as the normal distribution, log-normal distribution, Rosin-Rammler equation and Nukiyama-Tanasawa equation have commonly been used to

simulate the droplet size distributions (Lefebvre 1989). The normal distribution does not usually show a good fit to experimental data, since the distributions are rarely symmetrical. To fit experimental data, Rosin–Rammler (Angeli and Hewitt 2000) (Mugele and Evans 1951) and log-normal distributions are usually employed (Sood and Awasthi 2003).

The oil droplet distribution shows the size distribution of droplets in a given sample of oily wastewater. The volume of oil contributed by droplets of a particular size depends on the size of the droplets, since the volume of oil contributed by a droplet is a function of the droplet size (Equation 2.0).

$$V_{oil} = \frac{\pi}{6}d^3 \tag{2.0}$$

where,

 $V_{oil}$  - Volume of oil in a droplet (m<sup>3</sup>)

*d* - Oil droplet diameter (m)

The total volume of oil contributed by droplets of a particular size in a distribution is therefore:

$$V_t = n \times \frac{\pi}{6} d^3 \tag{2.1}$$

where,

- total volume of oil contributed by all droplets of a particular size  $(m^3)$ 

*n* - Number of droplets in a size range or interval

In this research, droplet distribution was generated using theoretical functions (Equation 2.2). A normal distribution was assumed. Based on an assumed mean droplet size and standard deviation, the normal distribution was obtained.

$$F(\mathbf{x}, \mathbf{x}_m, \sigma) = \frac{1}{\sigma\sqrt{2\pi}} e^{-\left[\frac{(\mathbf{x}-\mathbf{x}_m)^2}{2\sigma^2}\right]}$$
(2.2)

where,

x - Droplet size of interest (μm)

 $x_m$  - Mean droplet size (µm)

Page 18

 $\sigma$  - Standard deviation from the mean

## 2.3.2 General review of Immobilized Cell Bioreactors

Immobilized cell bioreactors are bioreactors in which cells are immobilized on support particles (packing). Cell immobilization offers a lot of advantages:

- It facilitates cell separation.
- It leads to a high cell concentration within the reactor.
- It allows continuous operation beyond washout flow rates.
- Protects against contamination.
- Permits manipulation of growth rate in continuous systems independently of dilution rate.
- Provides possibilities for spatial location of different microbial populations within reactors.

Atkinson 1986 identified that the performance of any reactor containing immobilized cells depends upon:

- the number of support particles,
- the average biomass hold-up per support particle
- the average specific rate of reaction of the immobilized biomass and
- the average overall yield coefficients of the immobilized biomass

The yield coefficient and specific rate of reaction of the immobilized biomass depend, amongst other factors, on the ability of the cells to obtain oxygen, which in turns depends on the support particle size (Atkinson 1986, Da Fonseca 1983). The average biomass hold up per particle is subject to diffusional limitations on the rates of reaction, which are also affected by the support particle size. It can then be concluded that the support particle (packing) is the key feature of an ICB. It acts as the surface on which cells are immobilized and it also acts as a static mixer for the bioreactor, eliminating the need for mechanical stirrers which can disrupt immobilized aggregates. The support packing also affects pressure drop in a packed bed reactor. Da Fonseca *et al.* (1986) conclude that particle size may be considered as a design parameter in bioreactor design. For this

reason, the effect of various sizes of plastic Pall support particle on mass transfer coefficient was investigated so as to minimize the cost due to aeration in the ICB.

Equation 2.3 represents the biodegradation process that takes place in an ICB.

$$C_n H_{2n} + O_2 \rightarrow CO_2 + H_2O + Cells \qquad (2.3)$$

Theoretically, for every 3.5 grams of oil oxidized, 3.5 grams of oxygen is required but the rate at which it is transferred depends on the mass transfer coefficient, which is affected by support particle size.

#### **2.3.2.1** Mass transfer theories

Oxygen transfer is the primary use of energy in the bioreactor and represents a significant capital cost as well as a running cost. The microbes are living cells which need oxygen for growth, product formation, and maintenance. The volumetric mass transfer coefficient indicates the rate at which oxygen is transferred. The higher the rate of oxygen transfer, the better the performance of the bioreactor and the lower the cost of aeration. Correct determination of the volumetric mass transfer coefficient is important in bioreactor design.

Various theories have been developed to describe oxygen mass transfer from the gaseous phase to the liquid phase. The most widely used theory is the Lewis Whitman theory. This theory states that when two fluid phases are brought in contact with each other, a thin layer of stagnant fluid exists on each side of the phase boundary (Lewis and Whitman 1924). Mass transfer by convection is assumed to be insignificant in this unstirred layer and the transport is achieved by steady state diffusion. This theory postulates that the gas concentration in the liquid film is in equilibrium with the gas phase while the gas concentration in the bulk liquid is kept uniform by turbulent mixing. The main resistance to mass transfer lies in the films. For a gas which is sparingly soluble in water, for example oxygen, Lewis theory postulates that the main resistance to transfer of oxygen lies in the liquid film and the mass transfer coefficient  $k_L$  is expressed as:

$$k_L = \frac{D}{\delta} \tag{2.4}$$

where,

- mass transfer coefficient in the liquid phase (m/s)
D - Diffusion coefficient of the oxygen in liquid (m<sup>2</sup>/s)

 $\delta$  - film thickness (m)

and the rate of mass transfer is expressed as

$$\frac{\partial m}{\partial t} = k_L A \left( C_s - C_L \right) \tag{2.5}$$

where,

*m* - Mass of oxygen (mg)

A - Interfacial area normal to mass transfer  $(m^2)$ 

 $C_s$  - Saturation dissolved oxygen concentration (mg/l)

 $C_L$  - dissolved oxygen concentration in liquid (mg/l)

Equation 2.5 divided by the liquid volume, V, gives

$$\frac{\partial C}{\partial t} = = k_L a \left( Cs - C_L \right) \tag{2.6}$$

where,

$$\frac{\partial C}{\partial t}$$
 - is the concentration change with respect to time

*a* - specific interfacial area,  $(A/V) (m^2/m^3)$ 

Other mass transfer theories are Higbie's and Danckwerts' theories. Higbie's theory refutes the interface as being unstirred but as being made up of small liquid elements, which are continuously brought up to the surface from the bulk of the liquid by the motion of the liquid itself (Higbie 1935). Higbie's theory postulates that the rate at which mass transfers to the liquid film is greater than the rate at which the mass leaves the film, so when a fresh element of liquid comes into the interface, unsteady state diffusion readily occurs since there is no gaseous component in the stream. Higbie's theory assumes that these fresh elements of liquid all stay the same time at the surface before being replaced by another stream of fresh liquid elements. Higbie's theory mass transfer coefficient is expressed as:

$$k_L = 2\sqrt{\frac{D}{\pi\theta}} \qquad (2.7)$$

 $\theta$  - is the time of exposure of the liquid element (s)

Danckwerts' theory also opposes the concept of a stagnant film as suggested by Lewis theory and builds on Higbie's theory. While Higbie proposed that liquid elements are continuously brought up and stay the same time at the surface, Danckwerts' theory, known as surface renewal theory, proposes that the liquid elements do not stay the same time at the surface but are renewed at a rate known as the surface renewal rate (Danckwerts 1951). Danckwerts' theory assumes that if turbulent mixing occurs in the liquid in the proximity of a bubble, it is likely that unsteady state molecular diffusion will take place in individual eddies.

Danckwerts' mass transfer coefficient is expressed as:

$$k_L = \sqrt{D.s} \qquad (2.8)$$

where,

*s* - Average frequency at which any particular vertical element is mixed or rate of surface renewal (s<sup>-1</sup>)

The surface renewal rate increases with turbulence in Equation 2.8.

Higbie's and Danckwerts' theories were not used in this investigation because of the difficulty of determining both the liquid film coefficient and the interfacial area. The Lewis Whitman theory combines these parameters together. Determination of the interfacial area requires measurements of the bubble size, their lifetime in air or water, and the statistical knowledge of their distribution. The two-film theory on the other hand uses both  $k_L$  and A/V together as a lumped parameter,  $k_{La}$ ; the combining of the two parameters essentially eliminates the difficulty in determining the interfacial area a, which strategically makes it more favourable than the two other theories. In the determination of the mass transfer characteristics of the various sizes of plastic Pall rings, the Lewis Whitman theory was employed.

### 2.3.2.2 Bioreactor design

The characteristics of a well-designed bioreactor have been spelt out by various authors. Williams 2002 reported that a well-designed bioreactor is one which takes reactor parameters for the biological, chemical, and physical system (macro-kinetic) and bio-

#### Investigation of Oily Wastewater Treatment Processes

reaction parameters into consideration during design. Freeman and Lilly 1998 define a well-designed immobilized cell bioreactor as one which takes a set of processing parameters into consideration during design. These include: the immobilization method, mode of operation (e.g., repeated batch vs. continuous), aeration and mixing, bioreactor configuration, medium composition (including feeding of substrates, precursors, or additional nutrients), temperature, pH and, whenever required, *in situ* product and/or excess biomass removal. In conclusion, a well-designed bioreactor is one which can specifically influence the metabolic pathways of the immobilized cells.

As stated previously, oxygen supply is a key issue in the design of an aerobic immobilized cell bioreactor (Da Fonseca *et al.* 1986). This problem is particularly acute in immobilized cells where oxygen has to diffuse through a solid phase. Immobilized cells depend on transport of oxygen with molecular diffusion or convective flow through the support materials used. The cells also depend on the diffusion of oxygen into the micro-colonies which have been formed by their growth. Diffusion limitations do exist in such colonies. Da Fonseca *et al.* 1986 compared the activity of immobilized cells at maximum cell hold-up using two aerating media. When aerating with air, the immobilized cell activity was estimated as 0.025, compared to 0.06 when aerating with pure oxygen. This confirmed the severity of oxygen diffusion through the solid phase.

Therefore to ensure complete degradation of oil in the bioreactor the following assumptions were made:

- No diffusion limitations exist.
- Sufficient cells are present to carry out the transformation.
- Immobilized cells are held to high concentration.
- Material of construction: carbon steel, coated to prevent corrosion or contamination.

To obtain complete degradation, bioreactors were sized to match oxygen demand for a given oxygen transfer rate (see Equation 2.9).

$$V_{ICB} = \frac{OD}{OTR} \tag{2.9}$$

 $V_{ICB}$  - Immobilized cell bioreactor volume (l)

*OD* - Oxygen demand (mg/s)

*OTR* - oxygen transfer rate (mg/l.s)

where,

 $OTR = k_{La} (Cs - C) \qquad (2.9a)$ 

#### 2.3.2.3 Bioreactor scale-up

The volumetric mass transfer coefficient is the most widely used index for bioreactor design and scale-up. Maintaining a constant volumetric mass-transfer coefficient  $(k_{La})$  is a common scale-up strategy in bioreactor scale-up. In this bioreactor design, to meet the oxygen demand, the laboratory immobilized cell bioreactor (ICB) was scaled up by expanding the capacity length-wise and width-wise whilst keeping the height (depth) constant to maintain the same hydrodynamics conditions in the model ICB.

#### 2.3.2.4 Aeration system design

Oxygen transfer in immobilized cell bioreactors is achieved by aeration. Aeration accounts for the largest portion of operating costs (Atkinson 1986).

In this research, oxygen transfer is achieved by sparging air into the bioreactor. The aeration system consists of a blower and a coarse bubble diffuser. The blower was sized based on the oxygen demand of the process. The oxygen transfer efficiency of the coarse bubble diffuser was assumed to be 29.53%/m diffuser submergence (29.53% of the oxygen contained in 0.00508 m<sup>3</sup>/s of air at standard conditions of 25°C and 100 kPa is transferred per metre of diffuser submergence). The blower horsepower requirements for each application were then estimated using the adiabatic compression formula:

$$BP = \frac{653.5Q_{A} \left[ \left( \frac{100 + P}{100} \right)^{0.283} - 1.0 \right]}{\eta}$$
(2.10)

where,

*BP* - Brake Power (kW)

- $Q_A$  Volumetric flow rate of air (m<sup>3</sup>/min)
- $\eta$  Mechanical efficiency (%)
- *P* Blower discharge pressure (kPa)

A mechanical efficiency of 70% was assumed in estimating the horsepower requirements of the blower.

Blower discharge pressure was estimated based on static submergence, inlet losses, and equipment losses and piping losses. An estimate of 13.8 kPa was used in all designs.

The air volume required based on the coarse bubble oxygen transfer coefficient can be calculated from:

$$Q_A = \frac{OD}{OT} \tag{2.11}$$

where,

OT - Oxygen actually transferred / air volume (kg/m<sup>3</sup>)

## 2.3.3 Oil water separation design theory

The theory which governs the design of gravity separators (horizontal flow and corrugated plate interceptors) is based on the relative velocities of the oil droplets and the waste stream. An oil droplet in a separator has two velocity components (http://www.megator.com/pdf/Parallel%20plate%20USA.pdf):

- Horizontal velocity due to the forward movement of the waste stream  $(v_h)$  and
- The rising velocity of the oil droplet or terminal velocity  $(v_o)$ .



Figure 2.1: Velocity components of an oil droplet in a flowing stream

The resultant of the two velocity components describes the trajectory of the oil droplet (Figure 2.1). An oil droplet flowing in a separator will be removed or recovered if the resultant trajectory of the oil droplet will allow it rise out of the effluent flow path before the bulk flow reaches the separator exit (Figure 2.2).



Figure 2.2: Trajectory of an oil droplet in a moving stream

Usually, in separator design, the objective is to completely remove droplets of a certain size: this droplet is called the design droplet or critical droplet. Separator dimensions are calculated based on the rise velocity of the critical droplet  $(v_t)$ .

To ensure separation the following must be achieved:

- The time it takes the critical droplet to rise to the surface where it will be captured or recovered must be less than the time it takes the waste stream to flow to the exit.
- The rate of rise of the oily waste stream (overflow rate) must be less than the rate of rise of the droplet velocity.

The overflow rate is defined as:

$$O_R = \frac{Q}{A_s} \tag{2.12}$$

where,

 $O_R$  - Overflow rate (velocity of rise of the waste stream) (m/s)

 $A_s$  - Surface area required for separation (m<sup>2</sup>)

Q - Volumetric flow rate of waste stream (m<sup>3</sup>/s)

The rise rate of a droplet as described by Stokes' Law (Nordvik *et al.* 1996) is expressed as:

$$v_o = \frac{\pi d^2 (\rho_w - \rho_o) g}{18\mu}$$
 (2.13)

where,

 $v_o$  - is the terminal velocity of the oil droplet (m/s)

*d* - is the diameter of the oil droplet (m)

$$\rho_w$$
 - is the density of water (kg/m<sup>3</sup>)

$$\rho_o$$
 - is the density of the oil (kg/m<sup>3</sup>)

- g is the acceleration due to gravity  $(m/s^2)$
- $\mu$  is the dynamic viscosity of the continuous phase (Pa.s)

The derivation of this law follows from the postulation that two forces act on an oil droplet moving in a fluid. These forces are the buoyancy and drag forces. The buoyancy force is related to the density differential between oil and water and is responsible for the upward movement of any droplet, while the drag force is related to the velocity of the droplet and opposes the upward movement of the oil droplet. Stokes postulated that the oil droplet reaches its terminal velocity when the two forces are equal. The buoyancy force is expressed as:

$$F_B = \frac{\pi}{6} d^3 (\rho_w - \rho_o) g \quad (2.14)$$

 $F_{\rm B}$  - the buoyancy force in Newton (N)

*d* - is the diameter of the oil droplet (m)

The drag force is expressed as

$$F_d = 3\pi\mu v_o d \qquad (2.15)$$

where,

 $F_d$  - is the drag force in Newton (N)

 $\mu$  - is the dynamic viscosity of the continuous phase (Pa.s)

*d* - is the diameter of the oil droplet (m)

when

$$F_B = F_d \tag{2.16}$$

then the terminal velocity  $v_o$  is attained (see Equation 2.13).

Stokes' Law adequately describes the rise rate of droplets only if the following conditions exist:

- Flow is laminar.
- Droplets are spherical.
- There is no interaction between oil droplets (ideal distribution).

## 2.3.4 General review of horizontal flow separators

The horizontal flow separator is basically a settling tank divided into sections:

- Inlet zone,
- Retention zone, and
- Outlet zone.

The baffles in the separator allow for smooth, slow, undisturbed flow along the length of the separator. This separator is an improvement of the settling tank which was only suitable for batch operations. The horizontal flow separator is designed for continuous operation.

### Investigation of Oily Wastewater Treatment Processes

Vapour losses from horizontal flow separators are minimized by use of fixed or floating covers. Various materials are used in the construction of a horizontal flow separator, they include: stainless steel, reinforced concrete and carbon steel.

### Advantages of the horizontal flow separator

- Easy to construct.
- Reliable and simple.
- Inexpensive little or no maintenance.



Figure 2.3: A typical horizontal flow gravity separator

## 2.3.4.1 Design of horizontal flow separators

The critical design droplet size for a horizontal flow separator is typically 150 microns. Using API specifications (American Petroleum Institute, February 1990), the separator is designed so that the two conditions stated below are satisfied.

- The rise rate of the wastewater must be less than or equal to the critical droplet rise rate.

 $O_R = v_t \tag{2.17}$ 

- The rise time of the critical droplet must be less than the residence time of the flow stream.

$$t_t \le t_w \tag{2.18}$$

- $v_t$  rise rate of the design droplet or critical droplet (m/s)
- $t_t$  rise time of critical droplet (s)
- $t_w$  Residence time of oil-water stream (s)

Separator dimensions are governed by Equation 2.18a.

$$\frac{\left(v_{h} \times H_{s}\right)}{L_{s}} < v_{t} \qquad (2.18a)$$

where,

- $H_s$  Separator depth (m)
- $L_s$  Separator length (m)

The American Petroleum Institute (API) has developed a set of guidelines for the design of horizontal flow separators; these guidelines were used in this research.

The principles governing separator design as laid down by the API are based on four relations:

- A minimum horizontal area.
- Horizontal velocity.
- A minimum vertical cross section area.
- Depth to width ratio.

#### Minimum horizontal area

The API stipulates that the minimum surface area required for separation has to be corrected by a factor F (see Equation 2.19). This factor allows for turbulence and short-circuiting (a lower than ideal performance).

$$A_s = F\left(\frac{Q}{v_t}\right) \tag{2.19}$$

- Design factor as a result of turbulence  $(F_t)$  and short circuiting  $(F_s)$ 

The  $F_s$  or short-circuiting factor is a function of  $(R_1)$  the ratio of the horizontal flow  $(v_h)$  velocity to the rise rate of the critical droplet  $(v_l)$ . Typical values of F are displayed in Appendix I.

#### Horizontal velocity

To limit the extent of turbulence which affects separator operation the API specifies that the horizontal velocity of the waste stream is:

$$v_h = 15 \times v_t \tag{2.20a}$$

or,

$$R_1 \times v_t = v_h = 0.015$$
 (2.20b)

where,

 $R_{I}$  - is ratio of horizontal flow velocity  $(v_{h})$  to rise velocity of critical droplet  $(v_{l})$ .

The smaller of the two values from Equations 2.20a or 2.20b is usually the desired  $v_h$ .

#### Minimum vertical cross section area

The minimum vertical cross-sectional area of the separator is then calculated from:

$$A_h = \frac{Q}{v_h} \tag{2.21}$$

Based on conditions specified by API the width of the separator is calculated, using Equation 2.21a.

$$w_s = \left(\frac{Q}{R_2 \times v_h}\right)^{0.5} \tag{2.21a}$$

To reduce turbulence the following ranges for separator dimensions are desirable:

$$0.9144 \ m \le H_s \le 2.4384 \ m \tag{2.21b}$$

$$1.8288 \ m \le w_s \le 6.096 \ m \tag{2.21c}$$

where,

 $w_s$  - Separator width (m)

Depth to width ratio

Desirable depth to width ratio is given in Equation 2.21d.

$$R_2 = \frac{H_s}{w_s} = 0.3 \text{ to } 0.5 \qquad (2.21d)$$

where,

$$R_2$$
 - is ratio of separator depth ( $H_s$ ) to width ( $w_s$ )

From Equation 2.21d the separator depth becomes:

$$H_s = R_2 \times w_s \tag{2.21e}$$

Based on the values of F,  $R_I$  and  $H_{s_i}$  the length of the separator can be calculated using Equation 2.21f.

$$L_s = F \times R_I \times H_s \tag{2.21f}$$

and,

$$L_s > H_s \tag{2.21g}$$

#### 2.3.4.2 Analysis of droplet capture in horizontal flow separator

Theoretically, the droplets that will be 100% recovered in a horizontal flow separator are those droplets with a rise rate equal to or greater than the overflow rate.

Any droplet with rise velocity less than the critical droplet velocity, or overflow rate, will be removed in the following proportions:

$$X = \frac{h_o}{H_s} \tag{2.22}$$

where,

*X* - Fraction of droplets of a particular size recovered

 $h_{\rm o}$  - height travelled by any droplet (m)

The height travelled by any droplet size is a function of the residence time of the wastewater.

$$h_o = v_o \times t_w \tag{2.22a}$$

 $t_w$  - residence time of oil-water mixture in separator (s)

and,

$$t_{w} = \frac{L_s}{v_h} \tag{2.22b}$$

When X=1, then all droplets of that size are recovered and when X is less than 1 then the fraction of droplets of that size captured is in the proportions defined by Equation 2.22. The fraction of droplets not recovered is:

$$Y = 1 - X$$
 (2.23)

where,

Y - is fraction of droplets of a particle size not recovered or captured

The un-recovered fraction flows to the separator exit.

This analysis is based on the principle of unhindered settling (Metcalf and Eddy 1991) which assumes the following:

- Oil droplets are spherical droplets.
- Ideal distribution (no interaction of oil droplets).
- Particles with the same rising velocity move in parallel paths.
- Even distribution of droplets across the inlet cross-sectional area.
- Laminar flow.

### 2.3.5 General review of Corrugated Plate Interceptors

A corrugated plate interceptor is basically a separation tank with coalescing plates inclined at an angle. The inclination of the plates allows for removal of solids, as well as oil, from the waste stream.

The plate interceptor improves on the horizontal flow gravity separator by providing a much smaller distance for oil droplets to rise and an increased surface area. The shorter distance travelled by the oil droplets is accomplished by having coalescing plates in the separation chamber. The oil droplets only have to rise to the next higher plate where they attach to other droplets. The coalesced oil droplets then rise alongside the underside of

#### Investigation of Oily Wastewater Treatment Processes

each plate. At the top of the plate, large droplets break away and rise to the surface where they are skimmed off (Figure 2.4).

The increased surface area of the corrugated plate separator is as a result of the inclusion of the plates. The combined area of all the plates constitutes the total surface area of the plate separator. Various materials are used in the construction of CPI units: stainless steel, reinforced concrete, and carbon steel.

Flow into a corrugated plate pack can be either down flow or cross-flow.

In down flow separators, the wastewater flows down between the plates; sludge deposits on the top of the plates and flows to the bottom of the separator, whilst oil accumulates beneath the plates and flows counter currently to the top of the separator.

In cross flow separators, flow enters the plate pack from the side and flows horizontally between the plates. Oil and sludge accumulate on the plate surfaces above and below the wastewater. As the oil and sludge build up the oil globules rise to the separator surface and sludge gravitates towards the bottom of the separator.

### Advantages of the Corrugated Plate Interceptor (CPI)

- Absence of moving parts.
- It is compact compared to horizontal flow separators.
- An inexpensive method of enhancing separation efficiency.
- Allows for higher wastewater flow compared to horizontal flow separators.



Figure 2.4: A cross flow Corrugated Plate Interceptor

### 2.3.5.1 Design of corrugated plate interceptors

The corrugated plate separator is usually sized to remove droplets of size 60 microns.

The effective surface area required for separation for any given flow rate is obtained from Equation 2.24.

$$A_s = F_{dp} \frac{Q}{v_r} \tag{2.24}$$

where,

 $F_{dp}$  -is a design factor for corrugated plate interceptors (%)

Since plates are inclined, only the projected horizontal area is effective for separation (see Figure 2.5).

The projected area available for separation for given plate dimensions is:

$$A_P = L_P \times w_P \times \cos \alpha \qquad (2.25)$$

where,

- $A_P$  projected plate area of a plate (m<sup>2</sup>)
- $L_P$  plate length (m)
- $w_P$  plate width (m)
- $\alpha$  angle of inclination of the plates to the horizontal (degrees)



Figure 2.5: Effective (projected) area of an inclined plate

Page 35

The total number of plates required for a given plate dimension and for a given flow capacity is obtained from Equation 2.26.

$$N_P = \frac{A_s}{A_P} \tag{2.26}$$

where,

 $N_{\rm P}$  - Number of plates required to achieve separation

The effective settling length is the plate length.

There are no set design guidelines for calculation of the tank dimensions; tanks are generally sized to contain the plates. The separators are usually designed to minimize turbulence, short circuiting, and channelling of the inflow especially through and around the plate packs. To avoid plugging, the plates are spaced apart. The spacing of the plates affects the velocity of flow through the separator. Usually the plates are spaced to ensure that laminar flow conditions exist in the separation bay. The separator is usually designed to have a Reynolds number of less than 500 (laminar flow). In this research, the separator was sized to contain the plate packs and in addition to the calculated tank height 0.45 m was added to allow for sludge settling and headspace above the plate packs.

$$L_{s} = (N_{P} - 1) \times l_{dp} + 2 \times L_{P} \times \cos \alpha \quad (2.27)$$
  

$$w_{s} = w_{P} \qquad (2.28)$$
  

$$H_{s} = L_{P} \times \sin \alpha + 0.1524 + 0.3048 \quad (2.29)$$

To verify laminar flow conditions exist, the Reynolds number through the bay was calculated using Equation 2.30.

$$Re = l_{dp} \times v_p \times \rho_w / \mu_w \qquad (2.30)$$

where,

$$l_{dp}$$
 - is the distance between plates (m)

 $v_p$  - Flow velocity along plates (m/s)

 $\rho_w$  - water density (kg/m<sup>3</sup>)

 $\mu_{w}$  - water viscosity (Pa.s)

The flow velocity is given by:

$$v_p = \frac{Q}{(l_{dp} \times w_p)N_p}$$
(2.30a)

#### 2.3.5.2 Analysis of droplet capture or recovery in corrugated plate interceptor

Yao 1973 predicted that the velocity of the oil droplet size 100% theoretically removed or recovered in a corrugated plate interceptor with a flow velocity along plates ( $v_{p}$ ) can be found by using the equation below:

$$v_t = v_p \times d \ (d \sin \alpha + w_P \cos \alpha) \tag{2.31}$$

The derivation of Equation 2.31 is given in Appendix I.

Theoretically, the droplets that will be 100% recovered in a corrugated plate interceptor are those droplets with a rise rate equal or greater than the rise rate of the critical droplet. The droplets with a lower rise rate than the overflow rate will only be removed in certain proportion.

Any droplet with a rise velocity less than the critical droplet velocity predicted by Yao will be removed in the following proportions:

$$X = \frac{v_o}{v_t} \tag{2.32}$$

where,

*X* - Fraction of droplets of a particular size recovered

 $v_o$  - rise rate of any droplet (m/s)

 $v_t$  - rise rate of the critical droplet, obtained from Equation 2.31

If X = 1 then all droplets of that size are recovered and if X is less than 1, then only the proportion defined by Equation 2.32 is captured. The fraction of droplets not recovered is:

$$Y = I - X \tag{2.33}$$

#### 2.3.6 Cost modelling

A cost factor model was used in estimating the cost of the process units (Equation 2.34).

$$C_e = CS^n \tag{2.34}$$

Page 37

- $C_e$  purchased cost (£)
- *S* Characteristic size parameter in units given in Appendix I
- *C* Cost constant (see Appendix I)

Costing of the process was based on the following:

- Cost of sludge disposal from pre-treatment units was not taken into consideration and a mixed culture is assumed in the ICB to prevent sludge formation.
- For the bioprocess, only two main costs were considered: cost of the bioreactor unit and cost of electricity for aeration.
- In the costing of the pre-treatment units cost of skimmers was not taken into consideration. But cost considered includes cost of unit and cost of recovered oil.
- Cost of recovered oil is assumed to be  $\pounds 0.62/m^3$ .
- Electricity rate was assumed to be £0.05/kWh.
- The life span of each unit was assumed to be 25 years.
- The cost of energy was not taken into consideration in investigating the effect of viscosity on the cost of integrated process.
- The cost of dissolving air to increase density difference was not taken into consideration.
- The cost analysis was done on a yearly basis.

## 2.3.7 Theory of Process Integration

The process integration technique involves adopting a holistic approach to achieve certain objectives such as technical, environmental, economical and safety goals. To achieve these objectives, the approach in process integration involves understanding the effect of changes in a unit and how it affects the process as a whole. Based on this knowledge, decisions can be made.

To this end, to reduce the cost of treatment in the integrated processes, a holistic approach was employed; this method involved investigating the effect of change in a unit on the cost of the overall process.

# Chapter 3: Materials, Equipment and Methods

## 3.1 Introduction

This chapter describes the materials, methods, and experimental procedures used in the practical aspect of this research, and also describe the spreadsheet calculation procedures used for the theoretical aspect of the research.

The practical part of this research involved the investigation of aeration in various designs of a laboratory scale Immobilized Cell Bioreactor (ICB) unit. The designs were created by using one size of plastic Pall ring at a time in the ICB unit. The investigation involved determination of the mass transfer characteristics of the various sizes of plastic Pall support particles. The importance of aeration and the underlying theory are given in the Chapter 2.

The theoretical aspect of this research involved:

- Generating the various oil droplet size distributions in oil-water systems,
- Designing separation units and predicting their performance,
- Sizing the bioreactor and associated system (blower) required in each scenario
- Costing individual units and the integrated process

The exercise was done to help make judgements in seeking the most cost effective method of improving oil recovery and improving the cost effectiveness of the integrated processes. The algorithms used and calculation procedures are presented and explained.

## 3.2 Materials

The fluids used in the experimental aspect of this research were nitrogen gas, compressed air, and water. Nitrogen was used to strip oxygen in solution. Plastic Pall rings were used as support particles or packing. The physical data of the plastic Pall rings are given in Table 3.1.

### Investigation of Oily Wastewater Treatment Processes

Size (mm)	Weight (kg/m <sup>3</sup> )	Surface Area m <sup>2</sup> /m <sup>3</sup>		
16	115	340		
25	80	196		
38	60	150		
50	60	106		

Table 3.1: Physical data of plastic Pall rings

The equipment used in the experiment were an oxygen probe/meter, a stop watch, and a rotameter.

### 3.2.1 Oxygen probe and meter

A calibrated polarographic automatic temperature-compensated oxygen electrode (Mettler Toledo probe/Mettler Model IP67) and meter (Mettler Toledo MO128) was used to monitor and read the variation of the dissolved oxygen (DO) concentration with time in the system.

### 3.2.1.1 Oxygen probe calibration

To ensure accuracy in probe readings, the oxygen probe was calibrated using the 2 point method.

The 2-point method involves calibrating the probe at 2 points, 100% saturation, and 0% saturation of oxygen. For the 100% calibration, the probe was calibrated in air and the output current was set when the meter read 100%. For the second calibration point (0%), the probe was dipped in a magnetically stirred beaker of nitrogen-saturated water; the output current was set when the meter read 0%. The 0% reading corresponded to zero oxygen concentration in the solution.

Once the probe was calibrated, the response time of the probe was determined by quickly moving the probe from an oxygen-rich solution to a nitrogen-rich solution and noting the time it took to response to change. The experiment was repeated in the reverse direction, moving the probe from a nitrogen-rich solution to an oxygen-rich solution. A short response time was observed, less than 1 second, and hence the probe response was not taken into consideration in  $k_{La}$  estimation.

### 3.2.2 Rotameter

A rotameter equipped with a control valve was used in measuring the airflow into the reactor. The rotameter was calibrated using the water displacement method.

The water displacement method involves using air collected from the column to displace a known volume of water and noting the time it takes to displace the water.

To obtain the equivalent flow rate for each observed scale on the rotameter, the air flow was set to each observed scale by adjusting the control valve; compressed air passed into the column at each scale reading was collected via tubing from the tightly sealed column. The air was then used in displacing a known volume of water, noting the time it took to achieve this. These values were then converted to volumetric air flow rates.

This procedure was repeated for each observed scale reading on the rotameter and a calibration graph was plotted (see Appendix II).

## 3.3 Experimental set-up

The experimental set-up is represented in Figure 3.1. The experiments were carried out in a glass cylindrical column with a tapered bottom. The total height of the column was 0.83 m, inner diameter 0.17 m and outer diameter 0.19 m. For each ICB unit design, the column was packed randomly with plastic Pall rings of a particular size; these plastic Pall rings were fixed in place by wire meshes at the top and bottom (Figure 3.1). Mains tap water was used to fill the column. The column was operated in a semi-batch mode: with respect to water it was operated in a batch mode and with respect to air it was operated in a continuous mode. The probe was located well above the packing height at an angle. This was done for two reasons. Firstly, the probe requires a high liquid velocity for proper operation as the output current depends on liquid velocity (Silver 1965); the second reason is to prevent air bubbles resting on the membrane. Votruba et al. 1976 and Votruba et al. 1977 reported that when air bubbles rest on the probe membrane, the probe measures a concentration between the equilibrium value and the bulk value, rather than the bulk concentration. The gas sparger was located below the mesh to ensure satisfactory gas distribution. All measurements were carried out at room temperature and atmospheric pressure.



Figure 3.1: The experimental set-up for mass transfer measurements

### 3.4 Method

#### 3.4.1 Determination of $k_{La}$

The static gas method was used to measure the change in dissolved oxygen with time. This involved stripping the water of oxygen by sparging with nitrogen and then sparging with oxygen at a steady flowrate while taking measurements of the dissolved oxygen concentration every ten seconds until the solution became saturated. The change in dissolved oxygen concentration was measured with and without packing. Oxygen concentration measurements were taken well above the height of the support particles, an ideal mixing of the liquid phase was assumed in that location.

In order to observe the effect of aeration on  $k_{La}$  in the various ICB designs (each filled with a support particle of a particular size), the aeration rate was varied between 8 ml/s and 57 ml/s for each ICB design. It was ensured that the total measurement time was long enough to obtain the oxygen saturation concentration,  $C^*$ . The value of C (oxygen concentration in the solution) was read off from the oxygen probe meter at equal intervals. Values for  $k_{La}$  were then determined for each aeration rate investigated in each ICB design using Equation 3.0.

$$dC/dt = k_{La}(C^*-C) \tag{3.0}$$

- $C^*$  the saturation value of DO at the gas-liquid interface (% saturation)
- C Concentration of oxygen in the bulk liquid phase (% saturation)
- $k_{La}$  volumetric liquid phase mass transfer coefficient (s<sup>-1</sup>)

## 3.5 Calculation procedure for generating oil droplet distribution

To generate the oil droplet distribution, a spreadsheet using Equation 2.2 was employed. The spreadsheet calculation procedure is presented in Figure 3.2.

The parameters specified in the spreadsheet calculation are flow rate, oil concentration, average oil density, droplet sizes, mean droplet size of the oil-water mixture, the standard deviation from the mean and an assumed number of droplets in the first instance. The spreadsheet uses these parameters to further define the influent oil water mixture characteristics, determine the number of droplets less than a particular size, and calculate the volume of oil contained by all droplets of a certain size.

### 3.5.1 Influent oil characteristics

To calculate the mass flow rate of the oil in the influent the spreadsheet uses the specified oily wastewater flow rate and oil concentration. This calculation is necessary to determine the volume of oil in the oil-water mixture. The spreadsheet uses the calculated mass flow rate and the specified average oil density to calculate the volume of oil in the mixture. Table 3.2 is an example of values generated as a result of specifying the required input data. These results are used in generating the oil droplet size distribution and the volume of oil in each size interval.

Oily Wastewater oil droplet size distribution										
Influent characteristics										
Water flo	owrate	10	m³/h							
Oil concentration Average oil density		1000 800	mg/l kg/m³							
Average Standard	droplet dia I deviation	60 15	µm -							
Calculations										
Mass of oil in influer 10 kg/h Volume of oil in inflւ 0.0125 m³/h										
No. of droplets in in 9.25E+10 - Solver variable										
Droplet size	Minimum	Maximum	Mean	Cumulative distribution	Fraction of	Number of droplets	Volume of droplets in interval			
interval (1)	size (µm) (2)	droplet size (µm) (3)	droplet size (µm) (4)	function (5)	droplets in interval (6)	in interval (7)	interval (m3) (8)			
2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 17	10 20 30 40 50 60 70 80 90 100 110 120 130 140 150 160	20 30 40 50 60 70 80 90 100 110 120 130 140 150 160 170	15 25 35 45 55 65 75 85 95 105 115 125 135 145 155	0.00383 0.02275 0.09121 0.25249 0.50000 0.74751 0.90879 0.97725 0.99617 0.99957 0.99997 1.00000 1.00000 1.00000 1.00000	0.00340 0.01892 0.06846 0.16128 0.24751 0.16128 0.06846 0.01892 0.00340 0.00040 0.00003 0.00000 0.00000 0.00000 0.00000	3.147E+08 1.751E+09 6.334E+09 1.492E+10 2.290E+10 2.290E+10 1.492E+10 6.334E+09 1.751E+09 3.147E+08 3.677E+07 2.789E+06 1.372E+05 4.370E+03 9.007E+01 1.200E+00	5.561E-07 1.432E-05 0.0001422 0.0019949 0.0032929 0.0032963 0.0020368 0.0007859 0.0001908 2.928E-05 2.852E-06 1.767E-07 6.975E-09 1.756E-10 2.823E-12			
18 19 20 21 22 23 24 25 26 27 Totals	170 180 190 200 210 220 230 240 250 260	180 190 200 210 220 230 240 250 260 270	175 185 205 215 225 235 245 255 265	1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000 1.00000	0.00000 0.00000 0.00000 0.00000 0.00000 0.00000 0.00000 0.00000 0.00000 1.00000	1.032E-02 0.000E+00 0.000E+00 0.000E+00 0.000E+00 0.000E+00 0.000E+00 0.000E+00 0.000E+00 9.252E+10	2.897E-14 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0			

Difference between calculated oil volumes

1E-06 - use Solver to make it zero.

 Table 3.2: Oil droplet size distribution





## 3.5.2 Droplet size distribution

By specifying a minimum (column 2) and maximum (column 3) droplet size, the droplet sizes are classified into equal intervals (see Table 3.2). The spreadsheet first calculates the mean droplet size (4) of each interval, which is then used to represent the droplets sizes in that interval and also for calculating the volume of oil contributed by each interval (8).

Based on a specified mean droplet size for the oil-water mixture and its standard deviation, the spreadsheet calculates the fraction of droplets less than the maximum droplet size (5) in each interval using the cumulative distribution function in Excel. This calculation aids in determining the fraction of droplets in each interval (6). The fraction of droplets in each interval in column six is calculated by subtracting the cumulative distribution fraction of the previous interval from the current cumulative distribution fraction, except in the first interval where the fraction of droplets in the interval is equal to the cumulative distribution function.

To obtain the number of droplets in an interval in column (7), the spreadsheet multiplies the assumed number of oil droplets in the oil-water mixture by the fraction of droplets in each interval in column (6). Figure 3.2 shows the spreadsheet calculation procedure.

### 3.5.3 Calculation of oil volume

To calculate the volume of oil contributed by droplets in each interval (column 8) the spreadsheet multiplies the calculated number of oil droplets in each interval (column 7) by the volume of oil contributed by the mean droplet size of each interval (see Equation 2.1 and column 4 in Table 3.2).

The spreadsheet compares the total volume of oil contributed by each interval (sum of column 8) with the volume of oil in the influent to check if the assumed total number of droplets was correct. If the summed oil volume is not equal to the calculated influent oil volume, the Excel solver function is used to reduce to zero the difference between the calculated oil volume and the summed oil volume by varying the assumed number of oil droplets in the oil-water mixture (see Table 3.2 and Figure 3.2). When these operations have been carried out the spreadsheet then presents the droplet size distribution of the oil-water mixture. The droplet size distribution can then be plotted using the probability mass function in Excel.

## 3.6 Calculation procedure for design of gravity separators

The spreadsheet uses specified values of flow rates, design droplet velocity, and assumed values of  $R_1$  (see Equation 2.20b) and  $R_2$  (see Equation 2.21d) to obtain the effective separation area, horizontal flow velocity, and dimensions of the separator. The spreadsheet calculation procedure is presented in Figure 3.3.

In the first step, the spreadsheet calculates the separation area required for separation of the oil-water mixture, based on the design droplet velocity and the specified flow rate of the oil water mixture.

Using the assumed  $R_1$  and the specified flow rate, the spreadsheet calculates the horizontal flow velocity using Equation 2.20b. The spreadsheet then compares the calculated horizontal flow velocity (Figure 3.3) with the guidelines set by the American Petroleum Institute (API) (see Equations 2.20a and 2.20b) and chooses the minimum of the two values.

The spreadsheet then uses the calculated horizontal velocity and the specified flow rate to calculate the cross sectional area of the separator using Equation 2.21.

The spreadsheet then calculates the dimensions (depth and width) of the cross sectional area of the separator using Equations 2.21e and 2.21a and compares the calculated width with the constraints set up by the API (Equation 2.21c). The parameters used in the calculation of the width are the calculated horizontal flow velocity, specified flow rate and an assumed  $R_2$  (Equation 2.21d).  $R_2$  is the variable that has to be changed if the calculated width is not within the limits of the constraints. The spreadsheet then calculates the separator length using Equation 2.21f.

The calculated separator dimensions are then compared to the constraints defined by API (Equations 2.21b, 2.21c, and 2.21g). If the calculated dimensions do not satisfy the imposed constraints then another value of  $R_2$  must be entered and the process repeated again until the separator dimensions satisfy the constraints.

Investigation of Oily Wastewater Treatment Processes





Page 48

#### 3.6.1 Predicting separator performance

The spreadsheet calculation procedure for oil droplet removal is presented in Figure 3.4. To analyze the droplets removed or recovered, separator dimensions, residence time of the oily wastewater stream in the separator, droplet distribution in the influent stream, and oil-water influent characteristics were specified. Using these values, the spreadsheet analyzes which droplets can be recovered from the oily wastewater stream before it reaches the exit of the separator. This analysis was done using the principle of unhindered settling (see section 2.3.4.2). The droplet size removal analysis was based on Equation 2.22. The spreadsheet calculation procedure is described below.

The spreadsheet first calculates the terminal velocity of each droplet size using Equation 2.13 and then calculates the height travelled by each droplet size based on the residence time of the oil-water mixture in the separator (see Equation 2.22a). The height travelled by each droplet size is then compared to the separator depth. If the height travelled by any droplet of that size is greater than the separator depth then the analysis assumes all droplets of that size are recovered or removed and the spreadsheet displays a value of 1; if the height travelled by any droplet in the given time is less than the separator depth, the spreadsheet then assumes only a fraction of that droplet size can be recovered. The fraction recovered is calculated using Equation 2.22 and the un-recovered fraction is calculated using Equation 2.23. The volume of oil recovered and unrecovered is then calculated using Equation 2.1.

Investigation of Oily Wastewater Treatment Processes



Figure 3.4: Calculation procedure for oil droplet removal

## 3.7 Calculation procedure for design of CPI

For the design of a Corrugated Plate Interceptor, the spreadsheet uses specified values of oil influent characteristics, design droplet velocity, and plate characteristics such as plate length or settling length, plate width, distance between plates, and angle of inclination of plates (Figure 3.5).

The first step in designing the Corrugated Plate Interceptor involves calculating the required separation area to achieve a given separation. The spreadsheet calculates this area using Equation 2.24 and an assumed design factor of 12%.

To obtain the number of plates required to achieve the effective separation area the spreadsheet uses Equation 2.25 to calculate the projected area of one plate and then uses Equation 2.26 to determine the number of plates required.

The dimensions of the separator are then calculated using Equations 2.27, 2.28 and 2.29. The flow velocity along the plates is also calculated using 2.30a.

The Reynolds Number is then calculated using Equation 2.30 to ensure laminar flow conditions are maintained in the interceptor.



Figure 3.5: Calculation procedure for design of plate interceptors

### 3.7.1 Predicting CPI performance

To analyze droplet capture in the Corrugated Plate Interceptor, separator dimensions, droplet sizes, plate characteristics, oil-water influent characteristics are specified for the spreadsheet calculations (Figure 3.6). The droplet removal analysis is based on Equation 2.31. Only droplets with a velocity greater than the rise rate of the critical oil droplet defined by Equation 2.31 are completely removed. Oil droplets with rise rates less than the rise rate of the critical droplet are removed in certain proportions. The fraction of droplets recovered is calculated by the spreadsheet using Equation 2.32. The spreadsheet calculation procedure is described below.

The spreadsheet first calculates the terminal velocity of the mean droplet size in each droplet size interval using Equation 2.13; it then compares the terminal velocity of the mean droplet in each interval with the velocity of the (critical) droplet which is 100% theoretically recovered (Equation 2.31). If velocity of rise of any oil droplet is greater or equal to the critical droplet velocity as defined by Equation 2.31, then the spreadsheet displays a value of 1; if any oil droplet's velocity is less than the critical droplet's velocity then only a fraction of the droplets of that size are recovered. The spreadsheet calculates the recovered fraction using Equation 2.32 and the unrecovered fraction is calculated using Equation 2.33.

The volume of oil recovered from each size interval is then summed to obtain the total volume of oil recovered per hour from each volume of oily wastewater treated. The spreadsheet calculation procedure is presented in Figure 3.6.



Figure 3.6: Calculation procedure for oil droplet removal in a CPI

## 3.8 Bioreactor design procedure

To calculate the bioreactor size required to treat a particular oil concentration in an oily wastewater stream, the spreadsheet requires specified values of wastewater flowrate, inlet oil concentration, the molecular formula of the oil, water density, average oil density, and mass transfer coefficient (Figure 3.7).

The spreadsheet first determines the oxygen requirement of the oil-water mixture using Equation 2.3. The specified parameter for this calculation is the oil concentration of the oil-water mixture.

The next set of calculation procedures after the oxygen requirement has been determined is the calculation of the oxygen transfer rate (Equation 2.9a) and the bioreactor size (2.9). For the calculation of the oxygen transfer rate, the specified inputs are the mass transfer coefficient, maximum oxygen solubility in water at 25°C, and the percentage saturation of oxygen in the bioreactor. Based on the oxygen transfer rate the spreadsheet calculates the bioreactor size to match oxygen supply to oxygen demand.

### 3.8.1 Power requirements of the blower

The spreadsheet estimates the blower horsepower requirements based on a calculated air volume value and an estimated discharge pressure (Figure 3.8).

In this research, the blower discharge pressure was estimated based on an assumed pressure loss in the aeration equipment, main piping, and diffuser.

First the spreadsheet calculates the air volume required to meet oxygen demand using Equation 2.11. Using the specified discharge pressure, mechanical efficiency, and calculated air volume the spreadsheet calculates the power requirements using Equation 2.10.



Figure 3.7: Bioreactor design calculation procedure


Figure 3.8: Blower size calculation procedure

# Chapter 4: Results and discussion - Aeration in various ICB models

## 4.1 Introduction

This chapter reports the results of aeration in various laboratory scale ICB models. Using the column described in Chapter 3, various models of laboratory scale ICB unit were designed by using various sizes of plastic Pall support particles. The mass transfer characteristics of various sizes of plastic Pall support particles in the column were investigated using the methods previously described in Chapter 3 and are presented in Figures 4.1- 4.4. The data used for the various charts are presented in Appendix II. This investigation was necessary to determine the effectiveness of various designs of ICB units in terms of aeration.

## 4.2 Results

The  $k_{La}$  values used in Figures 4.1 to 4.4 were determined using methods described in Section 3.4.

Figures 4.1 to 4.4 show the effect of aeration rate on volumetric mass transfer coefficient for the various support particle sizes. For each size of support particle investigated, the volumetric mass transfer coefficient increased as aeration rate increased. Figure 4.1 shows the effect of aeration rate on volumetric mass transfer coefficient for the column filled with 16 mm plastic Pall support particles. The volumetric mass transfer coefficient increased from 0.002 to 0.004 s<sup>-1</sup> as aeration rate increased from 9 to 48 ml/s. For the ICB unit filled with 25 mm plastic Pall support particles, the volumetric mass transfer coefficient increased from 0.002 to 0.0043 s<sup>-1</sup> as aeration rate increased from 13 to 56 ml/s (Figure 4.2). The volumetric mass transfer coefficient of the ICB unit filled with 38 mm plastic Pall support particles, increased from 0.003 to 0.0097 s<sup>-1</sup> as aeration rate increased from 9 to 47.6 ml/s (Figure 4.3). For the ICB unit filled with 50 mm plastic Pall support particles (Figure 4.4), the volumetric mass transfer coefficient increased from 0.0015 to 0.0058 s<sup>-1</sup> as aeration rate increased from 10.5 to 50 ml/s.



Figure 4.1: Oxygen MTC vs. aeration rate in ICB model with 16 mm Pall rings



Figure 4.2: Oxygen MTC vs. aeration rate in ICB model with 25 mm Pall rings



Figure 4.3: Oxygen MTC vs. aeration rate in ICB model with 38 mm Pall rings



Figure 4.4: Oxygen MTC vs. aeration rate in ICB model with 50 mm Pall rings

## 4.3 Discussion

For each of the ICB models, volumetric mass transfer coefficient increased with aeration rate. This was as expected. Nikakhtari and Hill (2005) investigated the mass transfer characteristics of a column with support particles and one without support particles and reported an increase in mass transfer coefficient as superficial air velocity increased in both columns and also observed a higher oxygen mass transfer coefficient in the column with support particles.

The support particles improve the mass transfer characteristics of the column in two ways. Firstly, they break up large air bubbles to smaller bubbles and hence increase the surface area available for mass transfer. Secondly, they prevent larger gas bubbles from escaping rapidly from the column by breaking them into smaller bubbles which have slower rise velocities (Stokes' Law); in other words, the support particles increase the residence time of the air bubbles and hence allow for more oxygen transfer. Nikakhtari and Hill (2005) also reported a partial drop in interstitial velocity of the air bubbles in the presence of support particles.

Increasing aeration rate increases the frequency of bubble formation, therefore increasing gas hold-up and consequently increasing oxygen mass transfer rate. Gas hold-up measurements were not carried out in these investigations but various studies have confirmed this phenomenon (Nikakhtari and Hill 2005, Bhatia *et al.* 2004).

The improvement in mass transfer characteristics as aeration rates increased for each support particle size may also be attributed to increased liquid circulation. Increasing liquid circulation decreases the liquid film thickness and hence improves mass transfer. It can be explained that as aeration rate increased the liquid film thickness reduced therefore improving mass transfer.

Mixing studies were not performed to ascertain the liquid circulation rate. The mixing time for each support particle size would have been indicative of the liquid circulation rate.

#### 4.3.1 Comparing aeration in various ICB models

The 38 mm packing had the best mass transfer characteristics compared to the other packing sizes. This was concluded because of the faster rate of rise of its curve compared

#### Investigation of Oily Wastewater Treatment Processes

to the other curves. The 50 mm support particle had better mass transfer characteristics than the 25 mm support particle at higher aeration rates (at about 40 ml/s).

The 16 mm support particles showed the least mass transfer capabilities. Ostergaard (1978) reported better mass transfer characteristics with 6 mm glass ballotini when compared to 1 and 3 mm glasses. This trend was not expected in this investigation considering the internal structure of the support particles (void space) (Figure 4.5 and Figure 4.6).



Figure 4.5: Internal structure of the 50 mm support particle



Figure 4.6: The various sizes of packing (mm): 50, 38, 25, 16 (from left to right)

The smallest support particle has the smallest void area (Figure 4.5 and Figure 4.6), so it was expected to have a higher probability of disintegrating large bubbles than the bigger support packing (38 mm and 50 mm), and hence better mass transfer capabilities were expected. However, this was not so. The probable reason for the poor mass transfer characteristics of the smaller support particles may be as a result of poor gas circulation due to low bed voidage. Considering that the various ICB models were operated in a semi-batch mode (liquid movement was solely due to gas flow) it follows that liquid circulation was poor when the column was filled with smaller particles (16 mm) due to poor gas circulation or distribution. For better mass transfer, good liquid circulation is needed to reduce the liquid film thickness.

Daraktschiev (1984) characterized the gas flow distribution in a packed bed and reported that particle arrangement affects gas distribution. In this investigation, it was suspected that not only had the arrangement affected the mass transfer but also the number of support particles in the bed. Since the height of the support particles in the column was fixed by the position of the wire mesh the bigger support particles resulted in a higher bed voidage (due to their internal structure and the smaller number of particles in the volume provided for the particles) while the smaller support particles had a low bed voidage due to a higher number of support particles. It can be concluded that the size and number of

particles in the fixed bed height are the reasons for the poor mass transfer characteristics of the smaller support particles.

Comparing the mass transfer capabilities of the 50 mm support particle and the 38 mm support particle, the expected trend was observed. The 38 mm support particles had better mass transfer characteristics than the 50 mm support particles. This trend was expected due to the internal structure of the support particles. The void spaces in the 50 mm support particle were very large, so the probability of disintegrating gas bubbles were less as a result a poor mass transfer characteristics when compared to the 38 mm support particle. However, good liquid circulation was observed with both support particle sizes.

Comparing the mass transfer characteristics of the smaller 16 mm support particle and the 25 mm support particle, the 25 mm support particle showed better mass transfer characteristics. The probable reasons may be as discussed earlier, poor liquid circulation in the ICB column filled with 16 mm support particle.

## 4.4 Conclusions

In order to reduce cost due to aeration, aeration in different designs of ICB models was investigated. These results show that design affects aeration. Design of the ICB includes the type, arrangement, and number of support particles in the unit. The 38 mm plastic Pall ring had the best mass transfer characteristics and may be said to be the particle size which best improves the cost of aeration; other factors also contribute to it, such as arrangement and internal structure. These results do not agree with those of Ostergaard 1978. Ostergaard 1978 investigated mass transfer characteristics of various sizes of glass ballotini and reported that the largest glass size had the best mass transfer characteristics. Lee and Buckley (1981) also reported that large diameters such as 4 mm and 6 mm are beneficial to aeration. One thing remains clear: the support particle size affects aeration.

In conclusion, these studies have shown that the physical structure of the ICB (or design) affects aeration and the solution to reducing cost due to aeration may lie in the design of the ICB unit rather than the use of oxygen *in situ*.

## Chapter 5: Results and Discussion – Optimizing Oil Recovery

## 5.1 Introduction

This chapter presents the results obtained from various scenarios where the effectiveness of oil recovery units and their effects on the integrated process were investigated.

The overall aim of this chapter is to provide a solution to improving oil recovery and reducing the cost of treatment in the integrated processes studied.

Section 5.2 presents the results of the various scenarios studied to investigate the effectiveness of the horizontal flow separator to recover oil and its effect on the integrated process, followed by an analysis and a discussion sub-section.

Section 5.3 presents the results of the various scenarios studied to investigate the effectiveness of the Corrugated Plate Interceptor to recover oil and its effect on the integrated process, followed by an analysis and a discussion sub-section.

In the presentation of these results, "bioprocess" refers to the bio-treatment and aeration, which in this case is achieved with an ICB and a blower; "separation process" refers to the gravity separation unit used in each integrated process and the skimming of the oil recovered. The overall cost of the plant is the cost of the plant less the revenue generated from recovered oil. The unit cost of oil was an estimated value (see Chapter 2).

## 5.2 Integrated Process of Horizontal Flow Separator and ICB

A design exercise was carried out using the models previously described (see Sections 2.3.1, 2.3.2, 2.3.4, 2.3.5, 3.6, and 3.7). Using various simulated oily wastewater systems, the effects of varying influent characteristics, such as oil concentration and average oil droplet size, on the cost per year of a horizontal separator and oil recovery per hour were studied. The effects of other factors associated with design of the separator were also investigated. For each factor investigated, a volume of oily water was fed into the separator and the performance predicted (see Section 3.6.1). For each case, the size of Immobilized Cell Bioreactor (ICB) required to treat the effluent from the separator was calculated based on the oxygen demand of the effluent (see Section 2.3.2.2). The costs of constructing and operating the separator and the ICB were then calculated in each case.

The separator dimensions, influent characteristics, value of oil recovered, and cost of each unit are presented in Appendix III.

## 5.2.1 Effect of varying influent oil concentration

The effectiveness of the horizontal flow separator to recover oil from influent streams with various oil concentrations was investigated using the models previously described in Chapters 2 and 3. Several values of oil concentration were used whilst holding other influent characteristics and separator dimensions constant. Separator dimensions, influent characteristics, and results are presented in Appendix III. The cost of the treatment is presented below.

Figure 5.1 shows the effect of influent oil concentration on the cost of separation. As the influent oil concentration is increased from 1000 to 1500 mg/l, the amount of oil treated increases from 40 to 60 tonnes per year and the cost of separation decreases from £35.53 to £31.01 per year. This decrease in cost is as a result of the increased revenue due to the greater volume of oil recovered. The volume of oil recovered increases from 0.0036 to 0.0055 m<sup>3</sup>/h (Figure 5.2), and the revenue from recovered oil increases from £9.03 to £13.55 per year (Figure 5.3) as the inlet oil concentration increases.

Figure 5.4 shows the relationship between oil concentration and separator efficiency. Increasing the influent oil concentration by does not affect separation efficiency; efficiency remains constant at 29%.

The relationships between oil concentration and oxygen demand, bioreactor volume and cost of the bioprocess are presented in Figures 5.5, 5.6 and 5.7. As oil concentration in the separator influent is increased, the amount of oxygen required to oxidize the oil arriving at the bioreactor increases from 2.46 to 3.69 g/l.h; consequently the bioreactor size required for complete oxidation increases from 113 to 170 m<sup>3</sup>, (Figures 5.5 and 5.6). The effect of oil concentration on the cost of the integrated process is shown in Figure 5.8. As oil concentration is increased, the cost of the integrated process increases from  $\pounds$  5189.25 to  $\pounds$ 7648.56 per year.



Figure 5.1: Effect of oil concentration on cost of separation



Figure 5.2: Effect of oil concentration on oil recovery

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.3: Effect of oil concentration on revenue generated from recovered oil



Figure 5.4: Effect of oil concentration on separator efficiency

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.5: Effect of oil concentration on oxygen demand



Figure 5.6: Effect of oil concentration on bioreactor volume

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.7: Effect of oil concentration on cost of bioprocess



Figure 5.8: Effect of oil concentration on cost of integrated process

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).

## 5.2.2 Effect of varying separator horizontal velocity

The effectiveness of using a horizontal flow separator operating at various horizontal velocities was investigated using the models previously described in Chapters 2 and 3. Horizontal flow velocity was varied whilst holding separator dimensions and influent characteristics constant. Separator dimensions, influent characteristics, and results are presented in Appendix III. The cost of treatment is presented below.

Figure 5.9 shows the effect of horizontal flow velocity of the separator on the cost of separation. As the horizontal flow velocity is reduced from 0.0049 to 0.0005 m/s, the cost of separation decreases from £35.53 to £13.69 per year. Figures 5.10 and 5.11 show that as horizontal velocity is reduced, the volume of oil recovered increases from 0.004 to 0.012 m<sup>3</sup>/h and hence the revenue generated from recovered oil increases from £9.03 to £30.87 per year. Figure 5.12 shows the relationship between horizontal velocity in the separator and separator efficiency. A 90% decrease in horizontal velocity improves the efficiency of the separator by 242% while a 50% decrease in horizontal velocity improves the efficiency by 98%.

Relationships between horizontal velocity and oxygen demand, bioreactor volume and cost of bioprocess are presented in Figures 5.13, 5.14 and 5.15. As horizontal velocity in the upstream separator is reduced, oxygen demand decreases from 2.46 to 0.01 g/l.h and bioreactor volume decreases from 113 to 1 m<sup>3</sup> (see Figures 5.13 and 5.14). Decreasing horizontal velocity therefore reduces both the cost of the separator and the bioprocess. In Figure 5.15, as horizontal velocity is reduced, the cost of the bioprocess decreases from  $\pounds$  5153.72 to  $\pounds$ 43.70 per year. The overall effect of decreasing horizontal velocity on the integrated process is shown in Figure 5.16, where reducing horizontal velocity decreases the cost of the integrated process from  $\pounds$ 5189.25 to  $\pounds$ 57.39 per year.



Figure 5.9: Effect of horizontal velocity on cost of separation



Figure 5.10: Effect of horizontal velocity on oil recovery

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.11: Effect of horizontal velocity on revenue generated from recovered oil



Figure 5.12: Effect of horizontal velocity on separator efficiency

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.13: Effect of horizontal velocity on oxygen demand



Figure 5.14: Effect of horizontal velocity on bioreactor volume

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.15: Effect of horizontal velocity on cost of bioprocess



Figure 5.16: Effect of horizontal velocity on cost of integrated process

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).

## 5.2.3 Effect of varying separator volume

The effectiveness of using horizontal flow separators of various volumes to treat an oily wastewater stream was investigated using the models previously described in Chapters 2 and 3. Separator length was varied whilst holding other dimensions and influent characteristics constant. Separator dimensions, influent characteristics, and results are given in Appendix III. The cost of treatment is presented.

Figure 5.17 shows the effect of separator volume on the cost of separation. As separator volume is increased from 4 to 47 m<sup>3</sup>, the cost of separation increases from £35.53 to £162.13 per year, despite the fact that oil recovery increases from 0.004 to 0.012 m<sup>3</sup>/h (Figure 5.18) and the revenue generated increases from £9.03 to £30.92 per year, see Figure 5.19. Figure 5.18 shows that beyond a certain size, increase in separator volume does not result in a significant improvement in oil recovery; this may be attributed to the presence of very small oil droplets which have very slow rise rates.

Figure 5.20 shows the relationship between separator volume and separator efficiency. Increasing separator volume results in a significant improvement in efficiency but beyond a volume increase of about 400%, the increase in separator efficiency, and hence improvement in oil recovery, is not significant.

Figures 5.21, 5.22, 5.23 show the relationships between the volume of the upstream separator and oxygen demand, bioreactor volume and cost of the bioprocess. As separator volume is increased, oxygen demand decreases from 2.46 to 0.01g/l.h (Figure 5.21) and bioreactor volume decreases from 113 m<sup>3</sup> to 400 litres (Figure 5.22); the cost of the bioprocess decreases from £5153.72 to £28.97 per year, Figure 5.23. From Figure 5.23, increasing separator volume beyond a certain size (about 23 m<sup>3</sup>) does not result in a significant reduction in the cost of the bioprocess; optimization therefore becomes very necessary.

The effect of separator size on the cost of the integrated process is shown in Figure 5.24. As separator volume is increased, the cost of the separation process increases but the cost of the integrated process decreases from £5189.25 to £191.09 per year.



Figure 5.17: Effect of separator volume on cost of separation



Figure 5.18: Effect of separator volume on oil recovery

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.19: Effect of separator volume on revenue generated from recovered oil



Figure 5.20: Effect of separator volume on separator efficiency

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.21: Effect of separator volume on oxygen demand



Figure 5.22: Effect of separator volume on bioreactor volume

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.23: Effect of separator volume on cost of bioprocess



Figure 5.24: Effect of separator volume on cost of integrated process

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).

## 5.2.4 Effect of varying average droplet size

The effectiveness of using a horizontal flow separator to recover oil from oily wastewater streams with varying average droplet sizes was investigated using the models previously described in Chapters 2 and 3. The average oil droplet size was varied whilst holding other influent characteristics and separator dimensions constant. Separator dimensions, influent characteristics, and results are presented in Appendix III. The cost of treatment is presented below.

Figure 5.25 shows the effect of average droplet size on cost of the separation. As average droplet size is increased from 60 to 150 microns the cost of separation decreases from  $\pounds$ 35.53 to £13.78 per year; oil recovery increases from 0.0036 to 0.0124 m<sup>3</sup>/h (Figure 5.26) and hence the revenue generated from recovered oil increases from £9.03 to £30.78 per year (Figure 5.27). Figure 5.28 shows that as average droplet size is increased from 60 to 150 microns the base separator efficiency increases from 29 to 99%.

The effect of droplet size on the oil oxidation part of the process is presented in Figures 5.29, 5.30 and 5.31. In Figure 5.29, as average droplet size is increased oxygen demand decreases from 2.46 to 0.02 g/l.h; the volume of the bioreactor reduces from 113 m<sup>3</sup> to 1 m<sup>3</sup> (Figure 5.30) and hence the cost of the bioprocess decreases from £5153.72 to £67.25 per year (Figure 5.31). Figure 5.32 shows the effect of average droplet size on the cost of the integrated process. As average droplet size is increased, the cost of the integrated process decreases from £5189.25 to £81.03 per year.



Figure 5.25: Effect of average droplet size on cost of separation



Figure 5.26: Effect of average droplet size on oil recovery

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.27: Effect of average droplet size on revenue generated from recovered oil



Figure 5.28: Effect of average droplet size on separator efficiency

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.29: Effect of average droplet size on oxygen demand



Figure 5.30: Effect of average droplet size on bioreactor volume

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.31: Effect of average droplet size on cost of bioprocess



Figure 5.32: Effect of average droplet size on cost of integrated process

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).

## 5.2.5 Effect of varying wastewater viscosity

The effectiveness of using a horizontal flow separator to recover oil from oily wastewater streams with varying water viscosities was investigated using the models previously described in Chapters 2 and 3. The average droplet size was varied whilst holding other influent characteristics and separator dimensions constant. Separator dimensions, influent characteristics, and results are presented in Appendix III. The cost of treatment is presented below.

Figure 5.33 shows the effect of oily wastewater viscosity on the cost of the separation. As viscosity is decreased from 0.0010 to 0.0003 Pa.s, the cost of separation decreases from £35.53 to £18.11 per year; oil recovery increases from 0.0036 to 0.0106 m<sup>3</sup>/h (Figure 5.34). Figure 5.35 shows that the value of recovered oil increases from £9.03 to £26.45 per year as viscosity is reduced. The effect of oily wastewater viscosity on separator efficiency is shown in Figure 5.36; separator efficiency increases from 29% to 85% as oily wastewater viscosity is decreased.

The effect of varying oily wastewater viscosity on the oil oxidation process is shown in Figures 5.37, 5.38 and 5.39. In Figure 5.37, as viscosity is decreased, oxygen demand decreases from 2.46 to 0.51 g/l.h and hence bioreactor volume and cost of the bioprocess decrease from 113 to 23 m<sup>3</sup> and from £5153.72 to £1079.17 per year in Figures 5.38 and 5.39 respectively.

Figure 5.40 shows the effect of varying oily wastewater viscosity on the cost of the integrated process. As viscosity is decreased, the cost of the integrated process decreases from £5189.25 to £1097.17 per year. It therefore follows that reducing water viscosity minimizes the cost of the integrated process.



Figure 5.33: Effect of wastewater viscosity on cost of separation



Figure 5.34: Effect of wastewater viscosity on oil recovery

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.35: Effect of wastewater viscosity on revenue generated from recovered oil



Figure 5.36: Effect of wastewater viscosity on separator efficiency

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.37: Effect of wastewater viscosity on oxygen demand





(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.39: Effect of wastewater viscosity on cost of bioprocess



Figure 5.40: Effect of wastewater viscosity on cost of integrated process

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).

#### 5.2.6 Effect of varying oil-water density difference

The effectiveness of using a horizontal flow separator to recover oil from oily wastewater streams with various density differences between the oil and water phases was investigated using the models previously described in Chapters 2 and 3. The density difference was varied whilst holding other influent characteristics and separator dimensions constant. Separator dimensions, influent characteristics, and results are presented in Appendix III. The cost of treatment is presented below.

Figure 5.41 shows the effect of density difference on the cost per year of separation.' As density difference is increased from 200 to 235 kg/m<sup>3</sup>, the cost of separation decreases from £35.53 to £33.95 per year. The decrease in cost of separation is due to the increased oil recovery. In Figure 5.42, oil recovery increases from 0.0036 to 0.0043 m<sup>3</sup>/h and hence the cost of separation is reduced by revenue generated from the recovered oil; the revenue generated increases from £9.03 to £10.61 per year (Figure 5.43). Figure 5.44 shows the effect of density difference on the efficiency of the separator. Efficiency increases from 29% to about 34% as density difference is increased.

The effect of density difference on the cost of the oil oxidation process is presented in Figures 5.45, 5.46 and 5.47. As density difference is increased, oxygen demand decreases from 2.46 to 2.28 g/l.h; bioreactor volume decreases from 113 to 105 m<sup>3</sup> and the cost of the bioprocess decreases from £5153.72 to £4796.47 per year. The effect on the cost of the integrated process is shown in Figure 5.48. The cost decreases from £5189.25 to £4830.42 per year. Increasing the density difference therefore improves the cost of the separator and minimizes the cost of the integrated process.



Figure 5.41: Effect of density difference on cost of separation



Figure 5.42: Effect of density difference on oil recovery

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).


Figure 5.43: Effect of density difference on revenue generated from recovered oil



Figure 5.44: Effect of density difference on separator efficiency

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.45: Effect of density difference on oxygen demand





(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).



Figure 5.47: Effect of density difference on cost of bioprocess



Figure 5.48: Effect of density difference on cost of integrated process

(Influent oil characteristics, separator dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix III).

## 5.2.7 Discussion of results and analysis

The various scenarios in which the horizontal flow separator can be operated have been investigated and their effects on the integrated process presented. Generally, the scenarios can be divided into two subsets; one subset considers operating the separator with varying influent conditions and the other considers varying the design of the separators.

The results revealed the following: operating the horizontal flow separator with oily wastewater streams with a high density difference between the oil droplets and the water stream improves the efficiency of the separator and also the cost of separation. Operating the separator with increased oil concentration does not improve separator efficiency but reduces the cost of separation (due to increased revenue from recovered oil). Operating the separator to recover oil from influent oily wastewater streams with a large average oil droplet size improves the efficiency of the separator and the cost of separation. Using the horizontal flow separator to treat oily wastewater streams of low viscosities increases the efficiency of the separator and improves the cost of separation.

Increasing the size of the separator to treat a particular oily wastewater stream improves the efficiency up to a point, beyond which further increase in separator volume does not result in significant increase in efficiency. However, increasing separator volume increases the cost of treatment. Reducing separator horizontal velocity increases efficiency of the separator and reduces the cost of separation.

It was observed that all the factors that increased separator efficiency without any limiting effects on efficiency improved the cost of integrated process.

# 5.2.7.1 Effect of influent oil concentration on the integrated process

Operating the separator with increasing influent oil concentration reduces the cost of the separator (Figure 5.1). This decrease in cost of separation was as a result of increasing revenue generated from recovered oil. As influent oil concentration increased, influent oil volume increased. From Figure 5.4, it can be seen that increasing oil concentration does not affect the efficiency of the separator: efficiency remains the same at 29%. Therefore, whatever the volume of oil that is fed into the separator, 29% of that volume will be recovered, so the greater the volume of oil in the influent, the greater the volume

of oil recovered, the greater the revenue generated (Figure 5.3), and the smaller the overall cost of the separator (Figure 5.1).

It also followed that as influent oil concentration increased the effluent oil concentration increased and consequently the cost of the bioprocess increased (Figure 5.7). Oxygen demand is a function of oil concentration; Figure 5.5 shows the relationship between oxygen demand and oil concentration. Increasing oil concentration results in increasing bioreactor volume (Figure 5.6), and hence increased cost of the bioprocess.

Comparing Figures 5.1 and 5.8 it can be seen that while the cost of separation decreases with increasing oil concentration, the costs of the bioprocess and the integrated process increase. It has been seen that increasing oil concentration does not improve the cost of the integrated process, but these oil concentration-cost relationships can be used to define economic areas of operation. From Figure 5.1 it can be seen it is more profitable to operate the horizontal flow separator when at high oil concentrations (about 1500 mg/l) than when the oil concentration is low. Definition of low and high oil concentrations and may differ; in this exercise the low oil concentration was taken to be 1000 mg/l. Using the horizontal flow separator to recover oil when oil concentrations are low does not improve the cost of treatment: this is because much of the oil cannot be recovered and the bioprocess still needs to be employed to meet environmental regulations, which increases the cost of treatment. Using only the bioprocess decreases the cost of treatment at low oil concentrations. At very high oil concentrations, such as 40,000 mg/l, a large amount of revenue from recovered oil is expected; this would improve the cost of the separation since the relationship between oil concentration and revenue is directly proportional This analysis assumes the droplet distributions for all influent oil (Figure 5.3). concentrations are the same.

Increasing oil concentration does not affect the size of the separator and neither does it affect the efficiency of the separator. The efficiency of the separator remained constant at 29% (Figure 5.4) for various influent oil concentrations. Therefore oil concentration cannot be optimized to improve oil recovery in separator units.

# 5.2.7.2 Effect of horizontal flow velocity on the integrated process and economic region of operation

Decreasing horizontal flow velocity reduces the cost of the separation unit (Figure 5.9) and also reduces the cost of the bioprocess (Figure 5.15). This improvement in the economics of both units is as a result of the increased residence time of the oily wastewater in the separator unit. Decreasing the velocity allows more time for smaller droplets to flow out of the effluent flow path, therefore increasing oil recovery; the effect is an increase in revenue (Figure 5.11) and the cost of separation decreases. As a consequence of the increased oil recovery in the separator, the effluent oil concentration decreases; the oxygen demand decreases (Figure 5.13), hence the size and cost of the bioprocess also decrease (Figures 5.14 and 5.15). The overall effect on the integrated process is a reduction in cost as horizontal flow velocity decreases.

The relationship between horizontal velocity and the various costs is inversely proportional. A similar trend is observed in the horizontal flow velocity versus cost relationship for both the separation unit (Figure 5.9) and the bioprocess (Figure 5.15), suggesting that horizontal velocity is a parameter which can be optimized to improve the profitability of the integrated process. Figure 5.16 shows that the most economical region to operate both units is at low horizontal velocities.

The size of the separator is of paramount importance to the industry. In fact, following the need to enhance separation, the issue of separator size was the genesis for the development of other technologies. Ways in which separator efficiency could be increased without increasing separator size were also sought, leading to the use of chemicals such as coagulants and flocculants.

Decreasing horizontal velocity does not affect the size of the separator. A reduction in horizontal velocity can be achieved by the addition of baffles. These baffles improve laminar conditions, therefore improving separator efficiency. Figure 5.12 shows that separator efficiency increased by 234% when the horizontal velocity was decreased by 90%.

As can be seen in Figure 5.12, decreasing the horizontal velocity has no limiting effect on the efficiency of the separator as observed in the case of increasing separator size in Figure 5.20. Therefore, since decreasing horizontal velocity improves the cost of the

separator, bioreactor, and the integrated process, it seems to be a potential parameter in optimizing oil recovery in separator units.

#### 5.2.7.3 Effect of separator volume on the integrated process

Increasing the separator volume increases the cost of the separation but decreases the cost of the bioprocess (Figures 5.17 and 5.23). The separator volume was increased by making the separator longer; the increasing cost was due to the increasing quantity of construction materials. The decrease in the cost of the bioprocess is as a result of a reduced oil concentration in the effluent from the separator.

Increasing the separator volume increases the residence time that the oily wastewater remains in the separator and hence allows more droplets to rise out of the effluent path. The result is a decrease in effluent oil concentration and hence a decrease in oxygen demand, bioreactor size and hence cost of bioprocess (Figures 5.21, 5.22 and 5.23).

Comparing the contrasting relationship between separator size and cost of separation and the relationship between separator size and cost of the bioprocess (Figures 5.17 and 5.23), it follows therefore that increasing separator size cannot improve the cost of the integrated process: optimization is necessary.

Figure 5.20 also shows that as separator size increases, efficiency increases significantly but beyond 500% (about  $23m^3$  in volume), increase in separator size does not increase efficiency significantly. However, a different trend is observed in the relationship between separator volume and cost (Figure 5.17). As separator volume increases, the cost of separation increases in an almost linear fashion; there is no limiting effect on cost of separation as compared with the effect of separator volume on separator efficiency (Figure 5.20). The limiting effect is also observed in the relationship between separator volume and revenue generated (Figure 5.19); this is as a result of the limiting effect noticed in the separator volume–efficiency curve.

Figure 5.24 shows that as separator size increased, the cost of the integrated process decreased. However, by projection, continuous increase in separator size would result in an increase in separation cost and hence cost of the integrated process. It therefore follows that a minimum cost is expected in the cost curve for the integrated process. The minimum cost point defines the separator size which minimizes the cost of treatment; at that point the integrated process is operating economically in terms of separator volume.

# 5.2.7.4 Effect of average droplet size on the integrated process

As average droplet size was increased, the costs of both the separation and the bioprocess decreased (Figures 5.25 and 5.31); this improvement in the cost of both units is as a result of increased oil recovery (Figure 5.26), increasing the revenue generated (Figure 5.27). Increasing oil recovery reduces effluent oil concentration, and therefore reduces oxygen demand (Figure 5.29) and hence the size (Figure 5.30) and the cost of the bioprocess (Figure 5.31).

Increasing droplet size improves oil recovery by increasing the rate of rise of the droplets, which results in more droplets rising out of the effluent flow path.

Since increasing droplet size improves both the cost of separation and cost of bioprocess, it seems to be a parameter for improving oil recovery. As previously mentioned, the industry uses this factor to improve oil recovery. This is achieved by the addition of chemicals that promote agglomeration and coalescence; this increases the buoyancy of the droplets and hence improves oil recovery.

Figure 5.32 shows that increasing droplet size improves the cost of the integrated process; therefore it seems profitable to improve oil recovery by increasing droplet size. However, this exercise did not take into account the cost of chemicals, so in reality this may increase the cost of separation. Other costs incurred due to the use of chemicals that were not considered in this analysis are the cost of treatment of the recovered oil, the cost of treatment of the effluent, and the cost of disposal of the sludge. All of these costs should increase cost of treatment.

As previously stated, the size of equipment is important to the industry. Increasing droplet size is achievable by use of chemicals and does not affect the size of the separator. Therefore it seems a suitable parameter for optimizing oil recovery.

As can be seen in Figure 5.28, increasing droplet size increases efficiency and has no limiting effect on the efficiency of the separator as observed in the case of increasing separator size (Figure 5.20). Therefore, since increasing droplet size improves the cost of the separation, bioprocess, and integrated process, it seems to be a suitable parameter in optimizing oil recovery in horizontal flow separators.

## 5.2.7.5 Effect of viscosity on the integrated process

Oily wastewater streams with low water viscosity improve oil recovery (Figure 5.34) and revenue generated (Figure 5.35) and hence the costs of separation and the bioprocess are reduced (Figures 5.33 and 5.39).

Low viscosity implies the drag effect on the oil droplet is low; the droplets have a faster rise rate and hence oil recovery is increased. Increasing oil recovery reduces effluent oil concentration and hence oxygen demand (Figure 5.37). Consequently the bioreactor volume and cost of the bioprocess decreased with low viscosity oily wastewater streams (Figures 5.38 and 5.39).

The similar trend observed in the viscosity-cost relationships (Figures 5.33, 5.39 and Figure 5.40) suggests that viscosity is a suitable parameter to manipulate to improve oil recovery.

Currently, there is only one method of reducing viscosity and this is by heating.

Reducing the viscosity of the medium does not affect the size of the unit but it increases separator efficiency in an almost linear fashion (Figure 5.36). The relationship between viscosity and efficiency is inversely proportional. Also there is no limiting effect on separation efficiency as viscosity decreases. Therefore it may be used as a parameter in improving oil recovery.

# 5.2.7.6 Effect of density difference on the integrated process

Increasing density difference increases the buoyancy of the oil droplets which results in more oil droplets rising rapidly to the surface, therefore improving oil recovery (Figure 5.42). The effect of increasing oil recovery is an increase in revenue generated (Figure 5.43) and hence a reduction in the cost of separation (Figure 5.41). As oil recovery increases, effluent oil concentration, and hence oxygen demand (Figure 5.45), decreases. Consequently the size of the bioreactor decreases and hence the cost of the bioprocess (Figures 5.46 and 5.47) is reduced.

The similar trends in the density difference-cost relationship on both units suggest density difference may be a suitable parameter for optimizing separator units and the integrated process.



### Investigation of Oily Wastewater Treatment Processes

In practice, to increase density difference, bubbles are dissolved into oil-water mixtures at high pressures: these bubbles attach to the oil droplets to form composite particles which are much less dense than water and hence rise rapidly to the surface where they are skimmed off. Density difference is not usually used to improve oil recovery in horizontal flow separators. This factor is used in flotation units.

In this exercise, increasing density difference, with a fixed separator volume, resulted in an increase in the efficiency of the separator (Figure 5.44). No limiting effect on the efficiency was observed as density difference was increased; therefore density difference seems to be a suitable parameter to improve oil recovery.

### 5.2.7.7 Summary of results discussion

In summary, the factors which influence oil recovery and cost of the integrated process can be grouped into influent stream characteristics and separator designs or configurations. All influent stream characteristics except oil concentration can be manipulated to improve separation efficiency. Separator design or configuration can be manipulated to improve oil recovery. Increasing separator size and decreasing horizontal flow velocities are methods by which separator efficiency can be improved by manipulating the design or configuration. While the design variables increase the residence time of the wastewater in the separator, varying influent stream characteristics improves the rise rate of the oil droplets. While increase in separator size requires the construction of a larger tank to allow smaller droplets to rise out of the effluent flow path, decreasing horizontal velocity and influent stream characteristics may not necessitate the construction of a larger tank. The effects of manipulating the influent stream characteristics, except for oil concentration, are all predicted by Stokes' Law.

The factors which improve the cost of the integrated process are: viscosity, density difference, droplet size, and horizontal velocity. Separator size and oil concentration do not seem to improve the cost of the integrated process. It was observed that most factors which improved separator efficiency reduced the cost of the integrated process. So it can be said that the better the efficiency of the separator, the lower the cost of integrated process.

#### 5.2.7.8 Analysis of Stokes' Law variables

Based on an analysis of Stokes' Law, the most influential factors which improve the rise rate of an oil droplet are the viscosity and droplet size. Figure 5.49 compares the variables in Stokes' Law. This analysis is based on data given in Appendix III.



Figure 5.49: Comparing variables in Stokes' Law

From Figure 5.49, it can be seen that increasing droplet size has the greatest effect on rise rate when the percentage change in each variable is less than or equal to 60%; beyond a 60% change in variable, reducing the medium viscosity has the most effect on rise rate. These results seem to be the basis for the use of chemical additives to increase droplet size in the industry. However, from the analysis of Stokes' Law, it is seen that reducing the viscosity can have the greatest effect on rise rate; however, this requires the input of a lot of energy to heat the wastewater stream. Thermally-enhanced separation processes are not common in the industry, probably due to the high cost of energy.

The results presented in Section 5.1 were analyzed and presented in Tables 5.1, 5.2 and 5.3 to compare the effect of the various factors on separator efficiency.

### Investigation of Oily Wastewater Treatment Processes

Increase in separator size (%)	Increase in separator efficiency as a result of increase in size (%)	Increase in average droplet size (%)	Increase in separator efficiency as a result of increase in average droplet size (%)
0	0	0	0
17	21	17	26
33	39	33	55
50	57	50	88
67	73	67	124
150	139	150	241

# Table 5.1: Effect of separator size and droplet size on separator efficiency

Decrease in viscosity (%)	Increase in separator efficiency as a result of decreasing viscosity (%)	Increase in average droplet size (%)	Increase in separator efficiency as a result of increase in average droplet size (%)
0	0	0	0
17	18	17	26
33	48	33	55
50	98	50	88
67	168	67	124
150	877	150	241

# Table 5.2: Effect of viscosity and droplet size on separator efficiency

%age decrease in horizontal velocity	%age increase in separator efficiency as a result of decreasing horizontal velocity	%age increase in average droplet size	%age increase in separator efficiency as a result of increase in average droplet size
0	0	0	0
17	28	17	26
33	63	33	55
50	109	50	88
67	164	67	124
150	566	150	241

## Table 5.3: Effect of horizontal velocity and droplet size on separator efficiency

# 5.2.7.9 Comparing increasing droplet size and increasing separator volume

Oil droplet size can be increased, to improved buoyancy, by the addition of chemicals. The effect of using chemicals to increase separator efficiency was simulated by using the various average droplet sizes in the streams investigated.

### Investigation of Oily Wastewater Treatment Processes

From Table 5.1, a 17% increase in droplet size increases the separator efficiency by 26% and a 17% increase in separator size increases the separator efficiency by 21%. It may be concluded that increasing average droplet size improves separator efficiency more than increasing separator size, hence the usual practice of using chemical additives.

### 5.2.7.10 Comparing increasing droplet size and decreasing viscosity

The effect of decreasing the viscosity of the wastewater stream, to improve separator efficiency, was simulated by assuming the various streams studied were heated to reduce their viscosities.

Table 5.2 compares the effect of decreasing the viscosity and increasing average droplet size on separator efficiency. From Table 5.2, it seems a reasonable amount of energy must be expended to achieve a separator performance better than that achieved by increasing droplet size. These results also tally with that obtained in the analysis of Stokes' Law. Another disadvantage of using the thermal process is that the oily wastewater has to be cooled before the effluent is fed to the bioprocess. This involves huge operating expenditures and time and may be the reason the industry currently prefers the use of chemicals.

### 5.2.7.11 Comparing increasing droplet size and decreasing horizontal velocity

The effect of decreasing the horizontal velocity in the separator was simulated by assuming that the horizontal flow velocity of a separator with a fixed volume of  $4 \text{ m}^3$  was decreased by addition of baffles for each velocity investigated.

Table 5.3 compares the effect of decreasing horizontal velocity and increasing average droplet size on separator efficiency. From Table 5.3 it seems a 33% change in horizontal velocity increases the separator efficiency by 63% while a 33% change in average droplet size only increased separator efficiency by 55%. From Table 5.3, it may be said that decreasing the horizontal velocity has more effect on separator efficiency than increasing droplet size.

### 5.2.7.12 Critical design factor

From this analysis, it seems the factors or scenarios which best improve the effectiveness of the horizontal flow separator to recover oil are associated with the design of the separator rather than manipulation of the influent oily wastewater stream characteristics. Reducing horizontal velocity seemed to improve separator effectiveness better than the use of chemicals in this analysis.

The issue of longer residence time due to decreasing horizontal velocity popularized the use of chemicals to increase droplet size and thereby enhance oil recovery.

# 5.2.7.13 Comparing decreasing horizontal velocity with use of chemicals to improve oil recovery in a horizontal flow separator

- Decreasing horizontal velocity does not pose any hazard to the environment; use of chemicals has adverse effects on the environment
- Decreasing horizontal velocity does not involve any expense or continuous operating cost, the cost involved is a capital cost. Use of chemicals involves continuous operating cost which further increases cost of treatment
- Decreasing horizontal velocity involves long residence time. Use of chemicals involves long treatment times since recovered oil must be treated and effluent sent to the bioprocess must be treated
- Use of chemicals involves addition of more treatment units to treat both the recovered oil and the effluent. Decreasing horizontal velocity does not require additional treatment therefore does not require addition of more treatment units to treat effluent and recovered oil.
- Decreasing horizontal velocity in the design of separators does not result in additional formation of sludge which is harmful to the environment and adds further cost. Use of chemicals involves sludge formation.
- Decreasing horizontal velocity does not necessitate an increase in separator size it is achieved by addition of baffles. Use of chemicals does not necessitate an increase in separator volume but may require large doses (Zeng *et al.* 2007).
- The chemicals may be corrosive, necessitating a more expensive material of construction of the separator. Decreasing horizontal velocity does not have such an effect.

# 5.3 Integrated treatment process using CPI and ICB

The Corrugated Plate Interceptor (CPI) is basically a horizontal flow separator as previously reported. The major difference between the horizontal flow separator and the CPI is the addition of plates. To improve oil recovery, horizontal flow separators are sometimes retrofitted by the addition of plates. This section investigates the effect of varying CPI designs so as to ascertain a cost effective design which improves oil recovery.

The underlying theory for the design and predicting the performance of the CPI has been previously described in Chapters 2 and 3. For each design investigated, a volume of oily water was fed into the interceptor and the performance predicted (see Section 3.7.1). For each design, the size of Immobilized Cell Bioreactor (ICB) required to treat the effluent from the interceptor was calculated based on the oxygen demand of the effluent (see Section 2.3.2.2). The costs of constructing and operating the interceptor and the ICB were then calculated in each case. The results of the investigations are presented below. Interceptor dimensions and influent characteristics are presented in Appendix IV.

#### 5.3.1 Effect of varying interceptor plate width

To investigate the effect of plate width, a number of Corrugated Plate Interceptors were designed, using models previously described in Chapters 2 and 3, to recover oil from a fixed oily wastewater stream. The width of the plate pack was varied whilst holding influent characteristics and other design parameters constant. Interceptor dimensions, plate pack details, influent characteristics, and results are presented in Appendix IV. The cost of the treatment is presented below.

Figures 5.50 and 5.51 show that as plate width is increased from 1 to 5 m, oil recovery increases from 0.007 to 0.012 m<sup>3</sup>/h and the revenue generated increases from £16.14 to £30.84 per year. The effect of plate width on the cost of separation is shown in Figure 5.52: as plate width is increased, the cost of separation increases from £120.28 to £312.61 per year. Figure 5.53 shows the interceptor volume increases from 2.14 to 7.86 m<sup>3</sup> with the same increase in plate width. The effect on interceptor performance is shown in Figure 5.54: as plate width is increased, interceptor efficiency increases from 52 to 99%.

The effect of interceptor plate width on the bioprocess is shown in Figures 5.55, 5.56 and 5.57. As plate width is increased, oxygen demand decreases due to decreasing effluent

## Investigation of Oily Wastewater Treatment Processes

oil concentration from the interceptor; bioreactor size decreases from 76 m<sup>3</sup> to 817 litres and the cost of bioprocess decreases from £3428.27 to £50.93 per year. Increasing plate width also decreases the cost of the integrated process from £3548.56 to £363.55 per year (Figure 5.57).



Figure 5.50: Effect of plate width on oil recovery

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.51: Effect of plate width on revenue generated from recovered oil



Figure 5.52: Effect of plate width on cost of separation

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.53: Effect of plate width on interceptor volume



Figure 5.54: Effect of plate width on interceptor efficiency

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.55: Effect of plate width on oxygen demand



Figure 5.56: Effect of plate width on bioreactor volume

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.57: Effect of plate width on cost of integrated process

#### 5.3.2 Effect of varying interceptor plate angle

To investigate the effect of plate angle, a number of Corrugated Plate Interceptors were designed, using the models previously described in Chapters 2 and 3, to recover oil from a fixed oily wastewater stream. The plate angle was varied whilst holding influent characteristics and other design parameters constant. Interceptor dimensions, plate pack details, influent characteristics, and results are presented in Appendix IV. The cost of treatment is presented below.

Figure 5.58 shows the effect of the plates' angle of inclination to the horizontal on oil recovery in the Corrugated Plate Interceptor. As the angle of inclination is reduced from  $60^{\circ}$  to  $10^{\circ}$ , oil recovery increases from 0.0065 to 0.0105 m<sup>3</sup>/h and the revenue generated increases from £16.14 to £26.10 per year (Figure 5.59). Figure 5.60 shows that decreasing plate angle reduces the cost of the separation; the cost of separation decreases from £120.28 to £108.52 per year. The effect of plate angle on interceptor volume is shown in Figure 5.61: as plate angle is decreased, interceptor volume increases to a maximum of 2.66 m<sup>3</sup> and afterwards decreases to a minimum value of 1.94 m<sup>3</sup> at  $10^{\circ}$  to

#### Investigation of Oily Wastewater Treatment Processes

the horizontal. The effect of plate angle on the efficiency of the Corrugated Plate Interceptor is shown in Figure 5.62: as plate angle is decreased, interceptor efficiency increases from 52 to 84%. The increase in interceptor efficiency is as a result of increasing effective separation area as plate angle decreases. Increasing separation area for a constant flowrate decreases the flow velocity along the plates.

The effect of plate angle on the bioprocess is shown in Figures 5.63 and 5.64: as plate angle is decreased, the influent oil concentration into the bioreactor decreases; consequently oxygen demand decreases from 1.66 to 0.55 g/l.h and bioreactor volume decreases from 76.4 to 25.2 m<sup>3</sup>. In Figure 5.65, as plate angle is decreased, the cost of the bioprocess decreases from  $\pounds$ 3428.27 to  $\pounds$ 1156.72 per year; the cost of the integrated process decreases from  $\pounds$ 3548.56 to  $\pounds$ 1265.24 per year.





(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.59: Effect of plate angle on revenue generated from recovered oil



Figure 5.60: Effect of plate angle on cost of separation

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.61: Effect of plate angle on interceptor volume



Figure 5.62: Effect of plate angle on interceptor efficiency

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.63: Effect of plate angle on oxygen demand



Figure 5.64: Effect of plate angle on bioreactor volume

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.65: Effect of plate angle on cost of integrated process

### 5.3.3 Effect of varying interceptor plate spacing

To investigate the effect of plate spacing, a number of Corrugated Plate Interceptors were designed, using the models previously described in Chapters 2 and 3, to recover oil from a fixed oily wastewater stream. The plate spacing was varied whilst holding influent characteristics and other design parameters constant. Interceptor dimensions, plate pack details, influent characteristics, and results are presented in Appendix IV. The cost of treatment is presented below.

Figure 5.66 shows that as plate spacing is increased from 0.02 to 0.05 m, oil recovery increases from 0.0065 to 0.0068 m<sup>3</sup>/h and interceptor volume increases from 2.14 to 2.38 m<sup>3</sup> (Figure 5.67), to accommodate the increased plate spacing. Figure 5.68 shows that as the plate spacing is increased, the cost of separation increases from £120.28 to £121.51 per year; the revenue generated from recovered oil increases from £16.14 to £16.92 per year (see Appendix IV). The increase in revenue is due to the increasing volume of oil recovered as plate spacing is increased.

#### Investigation of Oily Wastewater Treatment Processes

The increase in oil recovery with increasing plate spacing for the same flowrate, influent oil concentration, number of plates, plate angle and plate dimensions, is due to the reduction in flow velocity along the plate. Increasing plate spacing means increasing the area the wastewater flows through. The greater the area for flow for the same flowrate, the lower the velocity of flow along the plates; the oil droplets have more time to rise out of the effluent flow path and hence the oil recovery is improved. Figure 5.69 shows that as plate spacing is increased the interceptor efficiency increases from 52.1 to 54.6%.

The effect of plate spacing on the bioprocess is shown in Figures 5.70 and 5.71. As plate spacing is increased, the oxygen demand of the bioprocess decreases from 1.66 g/l.h to 1.58 g/l.h and hence the bioreactor volume reduces from 76.4 to 72.4 m<sup>3</sup>. The relationship between plate spacing and cost of the integrated process is shown in Figure 5.72 where the cost of the bioprocess decreases from  $\pounds$ 3428.27 to  $\pounds$ 3254.26 per year and the cost of the integrated process decreases from  $\pounds$ 3375.77 per year.



#### Figure 5.66: Effect of plate spacing on oil recovery

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.67: Effect of plate spacing on interceptor volume



Figure 5.68: Effect of plate spacing on cost of separation

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.69: Effect of plate spacing on interceptor efficiency



Figure 5.70: Effect of plate spacing on oxygen demand

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.71: Effect of plate spacing on bioreactor volume



Figure 5.72: Effect of plate spacing on cost of integrated process

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).

## 5.3.4 Effect of varying interceptor plate length

To investigate the effect of plate length, a number of Corrugated Plate Interceptors were designed, using the models previously described in Chapters 2 and 3, to recover oil from a fixed oily wastewater stream. The plate length (parallel to flow) was varied whilst holding influent characteristics and other design parameters constant. Interceptor dimensions, plate pack details, influent characteristics, and results are presented in Appendix IV. The cost of treatment is presented below.

As plate length is increased from 1 to 5 m, oil recovery increases from 0.0065 to 0.0124  $m^3/h$  (Figure 5.73). The revenue generated then increases from £16.14 to £30.83 per year (Figure 5.74). Despite the increase in revenue, the cost of separation increases from £120.28 to £412.08 per year (Figure 5.79). The relationship between plate length and the size of the interceptor is shown in Figure 5.75: as plate length is increased, interceptor volume increases from 2.14 to 36.48  $m^3$ . The relationship between plate length and efficiency is shown in Figure 5.76, as plate length is increased, efficiency increases from 52 to 99.5%. The increase in efficiency is as a result of increasing the effective separation area as plate length increases. Increasing the separation area for the same flow rate reduces the velocity of flow along the plates and therefore more droplets have enough time to rise out of the effluent path. The increase in plate length also reduces the velocity of flow along the velocity of flow along the plates. Secondly, it increases the frictional force along the plates, so for the same flow rate as plate length increases.

The effects of interceptor plate length on the bioprocess are presented in Figures 5.77 and 5.78. As plate length is increased, influent oil concentration into the bioprocess decreases and hence oxygen demand decreases from 1.66 g/l.h to 0.018 g/l.h (Figure 5.77); bioreactor volume decreases from 76 m<sup>3</sup> to 849 litres (Figure 5.78). The effect of plate length on the integrated process is presented in Figure 5.79; the cost of the bioprocess reduces from  $\pounds$ 3548.56 to  $\pounds$ 464.71 per year.



Figure 5.73: Effect of plate length on oil recovery



Figure 5.74: Effect of plate length on revenue generated from recovered oil

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.75: Effect of plate length on interceptor volume



Figure 5.76: Effect of plate length on interceptor efficiency

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.77: Effect of plate length on oxygen demand



Figure 5.78: Effect of plate length on bioreactor volume

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.79: Effect of plate length on cost of integrated process

### 5.3.5 Effect of varying number of plates in interceptor

To investigate the effect of the number of plates, a number of Corrugated Plate Interceptors were designed, using the models previously described in Chapters 2 and 3, to recover oil from a fixed oily wastewater stream. The number of plates was varied whilst holding influent characteristics and other design parameters constant. Interceptor dimensions, plate pack details, influent characteristics, and results are presented in Appendix IV. The cost of treatment is presented below.

As the number of plates is increased from 5 to 25, oil recovery increases from 0.0065 to  $0.0124 \text{ m}^3/\text{h}$  (see Figure 5.80) and the revenue generated from recovered oil increases from £16.14 to £30.84 per year (see Figure 5.81). Figure 5.82 shows that interceptor volume increases from 2.14 to 2.94 m<sup>3</sup> as the number of plates is increased; this increase in interceptor volume is as a result of the fixed plate spacing considered in this investigation. It is possible that interceptor size may not be affected by increasing number of plates.

## Investigation of Oily Wastewater Treatment Processes

The number of plates also affects the efficiency of the separator. Figure 5.83 shows that efficiency increases from 52% to 99.5%. The increase in efficiency of the interceptor is as a result of increasing separation area. Increasing the number of plates in the interceptor increases the area available for separation and for coalescing of oil droplets. Increasing the number of plates for a given flow rate not only increases the area for separation and coalescing but also reduces the velocity of flow along the plates: the net effect therefore is an increase in oil recovery. Figure 5.84 presents the effect of number of plates on the cost of the interceptor. As the number of plates is increased, the cost of the interceptor increases from  $\pounds 120.28$  to  $\pounds 282.88$  per year.

The effects of the number of plates on the bioprocess are shown in Figures 5.85 and 5.86. As the number of plates is increased the oil concentration flowing into the bioreactor reduces and hence oxygen demand reduces from 1.66 to 0.018 g/l.h; bioreactor volume reduces from 76.4  $\text{m}^3$  to 817 litres.

The effect on the integrated process is presented in Figure 5.87, where the cost of the bioprocess reduces from £3428.27 to £50.93 per year, the cost of separation increases from £120.28 to £282.88 per year, and the cost of the integrated process reduces from £3548.56 to £333.81 per year.



Figure 5.80: Effect of number of plates on oil recovery



Figure 5.81: Effect of number of plates on revenue generated from recovered oil

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).


Figure 5.82: Effect of number of plates on interceptor volume

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.83: Effect of number of plates on interceptor efficiency

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).

Page 129



Figure 5.84: Effect of number of plates on cost of separation

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.85: Effect of number of plates on oxygen demand

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).

Page 130



Figure 5.86: Effect of number of plates on bioreactor volume

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).



Figure 5.87: Effect of number of plates on cost of integrated process

(Influent oil characteristics, interceptor dimensions, and performance were obtained using models developed in Chapter 3. The details are presented in Appendix IV).

#### 5.3.6 Discussion of results and analysis

#### 5.3.6.1 Effect of plate width on the integrated process

From Figure 5.50, it can be seen that increasing plate width to increase oil recovery improves oil recovery significantly in an almost linear fashion. Beyond a plate width of 3 m the trend changes and further increase in plate width does not result in a significant increase in oil recovery; however, there is an approximately linear relationship between plate width and cost of separation. As plate width increases, cost of separation increases significantly and there is no limiting effect.

Figure 5.57 shows that as plate width increases, cost of separation increases and cost of the bioprocess decreases; this implies there is a plate width which minimizes the cost of the integrated process. Therefore, in the integration of a corrugated plate interceptor and an immobilized cell bioreactor, the plate width which minimizes the cost of integration should be obtained.

#### 5.3.6.2 Effect of plate angle on the integrated process

From Figures 5.60 and 5.62, decreasing plate angle to the horizontal improves separation efficiency and reduces cost. However, in Figure 5.61, the effect of plate angle on interceptor volume follows a slightly different trend at a plate angle of  $40^{\circ}$ . This is because the interceptor designed as the base interceptor with a plate angle of  $60^{\circ}$  to the horizontal is inefficient. At the plate angle of  $40^{\circ}$  to the horizontal, the separator size increases and the efficiency of the base interceptor is improved, hence the abnormality in the trend. In general, for a well-designed interceptor, decreasing plate angle to the horizontal decreases the interceptor volume and increases efficiency. It can be seen from Figures 5.60, 5.62 and 5.65 that operating the interceptor at small angles to the horizontal increases efficiency and hence reduces the cost of the integrated process but in practice, this is not so. A plate angle between  $45^{\circ}$  and  $60^{\circ}$  is used in the industry. This is to enhance sludge removal, since smaller plate angles are less steep, sludge may take a longer time to slide down the plates.

#### 5.3.6.3 Effect of plate spacing on the integrated process

From Figure 5.66, it can be seen that increasing the spacing between plates increases oil recovery for the same flow rate. Figures 5.68 and 5.70 show that increasing the spacing

between plates increases the cost of the interceptor but decreases the oxygen demand and hence the cost of the bioprocess. It follows therefore there is an optimal distance apart which minimizes the cost of the integrated process. No limiting effect was observed on the effect of plate spacing on interceptor efficiency (Figure 5.69).

#### 5.3.6.4 Effect of plate length on the integrated process

From Figure 5.73, it can be seen that increasing the plate length (parallel to flow) in the corrugated plate interceptor does not result in significant increase in oil recovery beyond a plate length of 3 m; however, the cost of separation has an approximately linear relation with plate length (see Figure 5.79). It therefore follows that, beyond a certain length, increase in plate length does not significantly increase oil recovery but significantly increases cost of separation.

Figure 5.79 shows that while the cost of separation increases with increasing plate length, the cost of the bioprocess decreases, so there is expected to be an optimal plate length which minimizes the cost of an integrated process; this would be found if the curves are extrapolated to longer plate lengths.

#### 5.3.6.5 Effect of number of plates

From Figure 5.80, increasing the number of plates in the interceptor increases oil recovery and like the other variables such as plate length and plate width, there is an optimal number of plates beyond which increase in number does not significantly increase oil recovery; in Figure 5.80 the optimal number of plates is about 15. Since the cost of the separation increases and cost of the bioprocess reduces with increasing number of plates, it follows that there will be a number of plates at which the cost of integration is minimized.

In summary, the number of plates, plate spacing, angle of inclination to the horizontal, and plate dimensions all affect oil recovery and hence the cost of an integrated process. Plate dimensions and number of plates have a limiting effect on oil recovery which means beyond a certain value, increase in any of these factors does not result in significant increase in oil recovery so there is a need for these variables to be optimized. This effect was also noticed in the horizontal flow separators where increase in separator volume has a limiting effect on efficiency. Number of plates, plate dimensions, and separator volume are all related to surface area. It follows therefore that all factors which improve oil recovery by increasing surface area have limiting effects on efficiency and should be optimized for profitable operations.

The spacing between plates and angle of inclination does not have a limiting effect on oil recovery but there is a need to optimize spacing between plates because there is significant increase in size of interceptor as a result of increasing plate spacing. Construction of large tanks to improve oil recovery is not appreciated in the industry. The practice of not reducing the plate angle to low values has already been discussed.

Analysis of these results shows that, unlike the horizontal flow separator, all design parameters that affect oil recovery in the Corrugated Plate Interceptor result in an increase in interceptor volume. The industry appreciates a small footprint and since efficiency entails an increase in separator size, a variable which has the least effect on interceptor volume and the most effect on increasing efficiency would be the first choice in optimizing the process. The next section therefore compares the various effects of these factors on interceptor volume and efficiency so as to determine the best configuration which entails a small change in separator size and big positive change in efficiency.

## 5.3.6.6 Comparing the various parameters

To identify the critical factor for improving oil recovery in a corrugated plate interceptor, the following variables are compared: number of plates, plate angle, plate spacing, and plate dimensions. This section compares the effect of these factors on interceptor size (governed by the size of the plate pack) and efficiency of oil recovery.

Change in any variable (%)	Change in interceptor volume due to decrease in plate angle (%)	Increase in interceptor volume due to increase in plate spacing (%)	Increase in interceptor volume due to increase in plate length (%)	Increase in interceptor volume due to increase in plate width (%)	Increase in interceptor volume due to increase in number of plates (%)
0	0	0	0	0	0
25	22	2	43	17	2
50	21	3	94	33	5
100	-38	7	219	66	9

Table 5.4: Effect of various factors on interceptor volume

Change in any variable (%)	Increase in interceptor efficiency due to decrease in plate angle (%)	Increase in interceptor efficiency due to increase in plate length (%)	Increase in interceptor efficiency due to increase in plate spacing (%)	Increase in interceptor efficiency due to increase in plate width (%)	Increase in interceptor efficiency due to increase in number of plates (%)
0	0	0	0	0	0
25	33	22	1	21	23
50	52	40	2	38	41
100	60	64	3	64	64

## Table 5.5: Effect of various factors on plate interceptor efficiency

Table 5.4 compares effects the effects of decreasing plate angle, increasing plate spacing, increasing plate dimensions, and increasing number of plates on interceptor volume. From Table 5.4, a 25% decrease in plate angle resulted in a 22% increase in interceptor volume; as explained earlier this was an anomaly - decreasing plate angle decreases interceptor size as can be seen in the 100% decrease in plate angle.

Increasing the number of plates increases the volume of the interceptor necessary to accommodate the plate pack. From Table 5.4, a 25% increase number of plates resulted in a 2% increase in interceptor volume while a 25% increase in plate width and plate length resulted in a 17% and 43% increase in interceptor volume. From Table 5.4, increasing the plate length has the most effect on separator volume compared to all other factors. Comparing number of plates and plate dimensions, increasing the number of plates has the least effect on interceptor volume.

Decreasing plate angle reduces the size of the interceptor; the anomaly in trend in Table 5.4 has been explained previously. However, in practice plate angles of  $45-60^{\circ}$  are preferred for easy removal of sludge.

From Table 5.4, increasing plate spacing to improve recovery has the smallest effect on interceptor volume, compared to increasing number of plates and plate dimensions. For a 25% increase in plate spacing interceptor volume increased by only 2%.

From Table 5.5, increasing plate spacing has the least effect on efficiency when compared with all other factors. Increasing plate spacing does not increase the surface area available for separation but reduces the velocity of flow. For a 25% increase in plate spacing, efficiency increased only by 2%. Plate angle will not be analysed here, even though this has the greatest effect on efficiency, because in practice the plate angle must be between  $45-60^{\circ}$  for easy removal of sludge.

The effect of number of plates, plate dimensions on efficiency are comparable. For a 25 % increase in any of these factors, the increase in efficiency was between (21-23%). The effects are comparable because they increase surface area and reduce velocity.

The critical design factor which improves oil recovery and reduces cost of treatment should be based on the factor which has the least effect on interceptor volume but also improves the efficiency of the interceptor. The number of plates has the least effect on separator volume whilst the effect of increase in number of plates on oil recovery is comparable to that achieved by increasing plate dimensions. Increasing the number of plates is therefore better than increasing plate dimensions as the impact on the overall cost of the process is smaller.

# 5.3.6.7 Comparing the use of baffles and plates to improve efficiency in horizontal flow separators

Oil recovery in horizontal flow separators can be enhanced by the addition of baffles inside the unit; these baffles do not increase the size of the separator but reduce the velocity of flow. In Section 5.2, it was shown that reducing the velocity of flow is more effective in improving separator efficiency than increasing droplet size; based on other considerations the use of baffles improves the cost of integration more than the use of chemicals.

The effect of addition of baffles or plates on oil recovery can be compared by analysing the effect of plate spacing and number of plates on oil recovery in the corrugated plate interceptor. Increasing plate spacing reduces flow velocity along the plates and increasing number of plates increases surface area. From Table 5.5, increasing number of plates improves oil recovery more than increasing plate spacing. This is because the plates not only increase surface area - they reduce the velocity of flow as well. Therefore, in improving oil separation in horizontal flow separators, addition of plates seems to be better than addition of baffles, though this may not improve the cost of treatment due to the high cost of plates compared to the use of baffles. Therefore improving oil recovery in horizontal flow separator should be a matter of design rather than manipulation of influent oil characteristics.

## 5.3.6.8 Conclusions

Various means to improve oil recovery in a CPI were investigated by altering the design of a base-case CPI unit. The effectiveness of the CPI was improved by increasing the number of plates and increasing plate dimensions rather than increasing plate spacing, and reducing plate angle to the horizontal. The investigations revealed the following:

- Numbers of plates and plate dimensions have a limiting effect on improving separator efficiency but varying the plate angle and distance between plates did not produce such effects but rather had a linear relation with interceptor efficiency.
- Increasing number of plates, plate dimensions and distance between plates for a given flow rate improved the separation efficiency while decreasing the plate angle improved separator efficiency.

All the design factors which improve separator efficiency involve increasing interceptor volume. The only factor which does follow this trend is the decreasing the plate angle, it not only improves separator efficiency but decreases interceptor size (for lower plate angles). However, the plate angle is limited to between 45 and 60° for practical reasons.

Amongst the design factors which improve oil recovery, increasing the number of plates has the least effect on interceptor volume; therefore this becomes the critical factor for minimizing the cost of separation and integration.

# **Chapter 6: Conclusions and Recommendations**

## 6.1 Conclusions

Common industrial practice is to integrate processes for the separation and treatment of oily wastewater. Integration is necessary because, for the majority of wastewater streams, no single unit can economically meet the environmental discharge limits and recover valuable oil. With the increasing price of oil, and ever tighter discharge limits, the industry seeks ways of improving the cost-effectiveness of integrated processes for oily wastewater treatment. To this end, this research concerned itself with the study of two integrated processes for oily wastewater treatment:

- Integration of horizontal flow separator and Immobilized Cell Bioreactor
- Integration of Corrugated Plate Interceptor and Immobilized Cell Bioreactor

By studying the various factors which affect oil recovery in the integrated processes and their effect on the cost of the integrated process, a solution to improving the cost effectiveness of integrated processes was reached. To further improve the cost effectiveness of integrated treatment processes using an aerobically operated ICB, aeration in an ICB was studied, since aeration constitutes a major cost in ICB operation.

The results obtained from the studies suggest that the solution to improving the cost effectiveness of integrated processes lies in the design of the units used in the process, rather than manipulation of the wastewater characteristics. For the integrated process of horizontal flow separator and ICB, a 33% decrease in horizontal velocity increased separator efficiency by 63% in terms of oil recovery, while a 33% increase in droplet size, which could be achieved by the use of chemicals, increased separator efficiency by only 55%. The reduction in velocity increases the residence time of the oil in the separator therefore allowing more droplets to rise out of the effluent water stream before it reaches the exit. This residence time approach has been discarded by the industry, the argument being that increasing the residence time requires a longer processing time. However the cost benefits of this approach are twofold:

• It avoids the additional operating cost of buying chemicals (coagulants and flocculants) since reduction of horizontal velocity can be brought about by addition of baffles.

• It avoids the additional capital cost of the extra treatment units required to treat the recovered oil, effluent water, and sludge when chemicals are employed.

In the integrated process of Corrugated Plate interceptor and ICB, the results suggest that increasing the number of plates is the most cost effective method of improving oil recovery in plate interceptors. For a 25% increase in number of plates, interceptor efficiency increased by 23% while for a 25% increase in plate dimensions (length or width) interceptor efficiency increased by 22% or 21% respectively. These figures are close. The effect of increasing these design parameters on the interceptor volume was used as a deciding factor on choosing the most cost-effective solution to improving oil recovery. Increasing the plate length by 25% resulted in a 43% increase in interceptor volume, whilst increasing the number of plate by 25% resulted in only a 2% increase in interceptor volume. Increasing interceptor volume suggests increase in cost of construction and hence increased cost of separation.

The mass transfer characteristics in a laboratory scale ICB were studied for various different support particles. The 38 mm plastic Pall ring support particles were observed to have the best mass transfer characteristics when compared with the other sizes of 16 mm, 25 mm, and 50 mm. Though the smaller support particles were expected to produce better mass transfer characteristics, this was not observed. The reasons for this may be due to the arrangement of the support particles in the ICB unit and the internal structure of the support particles. However, these results show that optimization of the design plays a very important role in improving the cost of aeration.

Analysis of the various cost relationships in each of the integrated processes also revealed the following cost effective windows of operation in oily wastewater treatment or separation:

- It is more cost-effective to operate only an Immobilized Cell Bioreactor (ICB) when the average oil droplet size is small (less than 60 microns).
- It is more cost-effective to operate only an Immobilized Cell Bioreactor when only low oil concentrations of oil are present in the wastewater stream.

In conclusion, the solution to improving the cost effectiveness of integrated processes may lie in the design of the units rather than manipulation of influent oil characteristics as currently practiced in the industry.

## 6.2 Recommendations

The design and performance of the separators investigated in this research was based solely on theoretical models and considerations. The results obtained may not actually reflect what happens empirically. Therefore it is recommended that empirical studies be carried out to further investigate the claims of this study.

Also, in this research an actual oily wastewater stream was not characterized and used. Complex interactions occur between oil droplets, which may improve separator performance. This study did not consider interactions between droplets: the principle of unhindered settling was used in analyzing separator performance. A study which takes oil droplet interaction into consideration would present better results.

Another factor which could affect plate interceptor performance is the location of the plates in the interceptors. This was not studied and could have a considerable effect on separator efficiency and cost of separation. It is recommended further studies be carried out to investigate the effect of plate pack location on interceptor performance.

In the costing of the separation process, the cost of supporting equipment for the oil recovery units (such as oil skimmers) was not taken into consideration. These units are necessary to ensure that separators function properly; to make better judgements, the cost of these units should be taken into consideration.

In the study of aeration in ICB units, only a particular size of support particle was used in each design of ICB. Possibly, a mixture of various sizes of support particles may have resulted in better mass transfer characteristics but this was not investigated. It is recommended further work be done in this area. Also the effect of using support particles of various designs on aeration should be investigated. This research did not confirm the effect of arrangement of support particle on aeration. A random arrangement was used in the experimental studies; the effect of structured packing and random packing could be compared to investigate which improves aeration and correlations developed to aid in decision making. A mixing experiment should have been carried out to investigate the effect of support particle arrangement on liquid circulation rate. Also, in the aeration

investigations, oily wastewater was not used as the medium. It is recommended that oily wastewater, with microbes attached to the support particles, be used for further studies.

# **Bibliography**

- 1. Adlercreutz, P., Holst, O., and Mattiasson, B. (1982) "Oxygen supply to immobilised cells: 2. Studies on a co-immobilised algae-bacteria preparation with *in situ* oxygen generation" *Enzyme and Microbial Technology*, 4/6, 395-400.
- 2. Al-Shamrani, A.A., James, A., and Xiao, H. (2002) "Destabilisation of oil-water emulsions and separation by dissolved air flotation" *Water Research*, 36/6, 1503-1512.
- 3. American Petroleum Institute (API) (February 1990) Management of Water Discharges: Design and Operations of Oil-Water Separators, 1st ed., American Petroleum Institute.
- 4. Angeli, P., and Hewitt, G.F. (2000) "Drop size distributions in horizontal oil-water dispersed flows" *Chemical Engineering Science*, 55, 3133-3143.
- 5. Atkinson, B. (1986) "Immobilised cells, their applications and potential" in C. Webb, G.M. Black, and B. Atkinson (ed.) *Process Engineering aspects of immobilised cell systems* (Rugby: Institution of Chemical Engineers).
- 6. Beisinger, M.G., Vining, T.S., and Shell, G.L. (1974) "Industrial experience with dissolved air flotation", in *Proceedings of the 29th Purdue Industrial Waste Conference*, West Lafayette, IN, 1974, 290.
- 7. Bhatia, B., Nigam, K. D. P., Auban, D. and Hebrard, G. (2004) "Effect of a new high porosity packing on hydrodynamics and mass transfer in bubble columns" *Chemical Engineering and Processing*, 43/11, 1371-1380.
- 8. Bratby and Marais, G.V.R. (1975) "Saturator performance in dissolved air (pressure) flotation" *Water Res.* 9, 929.
- Brodelius, P., Nilsson, K. and Mosbach, K. (1981) "Production of alpha-keto acids. Part I. Immobilised cells of Trigonopsis variabilis containing D-amino acid oxidase" *Applied Biochemical. Biotechnology*, 6, 293-308.
- 10. Busca, G. (2004) Treatment of semi synthetic metalworking fluids: membrane filtration and bioremediation, PhD. Thesis, University of Nottingham, UK.
- 11. Capek, I. (2004) "Degradation of kinetically-stable o/w emulsions" Advances in Colloid and Interface Science, 107/2-3, 125-155.
- Cheryan, M. and Rajagopalan, N. (1998) "Membrane processing of oily streams, wastewater treatment and waste reduction" *Journal of Membrane Science*, 151/1, 13-28.
- 13. Coulson, J.M., Richardson, J.F., and Sinnot, R.K. (1991) Chemical Engineering Series Volume 6 (SI units) Design, (Oxford, Pergamon Press).

- 14. Da Fonseca, M.M.R, Black, G.M., and Webb, C. (1986) "Reactor configurations for immobilised cells" in C. Webb, G.M. Black, and B. Atkinson Process Engineering aspects of immobilised cell systems Chapter 4 pp 63-73.
- 15. Da Fonseca, M.M.R. (1983), PhD thesis, University of Manchester, Manchester UK.
- 16. Danckwerts, P.V. (1951) "Significance of liquid-film coefficients in gas absorption" Industrial & Engineering Chemistry 43, 1460–1467.
- Daraktschiev, R. (1984) "Ergebnisse einer Untersuchung der Gasverteilung über dem Querschnitt einer Füllkörperkolonne" *Chemical Engineering and Processing* 18/6, 317-322.
- Elibol, M. and Mavituna, F. (1995) "Effect of perfluorodecalin as an oxygen carrier on actinorhodin production by Streptomyces coelicolor A3 (2)". *Appl. Microbiol. Biotechnol.* 43, 206-210.
- 19. Freeman, A. and Lilly, M.D. (1998) "Effect of processing parameters on the feasibility and operational stability of immobilized viable microbial cells" *Enzyme and Microbial Technology*, 23/5, 335-345.
- 20. Gardner, N.A. (1972) "Flotation techniques applied to the treatment of effluents" *Effluent and Water Treatment Journal*, 12, 82.
- 21. Gray, N.F. (1989) Biology of wastewater treatment. Oxford University Press, UK.
- 22. Gryta, M., Karakulski, K. and A. W. Morawski (2001) "Purification of oily wastewater by hybrid UF/MD" *Water Research*, 35/15, 3665-3669.
- 23. Hashmi, K. A., Hamza, H.A. and Wilson, J.C. (2004) "CANMET hydrocyclone: an emerging alternative for the treatment of oily waste streams" *Minerals Engineering*, 17/5, 643-649.
- 24. Higbie, R. (1935) "The rate of absorption of a pure gas into a still liquid during short periods of exposure" Trans. American Institute of Chemical Engineers 31, 365.
- 25. Holst, O., Enfors, S.O. and Mattiasson, B. (1982) "Oxygenation of immobilized cells using hydrogen peroxide: a model study of Gluconobacter oxydans converting glycerol to dihydroxyacetone" *European journal of Applied Microbiology and Biotechnology*, 14/2, 64-68.
- 26. http://www.megator.com/pdf/Parallel%20plate%20USA.pdf
- 27. http://www.torrcanada.com, (2007).
- 28. Iggleden, G.J. (1974) "The design and application of tilted plate separator oil interceptors", *Chemistry and Industry* November 1974.
- 29. Karakulski, K., Kozlowski, A., and Morawski, A. W. (1995) "Purification of oily wastewater by ultrafiltration" *Separations Technology*, 5/4, 197-205.

- 30. Kim, J., Jung, J., Yeom, I., and Aoh, G. (2007) "Electric fields treatment for the reduction of membrane fouling, the inactivation of bacteria and the enhancement of particle coagulation" *Desalination*, 202/1-3, 31-37.
- Kitchener, J.A. (1985) "The froth flotation process: past, present and future -in brief. In the scientific basis of flotation Part 1. NATO Advances Study Institute, 1-26.
- 32. Kobayashi, T., Kobayashi, T., Hosaka, Y., and Fujii, N. (2003) "Ultrasoundenhanced membrane-cleaning processes applied water treatments: influence of sonic frequency on filtration treatments" *Ultrasonics*, 41, 85-190.
- 33. Lash, L.D. and Kominek, E.G. (1975) "Primary waste treatment methods" Chem. Eng. 82, 4.
- 34. Le-Clech, P., Jefferson, B., and Judd, S. J. (2003) "Impact of aeration, solids concentration and membrane characteristics on the hydraulic performance of a membrane bioreactor" *Journal of Membrane Science*, 218/1-2 117-129.
- 35. Lee, J.C., and Buckley, P.S. "Fluid mechanics and aeration characteristics of fluidised beds", in P.F. Cooper and B. Atkinson, (Eds.), *Biological Fluidized Bed Treatment of Water and Wastewater*, Ellis Horwood, Chichester UK, 1981, pp. 62-74.
- 36. Lefebvre, A.H. (1989) Atomization and Spray (Washington Hemisphere Publishing, Corporation).
- 37. Lewis, W.K. and Whitman, W.G. (1924) "Principles of gas absorption" Industrial & Engineering Chemistry 16, 1215–1220.
- Li, H., Cao, Y., Qin, J., Jie, X., Wang, T., Liu, J., and Yuan, Q. (2006), "Development and characterization of anti-fouling cellulose hollow fiber UF membranes for oil-water separation" *Journal of Membrane Science*, 279/1-2, 328-335.
- 39. Li, Y., He, Y., Liu, Y., Yang, S., and Zhang, G. (2005) "Comparison of the filtration characteristics between biological powdered activated carbon sludge and activated sludge in submerged membrane bioreactors" *Desalination*, 174/3, 305-314.
- 40. Li, Yu Shui, Lu Yan, Chai Bao Xiang and Liu Jiang Hong (2006) "Treatment of oily wastewater by organic-inorganic composite tubular ultrafiltration (UF) membranes" *Desalination*, 196/1-3, 76-83.
- 41. Low, E.W. and Chase, H.A. (1999) "Reducing production of excess biomass during wastewater treatment" *Water Research* 33/5, 1119-1132.
- 42. Lundh, M., Jönsson, L., and Dahlquist (2002) "The influence of contact zone configuration on the flow structure in a dissolved air flotation pilot plant" *Water Research*, 36/ 6, 1585-1595.

- 43. Ma, H., Bowman, C.N., and Davis, R.H. (2000) "Membrane fouling reduction by backpulsing and surface modification" *Journal. Membrane Sci.* 173, 191.
- 44. Maddock, J.E.L. and Tomlinson, E.J. (1980) "The clarification of effluent from an active-sludge plant using dissolved-air flotation". *Wat. Pollut. Control.* 79, 117-125.
- 45. Martin, M., Mengs, G., Plaza, E., Garbi, C., M. Sanchez, Gibello, A., Gutierrez, F., and Ferrer, E. (2000) "Propachlor removal by Pseudomonas strain GCH1 in an immobilized-cell system". *Applied. Environmental. Microbiology*, 66. 1190-1194.
- 46. Metcalf & Eddy, Tchobanoglous, G., Burton, F. (1991) Wastewater Engineering: Treatment, Disposal, Reuse: 3rd ed. (McGraw Hill- New York, London).
- 47. Mohammadi, T., Kazemimoghadam, M., and Saadabadi, M. (2003) "Modelling of membrane fouling and flux decline in reverse osmosis during separation of oil in water emulsions" *Desalination*, 157 / 1-3, 369-375.
- 48. Moosai, Roshni, and. Dawe, R.A. (2003) "Gas attachment of oil droplets for gas flotation for oily wastewater cleanup" Separation and Purification Technology, 33/3, 303-314.
- 49. Mueller, J., Cen, Y. and Davis, R.H. (1997) "Crossflow microfiltration of oily water" *Journal. Membrane Science*, 129/2, 221.
- 50. Mugele, R.A. and Evans, H.D. (1951) "Droplet size distribution in sprays" Ind. Eng. Chem., 43, 1317-1324.
- 51. Mysore, D., Viraraghavan, T., and Jin, Y. (2005) "Treatment of oily waters using vermiculite" *Water Research*, 39/12, 2643-2653.
- Nakhla, G., Liu, V., and Bassi, A. (2006) "Kinetic modelling of aerobic biodegradation of high oil and grease rendering wastewater" *Bioresource Technology*, 97/1, 131-139.
- 53. Nassif, N.G. and Hansard, W.N. (2003) "Advanced Laminar flow oil /water technology" in *practical solutions for drilling challenges* (Radisson Astrodome, Houston, Texas, April 1-3, 2003).
- 54. Nemerow, N.L. (1978) Industrial Water Pollution: Origins, Characteristics and Treatment, 2<sup>nd</sup> Edition, (Addison-Wesley, Reading, MA).
- 55. Nikakhtari, H., and Gordon, A. H. (2005) "Hydrodynamic and oxygen mass transfer in an external loop airlift bioreactor with a packed bed" *Biochemical Engineering Journal*, 27/2, 138-145.
- 56. Nordvik, A.B., Simmons, J.L., Bitting, K.R., Lewis, A., and Strøm-Kristiansen, T. (1996) "Oil and water separation in marine oil spill clean-up operations" Spill Science & Technology Bulletin, 3/3, 107-122.

- 57. Ostergaard (1978) in Webb, C., Black, G.M., and Atkinson, B. (ed.) *Process* Engineering aspects of immobilised cell systems (Rugby: Institution of Chemical Engineers).
- 58. Park, Y.S., Yun, J.W., Kim, D.S., and Song, S.K. (1998) "Wastewater treatment in a packed bed reactor with immobilized cells onto a ceramic carrier" *Biotechnology Techniques*, 12/6, 459-462.
- 59. Poulsen, A. K., Arleth, L., Almdal, K., and Scharff-Poulsen, A.M. (2007) "Unusually large acrylamide induced effect on the droplet size in AOT/Brij30 waterin-oil microemulsions" *Journal of Colloid and Interface Science*, 306/1, 143-153.
- 60. Rhee, C.H. Martyn, P.C., and Kremer, J.G. "Removal of oil and grease in the hydrocarbon processing industry" in *Proceedings of the 42nd Purdue Industrial Waste Conference*, West Lafayette, IN 1988, 143-150.
- 61. Rommel, W., Blass, E., and Meon, W. (1992) "Plate separators for dispersed liquidliquid systems: Multiphase flow, droplets coalescence, separation performance and design" *Chemical Engineering Science*, 47/3, 555-1992.
- 62. Schultz, T.E. (2005) "Biological wastewater treatment" Chem. Eng. 44-51.
- 63. Silver, I.A. (1965) "Oxygen micro electrodes" Med. electron. Biol. Engng 3, 377-387.
- 64. Smith, R.W. (1989) "Flotation of algae" Mineral Processing and Extractive Metallurgy, Review 4, 277-299.
- 65. Sood, A. and Awasthi, S.K. (2003) "Initial droplet size distribution in miniemulsion polymerization" *Journal of Applied Polymer Science*, 88/14, 3058-3065.
- 66. Srijaroonrat, P., Julien, E., and Aurelle Y. (1999) "Unstable secondary oil/water emulsion treatment using ultrafiltration: fouling control by backflushing" Journal of Membrane Science, 159/ 1-2, 11-20.
- 67. Szwajcer, E., Brodelius, P., and Mosbach, K. (1982) "Production of alpha-keto acids: 2. Immobilised whole cells of Providencia sp. PCM 1298 containing L-amino acid oxidase" *Enzyme Microbial Technology*, 4, 409-413.
- 68. Takahashi, T., Miyaharam T., and Mochizukj, H. (1979) "Fundamental study of bubble formation in dissolved air pressure flotation". J.Chem.Engr.Jpn. 4, 275-280.
- 69. Tin, V. and Cheng, D.M. (1994-1995) "Advances in oil/water separation technology" Annual issue of HE (HK), 20-24.
- 70. Villas-Bôas, R.C. and Barreto, M.L. (1996) "Clean technologies for the mineral industries: the need of P<sup>2</sup> solutions" in M. Sanchez, F. Vergara and S. Castro Clean Technologies for the Mining Industry, (ed.) (Santiago-Chile) 43-57.
- 71. Votruba, J., Sobotka, M., and Prokop, Z. (1976) in communications to the editor on the effect of air hold up on readings of dissolved oxygen probe *Biotechnology and Bioengineering* 19/3, 435-438.

- 72. Votruba, J., Sobotka, M., and Prokop, Z. (1977) in communications to the editor on the "effect of air hold up on readings of dissolved oxygen probe II" *Biotechnology* and *Bioengineering* 20/6, 913-916.
- 73. Wang, X., Li, X., and Huang, X. (2006) "Membrane fouling in a submerged membrane bioreactor (SMBR): Characterisation of the sludge cake and its high filtration resistance" Separation and Purification Technology, 52/3, 439-445.
- 74. Warhurst, A. and Bridge, G. (1996) "Improving environmental performance through innovation: recent trends in the mining industry" *Minerals Engineering*, 9, 907-921.
- 75. Wei, Y., Van Houten, R.T., Borger, A.R., Eikelboom, D.H., and Fan, Y. (2003) "Minimization of excess sludge production for biological wastewater treatment." *Water Research*, 37/18, 4453-4467.
- 76. Weintraub, M.H., Gealer, R.L., Golovoy, A., Dzieciuch, M.A., and Durham, H. (1983) "Development of electrolytic treatment of oily wastewater" *Environ. Progr*, 2, 32-37.
- 77. Williams, J.A. (2002) "Keys to bioreactor selection" Chemical Engineering Magazine, 34-46.
- 78. Wolfe, P.R. (1992) "Economics of used oil recycling: still slippery" *Resource Recycling*, 23-40.
- 79. Yao K.M. (1973) "Design of high rate settlers" JASCE, Env. Eng.Div., 99, 621-637.
- 80. Zeng, Y., Yang, C., Zhang, J., and Pu, W. (2007) "Feasibility investigation of oily wastewater treatment by combination of zinc and PAM in coagulation/flocculation" *Journal of Hazardous Materials, In Press.*
- Zhao, X., Wang, Y., Ye, Z., Borthwick, A.G.L., and Ni, J. (2006) "Oil field wastewater treatment in Biological Aerated Filter by immobilized microorganisms" *Process Biochemistry*, 41/7, 1475-1483.
- 82. Zhi-Guo, M., Feng-lin, Y., and Xing-Wen, Z. (2005) "MBR focus: do nonwovens offer a cheaper option?" *Filtration & Separation*, 42/5, 28-30.
- 83. Zhong, J., Sun, X., and Wang, C. (2003) "Treatment of oily wastewater produced from refinery processes using flocculation and ceramic membrane filtration" *Separation and Purification Technology*, 32/1-3, 1 93-98.
- 84. Zouboulis, A. I., and Avranas, A. (2000) "Treatment of oil-in-water emulsions by coagulation and dissolved-air flotation" Colloids and Surfaces A: Physicochemical and Engineering Aspects, 172/1-3, 153-161.

# Appendix I Design and cost data for oil-water separators

## 1 Factors used in the design of a horizontal flow separator

The product of the short-circuiting factor  $F_s$  and turbulent factor  $F_t$  yields the design factor F. A curve can be plotted to obtain the factor F from Table 1.

$v_h / v_t$	F
20	1.45
15	1.37
10	1.27
6	1.14
3	1.07

Table 1: Typical values of design factor (F)

## 2 Plate Separators

There are four basic types of plate separator. These are inclined plate, tilted plate, crossflow, and matrix type plate separators.

## 2.1 Inclined Plate Separator

The plates within this separator are flat, parallel, and inclined at 45 degrees. This unit is capable of removing droplets of diameter 60 microns and larger. The majority of entrained solids fall out at the inlet of inclined plate separators; oily water flows though the plate packs in laminar flow conditions. Oil rises and adheres to the underside of the plates. Separated oil flows as a film up the plates, disengages from the plates and rise to the liquid surface in large drops where it is skimmed off; clean water underflows an oil retention baffle and leaves the unit.

## 2.2 Corrugated Plate Interceptor

The plates in this unit are corrugated. The corrugations in the plates do not increase the effective plate area but bring about two benefits to the separator. Firstly, they serve to segregate the oil from the solids being separated and enhance coalescence/agglomeration of each of the liquid phases. Secondly, the corrugations add rigidity to the plate assembly. As with the inclined plate separator, oil droplets rise to the underside of the plates. This unit is capable of removing droplets of diameter 60 microns and larger.

## 2.3 Cross-flow Separators

Cross-flow separators may have corrugated plates; the major difference between the corrugated plate separator and the crossflow separator is the water flow direction. In order to minimize separator size while maintaining throughput, the water flow is arranged across rather than along the corrugations. The oil and sludge flow at 90° to the water: oil flows upwards to the underside of the plates and sludge downwards to the upper surface of the plates. This unit is capable of removing droplets of diameter 60 microns and larger.

#### 2.4 Matrix Plate Separators

The plates in this separator are located in a horizontal cylindrical vessel. The open design of the plates maintains a large contact area for oil separation whilst minimizing the possibility of oil and solids mutual attachment. This unit is also capable of removing droplets of diameter 60 microns and larger.

#### **3** Analysis of droplet capture in a Corrugated Plate Interceptor

The actual path travelled by an oil droplet takes the form of a curve (EF) but Yao 1973 approximated this path to a straight-line (BD) (Figure 1).



Figure 1: Theory of plate settlers

The path of the droplet is complex because of the effect of parabolic velocity gradient. However the straight line BD simplifies the path of the particle with the slowest rise rate.

Inspecting Figure 1,

$$v_t / v_p = AB/AD \tag{1.0}$$

hence,

$$v_t = v_p * AB/AD \tag{1.0a}$$

but,

$$AB = d/\cos\alpha \qquad (1.1)$$

and,

$$AD = w_P + d / \tan \alpha \qquad (1.2)$$

therefore the rise rate of the critical droplet is given by:

$$v_t = v_p * d / \cos \alpha * l / (w_P + d / \tan \alpha) \qquad (1.3a)$$

simplifying,

$$v_t = v_p * d (d \sin \alpha + w_P \cos \alpha) \qquad (1.3b)$$

## 4 Cost Factors

Purchase cost of equipment, cost factors used in Equation 2.34 (Coulson J.M. et al 1991).

Equipment	Size unit (S)	Size range	Constant C(£)	Index (n)	Comment
Plate	Area m <sup>2</sup>	5-50	1000	0.60	Carbon steel
Tanks Process Vertical Horizontal	Capacity m <sup>3</sup>	1-50 10-100	500 600	0.59 0.60	Atmos. press. Carbon steel

Table	2:	Cost	Fa	cto	rs
-------	----	------	----	-----	----

## Appendix II Experimental Results

The calibration data obtained for the rotameter was plotted to obtain a calibration curve (see Figure 1).



Figure 1: Calibration curve for the rotameter

Calculated values of mass transfer coefficient versus aeration rate are presented in Tables 1 to 4 below.

Aeration rate (ml/s)	Mass transfer coefficient (s <sup>-1</sup> )
48.0	0.0039
38.0	0.0035
24.0	0.0024
9.0	0.0020

Table 1: Mass transfer coefficient vs. aeration rate for 16 mm support particle

.

Aeration rate (ml/s)	Mass transfer coefficient (s <sup>-1</sup> )
47.0	0.0097
41.6	0.0086
28.0	0.0055
17.5	0.0047
9.0	0.0030

Table 2: Mass transfer coefficient vs. aeration rate for 38 mm support particle

Aeration rate (ml/s)	Mass transfer coefficient (s <sup>-1</sup> )
56.0	0.0043
43.6	0.0042
31.1	0.0033
27.0	0.0029
13.0	0.0020

 Table 3: Mass transfer coefficient vs. aeration rate for 25 mm support particle

Aeration rate (ml/s)	Mass transfer coefficient (s <sup>-1</sup> )
50.0	0.0058
40.6	0.0035
30.0	0.0021
15.0	0.0020
10.5	0.0015

Table 4: Mass transfer coefficient vs. aeration rate for 50 mm support particle

# Appendix III Theoretical results: Integration of a horizontal flow separator and an ICB

Input data and results for investigations of integrated processes using a horizontal flow separator and an Immobilized Cell Bioreactor.

Horizontal flow separator performance prediction spreadsheet

Effect of oil concentration Effect of horizontal flow velocity Effect of separator volume Effect of average droplet size Effect of viscosity Effect of density difference

Analysing Stokes' Law

# Horizontal flow separator performance prediction spreadsheet

Influent characteristics		Separator	Separator dimensions			
Water flowrate Water density	10 m³/h 1000 kg/m³ 0.002 Ba s	Length Width Height	2.3248 m 1.8288 m			
Oil concentration Average oil density	1000 mg/ł 800 kg/m³	neigin	0.3144 11			
Calculations		volume	3.8876085 m <sup>3</sup>			

		SA	4.2515404 m <sup>2</sup>
Mass of oil in influent	10 kg/h	XSA	1.6722547 m <sup>2</sup> cross-sectional area
Volume of oil in influent	0.0125 m³/h	horiz. vel.	0.004905 m/s (0.97 ft/min)
		res. time	474 s (7.9 min)

		Infl	uent			Recovered oil Water effluent						
											Percentage	
					Fraction					Cumulative	of oil	Cumulative
Droplet	Mean	Number of	Volume of		lo	Number of	Volume of	Number of	Volume of	volume of oil	contributed	percentage of
size	droplet	droplets in	droplets in	Terminal	droplets	droplets	droplets	droplets	droplets	in each	by each	oil in each
interval	size	interval	interval	velocity	removed	removed	removed	remaining	remaining	interval	interval	interval
	(µm)		(m³)	(m/s)			(m³)		(m³)	(m³)		
1	5	3.970E+07	2.598E-09	1.4E-06	0.0007	2.804E+04	1.835E-12	3.967E+07	2.596E-09	2.598E-09	0.00002	0.00002
2	15	3.147E+08	5.561E-07	1.2E-05	0.0064	2.000E+06	3.535E-09	3.127E+08	5.526E-07	5.587E-07	0.00445	0.00447
3	25	1.751E+09	1.432E-05	3.4E-05	0.0177	3.091E+07	2.529E-07	1.720E+09	1.407E-05	1.488E-05	0.11458	0.11905
4	35	6.334E+09	1.422E-04	6.7E-05	0.0346	2.192E+08	4.921E-06	6.115E+09	1.373E-04	1.571E-04	1.13770	1.25675
5	45	1.492E+10	7.120E-04	1.1E-04	0.0572	8.536E+08	4.073E-05	1.407E+10	6.713E-04	8.691E-04	5.69642	6.95317
6	55	2.290E+10	1.995E-03	1.6E-04	0.0855	1.957E+09	1.705E-04	2.094E+10	1.824E-03	2.864E-03	15.96087	22.91404
7	65	2.290E+10	3.293E-03	2.3E-04	0.1194	2.733E+09	3.930E-04	2.017E+10	2.900E-03	6.157E-03	26.34563	49.25968
8	75	1.492E+10	3.296E-03	3.1E-04	0.1589	2.371E+09	5.238E-04	1.255E+10	2.772E-03	9.453E-03	26.37231	75.63198
9	85	6.334E+09	2.037E-03	3.9E-04	0.2041	1.293E+09	4.157E-04	5.042E+09	1.621E-03	1.149E-02	16.29599	91.92798
10	95	1.751E+09	7.859E-04	4.9E-04	0.2549	4.463E+08	2.004E-04	1.304E+09	5.855E-04	1.228E-02	6.28733	98.21531
11	105	3.147E+08	1.908E-04	6.0E-04	0.3114	9.801E+07	5.941E-05	2.167E+08	1.313E-04	1.247E-02	1.52614	99.74146
12	115	3.677E+07	2.928E-05	7.2E-04	0.3736	1.374E+07	1.094E-05	2.303E+07	1.834E-05	1.250E-02	0.23426	99.97571
13	125	2.789E+06	2.852E-06	8.5E-04	0.4414	1.231E+06	1.259E-06	1.558E+06	1.593E-06	1.250E-02	0.02282	99.99853
14	135	1.372E+05	1.767E-07	9.9E-04	0.5148	7.061E+04	9.097E-08	6.654E+04	8.573E-08	1.250E-02	0.00141	99.99994
15	145	4.370E+03	6.975E-09	1.1E-03	0.5939	2.595E+03	4.143E-09	1.774E+03	2.832E-09	1.250E-02	0.00006	100.00000
16	155	9.007E+01	1.756E-10	1.3E-03	0.6787	6.113E+01	1.192E-10	2.894E+01	5.643E-11	1.250E-02	0.00000	100.00000
17	165	1.200E+00	2.823E-12	1.5E-03	0.7691	9.230E-01	2.171E-12	2.772E-01	6.519E-13	1.250E-02	0.00000	100.00000
18	175	1.032E-02	2.897E-14	1.7E-03	0.8651	8.931E-03	2.506E-14	1.392E-03	3.907E-15	1.250E-02	0.00000	100.00000
19	185	0.000E+00	0.000E+00	1.9E-03	0.9668	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
20	195	0.000E+00	0.000E+00	2.1E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
21	205	0.000E+00	0.000E+00	2.3E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
22	215	0.000E+00	0.000E+00	2.5E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
23	225	0.000E+00	0.000E+00	2.8E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
24	235	0.000E+00	0.000E+00	3.0E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
25	245	0.000E+00	0.000E+00	3.3E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
26	255	0.000E+00	0.000E+00	3.5E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
27	265	0.000E+00	0.000E+00	3.8E-03	1.0000	0.000E+00	0.000E+00	0.000E+00	0.000E+00	1.250E-02	0.00000	100.00000
Totals		9.252E+10	1.250E-02			1.002E+10	1.821E-03	8.251E+10	1.068E-02		100.00000	

Effect of oil concentration on the integrated process of horizontal flow separator and ICB: Input parameters

Influent characteristics		Separator dimensions	
Water flowrate Water density	10 m³/h 1000 ka(m³	Width Heicht	1.8288 m 0 9144 m
Water viscosity	0.001 Pa.s	Length	2.3248 m
Oil concentration Average oil density	VARIABLE mg/l 800 kg/m³		
Average droplet size Standard deviation	60 µm 15 µm		
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

S
S
~
m
ΰl
뀕
aı
H
5
2
g
5
S
ĭ <u>≤</u>
2
1
9
E
3
·51
5
È.
<b>H</b>
~
ŝ
ဗ
ē
ā
١Ē
5
둾
5
9
Ø
<u> </u>
ö
디
0
믭
Ë
E
ဗျ
ă
- 81
_
. <u>9</u>
آيد
<u> </u>
5
ણા
μц

Influent oil	Oil flow to	Separator	Residence	Oil	Effluent oil	Separator	Oil flow to	Oxygen	Bioreactor
concentration	separator	volume	time	recovery	concentration	efficiency	bioreactor	demand	volume
(mg/l)	(m³/h)	(m³)	(min)	(m³/h)	(I/gm)	(%)	(m³/h)	(g/I.h)	(m³)
1000	0.01250	3.888	7.9	0.0036	602	29.1	0.00886	2.46	113.1
1200	0.01500	3.888	7.9	0.0044	850	29.1	0.01063	2.95	135.7
1300	0.01625	3.888	7.9	0.0047	921	29.1	0.01151	3.20	146.9
1400	0.01750	3,888	7.9	0.0051	<u> 9</u> 92	29.1	0.01239	3.44	158.2
1500	0.01875	3.888	7.9	0.0055	1063	29.1	0.01328	3.69	169.5

Influent oil	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
concentration	separator	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(I/gm)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
1000	44.56	9.03	35.53	561.84	4591.89	5153.72	5189.25
1200	44.56	10.84	33.72	632.49	5509.79	6142.28	6176.00
1300	44.56	11.74	32.82	666.20	5967.95	6634.15	6666.97
1400	44.56	12.65	31.91	699.07	6426.97	7126.04	7157.96
1500	44.56	13.55	31.01	731.16	6886.39	7617.55	7648.56

Effect of horizontal flow velocity on the integrated process of horizontal flow separator and ICB: Input parameters

Influent characteristics		Separator dimensions	
Water flowrate Water density	10 m³/h 1000 kơ/m³	Width Heicht	1.8288 m 0.9144 m
Water viscosity	0.001 Pa.s	Length	2.3248 m
Oil concentration	1000 mg/l		
Average oil density	800 kg/m³		
Average droplet size	60 µm		
Standard deviation	15 µm		
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

lts
kesu
B: F
d IC
r an
rato
sepa
MO
tal fl
zon
hori
s of
oces
d pr
rate
nteg
the j
uo /
Deity
velo
flow
ıtal
rizor
f hoi
ict o
Effe

Horizontal	Oil flow to	Separator	Residence	Oil	Effluent oil	Separator	Oil flow to	Oxygen	Bioreactor
velocity	separator	volume	time	recovery	concentration	efficiency	bioreactor	demand	volume
(m/s)	(m³/h)	(m³)	(min)	(m³/h)	(I/gm)	(%)	(m³/h)	(g/I.h)	(m³)
0.0049050	0.0125	3.888	7.9	0.0036	602	29.1	0.00886	2.46	113.1
0.0024525	0.0125	3.888	15.8	0.0072	424	57.6	0.00529	1.47	67.6
0.0019620	0.0125	3.888	19.75	0.0087	303	69.7	0.00378	1.05	48.3
0.0014715	0.0125	3.888	26.33	0.0104	168	83.3	0.00209	0.58	26.7
0.0009810	0.0125	3.888	39.5	0.0118	56	94.4	0.00070	0.19	8.9
0.0004905	0.0125	3.888	78.99	0.0124	4	9.66	0.00005	0.01	0.7

Horizontal	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
velocity	separator	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(m/s)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
0.0049050	44.56	9.03	35.53	561.84	4591.89	5153.72	5189.25
0.0024525	44.56	17.87	26.69	300.62	2743.91	3044.52	3071.21
0.0019620	44.56	21.61	22.95	197.09	1962.52	2159.61	2182.56
0.0014715	44.56	25.81	18.75	138.93	1085.04	1223.98	1242.73
0.0009810	44.56	29.27	15.29	72.79	362.73	435.51	450.81
0.0004905	44.56	30.87	13.69	15.97	27.73	43.70	57.39

Effect of separator volume on the integrated process of horizontal flow separator and ICB: Input parameters

Influent characteristics		Separator dimensions	
Water flowrate	10 m³/h 1000 ho/m³	Width Loicht	1.8288 m 0.0111 m
water density Water viscosity	0.001 Pa.s	Length	VARIABLE m
Oil concentration Average oil density	1000 mg/l 800 kg/m³		
Average droplet size	60 µm		
Standard deviation	15 µm		
Basis of costing			
Cost of oil Eauipment cost model	0.62 £/m³ CS <sup>n</sup>		
Daily hours of operation Annual days of operation Period of operation	16 h/day 250 days/yr 25 yr		

Effect of separator volume on the integrated process of horizontal flow separator and ICB: Results

Separator	Separator	Oil flow to	Residence	Oil	Effluent oil	Separator	Oil flow to	Oxygen	Bioreactor
volume	length	separator	time	recovery	concentration	efficiency	bioreactor	demand	volume
(m³)	(m)	(m²/h)	(min)	(m³/h)	(mg/l)	(%)	(m³/h)	(g/I.h)	(m³)
3.888	2.3248	0.0125	7.9	0.0036	602	29.1	0.00886	2.46	113.1
7.851	4.6950	0.0125	16.0	0.0073	419	58.1	0.00523	1.45	66.8
11.663	6.9743	0.0125	23.7	0.0098	212	78.8	0.00265	0.74	33.9
15.550	9.2991	0.0125	31.6	0.0112	103	89.8	0.00128	0.36	16.3
19.438	11.6239	0.0125	39.5	0.0118	56	94.4	0.00070	0.19	8.9
23.326	13.9486	0.0125	47.4	0.0122	25	97.5	0.00032	0.09	4.1
27.213	16.2734	0.0125	55.3	0.0123	18	98.2	0.00023	0.06	2.9
31.101	18.5982	0.0125	63.2	0.0124	<del>, -</del>	98.9	0.00013	0.04	1.7
34.988	20.9229	0.0125	71.1	0.0124	S	99.5	0.00006	0.02	0.8
38.876	23.2477	0.0125	79.0	0.0124	4	9.66	0.00005	0.01	0.7
46.652	27.8974	0.0125	94.8	0.0125	ю	99.7	0.00003	0.01	0.4

Effect of separator volume on the integrated process of horizontal flow separator and ICB: Results

Separator	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
volume	separator	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(m³)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
3.888	44.56	9.03	35.53	561.84	4591.89	5153.72	5189.25
7.851	67.46	18.02	49.44	298.55	2712.53	3011.08	3060.52
11.663	85.20	24.42	60.78	159.78	1375.18	1534.96	1595.75
15.550	100.96	27.82	73.14	103.99	664.02	768.01	841.15
19.438	115.17	29.27	85.90	72.79	362.73	435.51	521.42
23.326	128.25	30.21	98.04	45.70	164.83	210.53	308.57
27.213	140.46	30.44	110.02	37.39	117.27	154.65	264.67
31.101	151.97	30.67	121.30	27.51	69.71	97.22	218.52
34.988	162.91	30.84	132.07	17.72	33.09	50.82	182.88
38.876	173.36	30.87	142.49	15.97	27.73	43.70	186.19
46.652	193.05	30.92	162.13	11.97	17.00	28.97	191.09

Effect of average droplet size on the integrated process of horizontal flow separator and ICB: Input parameters

Influent characteristics		Separator dimensions	
Water flowrate	10 m³/h	Width	1.8288 m
Water density	1000 kg/m³	Height	0.9144 m
Water viscosity	0.001 Pa.s	Length	2.3248 m
Oil concentration Average oil density	1000 mg/l 800 ka/m³		
Average droplet size			
Standard deviation	15 µm		
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

<u>Results</u>
ICB:
or and
eparat
flow s
izontal
of hor
rocess
rated r
e integ
e on th
olet siz
se drot
averag
ffect of
ш

Average	Oil flow to	Separator	Residence	lio	Effluent oil	Separator	Oil flow to	Oxygen	Bioreactor
droplet size	separator	volume	time	recovery	concentration	efficiency	bioreactor	demand	volume
(mŋ)	(m³/h)	(m³)	(min)	(m³/h)	(I/gm)	(%)	(m³/h)	(g/I.h)	(m³)
60	0.0125	3.888	7.9	0.0036	602	29.1	0.00886	2.46	113.1
70	0.0125	3.888	7.9	0.0046	634	36.6	0.00792	2.20	101.2
80	0.0125	3.888	7.9	0.0056	549	45.1	0.00686	1.90	87.5
06	0.0125	3.888	7.9	0.0068	453	54.7	0.00566	1.57	72.2
100	0.0125	3.888	7.9	0.0081	348	65.2	0.00435	1.21	55.5
150	0.0125	3.888	7.9	0.0124	7	99.3	0.00009	0.02	1.1

Average	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
droplet size	separator	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(mµ)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
60	44.56	9,03	35.53	561.84	4591.89	5153.72	5189.25
70	44.56	11.34	33.22	522.65	4108.49	4631.14	4664.36
80	44.56	13.99	30.57	351.16	3555.08	3906.23	3936.81
06	44.56	16.96	27.60	312.88	2933.02	3245.90	3273.50
100	44.56	20.21	24.35	267.20	2254.62	2521.82	2546.17
150	44.56	30.78	13.78	21.46	45.78	67.25	81.03
# Effect of viscosity on the integrated process of horizontal flow separator and ICB: Input parameters

Influent characteristics		Separator dimensions	
Water flowrate Water density Water viscosity	10 m³/h 1000 kg/m³ VARIABLE Pa.s	Width Height Length	1.8288 m 0.9144 m 2.3248 m
Oil concentration Average oil density	1000 mg/l 800 kg/m³		
Average droplet size Standard deviation	60 µm 15 µm		
Basis of costing			
Cost of oil Equipment cost model	0.62 <i>£/</i> m³ CS <sup>n</sup>		
Daily hours of operation Annual days of operation Period of operation	16 h/day 250 days/yr 25 yr		

•

: Results
r and ICB
v separato
zontal flov
ess of hori
ated proce
the integra
cosity on 1
ffect of vis
Щ

Bioreactor	volume	(m³)	113.1	88.4	62.0	39.2	23.4
Oxygen	demand	(d/l.h)	2.46	1.92	1.35	0.85	0.51
Oil flow to	bioreactor	(m³/h)	0.00886	0.00692	0.00485	0.00307	0.00183
Separator	efficiency	(%)	29.1	44.6	61.1	75.5	85.3
Effluent oil	concentration	(mg/l)	602	554	389	246	147
Oil	recovery	(m³/h)	0.0036	0.0056	0.0076	0.0094	0.0107
Residence	time	(min)	6.7	7.9	7.9	7.9	7.9
Separator	volume	(m³)	3.888	3.888	3.888	3.888	3.888
Oil flow to	separator	(m³/h)	0.0125	0.0125	0.0125	0.0125	0.0125
Water	viscosity	(Pa.s)	0.00100	0.00065	0.00047	0.00036	0.00028

Water	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
viscosity	separator	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(Pa.s)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
0.00100	44.56	9.03	35.53	561.84	4591.89	5153.72	5189.25
0.00065	44.56	13.82	30.74	353.26	3590.63	3943.89	3974.63
0.00047	44.56	18.96	25.60	285.44	2516.91	2802.34	2827.95
0.00036	44.56	23.39	21.17	174.08	1590.25	1764.33	1785.50
0.00028	44.56	26.45	18.11	128.51	950.67	1079.17	1097.28

Effect of density difference on the integrated process of horizontal flow separator and ICB: Input parameters

Influent characteristics		Separator dimensions	
Water flowrate	10 m³/h	Width	1.8288 m
Water density	1000 kg/m³	Height	0.9144 m
Water viscosity	0.001 Pa.s	Length	2.3248 m
Oil concentration	1000 mg/l		
Average oil density	800 kg/m³		
Average droplet size	60 µm		
Standard deviation	15 µm		
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

Effect of density difference on the integrated process of horizontal flow separator and ICB: Results

Density	Oil flow to	Separator	Residence	Oil	Effluent oil	Separator	Oil flow to	Oxygen	Bioreactor
difference	separator	volume	time	recovery	concentration	efficiency	bioreactor	demand	volume
(kg/m³)	(m³/h)	(m³)	(min)	(m³/h)	(mg/l)	(%)	(m³/h)	(g/I.h)	(m³)
200	0.0125	3.888	7.9	0.0036	602	29	0.00886	2.46	113.1
225	0.0125	3.888	7.9	0.0041	672	33	0.00840	2.33	107.2
235	0.0125	3.888	7.9	0.0043	658	34	0.00822	2.28	104.9

						and the state is shown into	
Density	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
difference	separator	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(kg/m³)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
200	44.56	9.03	35.53	561.84	4591.89	5153.72	5189.25
225	44.56	10.16	34.40	542.88	4355.69	4898.57	4932.96
235	44.56	10.61	33.95	535.20	4261.27	4796.47	4830.42

### Analysing Stokes' law

### $v = d^2 (p_w - \rho_o) g / 18 \mu$

- v = droplet rise velocity (m/s)
- d = particle diameter (μm)
- g = acceleration due to gravity (m/s<sup>2</sup>)
- $\rho_w$  = density of water (kg/m<sup>3</sup>)
- $\rho_o = density of oil (kg/m^3)$
- μ = viscosity of water (Pa.s)

### Effect of increasing droplet size on rise velocity

	Increase in	Droplet rise	Increase in
Droplet size	droplet size	velocity	rise velocity
(μm)	(%)	(m/s)	(%)
20	0	4.360E-06	0
22	10	5.276E-06	21
24	20	6.278E-06	44
26	30	7.368E-06	69
28	40	8.546E-06	96
30	50	9.810E-06	125
32	60	1.116E-05	156
34	70	1.260E-05	189
36	80	1.413E-05	224
38	90	1.574E-05	261
40	100	1.744E-05	300
42	110	1.923E-05	341
44	120	2.110E-05	384

### Effect of increasing density difference on droplet rise velocity

	Increase in		
Density	density	Droplet rise	Increase in
difference	difference	velocity	rise velocity
(kg/m³)	(%)	(m/s)	(%)
200	0	3.924E-05	0
220	10	4.316E-05	10
230	15	4.513E-05	15
247	23	4.840E-05	23
262	31	5.134E-05	31
277	38	5.428E-05	38
292	46	5.723E-05	46
307	53	6.017E-05	53
322	61	6.311E-05	61
337	68	6.605E-05	68
352	76	6.900E-05	76
367	83	7.194E-05	83
382	91	7.488E-05	91

### Effect of decreasing viscosity on droplet rise velocity

	Decrease in		
Water	water	Droplet rise	Increase in
viscosity	viscosity	velocity	rise velocity
(cP)	(%)	(m/s)	(%)
0.13	0	3.018E-06	0
0.12	8	3.270E-06	8
0.11	15	3.567E-06	18
0.1	23	3.924E-06	30
0.09	31	4.360E-06	44
0.08	38	4.905E-06	63
0.07	46	5.606E-06	86
0.06	54	6.540E-06	117
0.05	62	7.848E-06	160
0.04	69	9.810E-06	225
0.03	77	1.308E-05	333
0.02	85	1.962E-05	550
0.01	92	3.924E-05	1200

,

### Appendix IV Theoretical results: Integration of a CPI and an ICB

Input data and results for investigations of integrated processes using a Corrugated Plate Interceptor and an Immobilized Cell Bioreactor.

Influent oil characteristics for the CPI

Effect of plate width Effect of plate angle Effect of plate spacing Effect of plate length Effect of number of plates

### Influent oil characteristics for the CPI

### **Oily Wastewater oil droplet size distribution**

### Influent characteristics

Water flowrate	<b>10</b> m³/h
Oil concentration	1000 mg/l
Average oil density	800 kg/m <sup>3</sup>
Average droplet diameter	<b>60</b> μm
Standard deviation	<b>15</b> μm
Calculations	
Mass of oil in influent	10 ka/h

Volume of oil in influent 0.0125 m<sup>3</sup>/h

No. of droplets in influent

9.25E+10 - Solver variable

Droplet	Minimum	Maximum	Mean	Cumulative	Fraction of	Number of	Volume of
size	droplet			distribution	droplets in	droplets in	droplets in
interval	size	droplet size	droplet size	function	interval	interval	interval
	μm	μm	μm				m³
1	0	10	5	0.00043	0.00043	3.970E+07	2.6E-09
2	10	20	15	0.00383	0.00340	3.147E+08	5.56E-07
3	20	30	25	0.02275	0.01892	1.751E+09	1.43E-05
4	30	40	35	0.09121	0.06846	6.334E+09	0.000142
5	40	50	45	0.25249	0.16128	1.492E+10	0.000712
6	50	60	55	0.50000	0.24751	2.290E+10	0.001995
7	60	70	65	0.74751	0.24751	2.290E+10	0.003293
8	70	80	75	0.90879	0.16128	1.492E+10	0.003296
9	80	90	85	0.97725	0.06846	6.334E+09	0.002037
10	90	100	95	0.99617	0.01892	1.751E+09	0.000786
11	100	110	105	0.99957	0.00340	3.147E+08	0.000191
12	110	120	115	0.99997	0.00040	3.677E+07	2.93E-05
13	120	130	125	1.00000	0.00003	2.789E+06	2.85E-06
14	130	140	135	1.00000	0.00000	1.372E+05	1.77E-07
15	140	150	145	1.00000	0.00000	4.370E+03	6.98E-09
16	150	160	155	1.00000	0.00000	9.007E+01	1.76E-10
17	160	170	165	1.00000	0.00000	1.200E+00	2.82E-12
18	170	180	175	1.00000	0.00000	1.032E-02	2.9E-14
19	180	190	185	1.00000	0.00000	0.000E+00	0
20	190	200	195	1.00000	0.00000	0.000E+00	0
21	200	210	205	1.00000	0.00000	0.000E+00	0
22	210	220	215	1.00000	0.00000	0.000E+00	0
23	220	230	225	1.00000	0.00000	0.000E+00	0
24	230	240	235	1.00000	0.00000	0.000E+00	0
25	240	250	245	1.00000	0.00000	0.000E+00	0
26	250	260	255	1.00000	0.00000	0.000E+00	0
27	260	270	265	1.00000	0.00000	0.000E+00	0
Totals					1.00000	9.252E+10	0.012499

Difference between calculated oil volumes

1E-06 - use Solver to make it zero.



Oil droplet distribution

### Effect of plate width on the integrated process of CPI and ICB: Input parameters

Influent characteristics		Interceptor dimensions	
Water flowrate	10 m³/h	Width	VARIABLE m
Water density	1000 kg/m³	Height	1.32 m
Water viscosity	0.001 Pa.s	Length	1.08 m
Oil concentration	1000 mg/l	Number of plates	S
Average oil density	800 kg/m³	Plate width	VARIABLE m
Average droplet size	60 µm	Plate length	E T
Standard deviation	15 µm	Plate spacing	0.02 m
		Plate angle to horizontal	° 09
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

25 yr

and ICB: Results
I process of CPI
n the integrated
olate width o
Effect of 1

	ciency	(%)	52.1	85.8	95.7	98.3	99.5	
it oil   Inte	ation eff	()						
Effluer	concentr	(mg/	479	142	43	17	5	
ō	recovery	(m <sup>s</sup> /h)	0.0065	0.0107	0.0120	0.0123	0.0124	
Reynolds	Number		556	278	185	139	111	
Velocity	through plates	(m/s)	0.0278	0.0139	0.0093	0.0069	0.0056	
Interceptor	volume	(m³)	2.144	3.573	5.002	6.431	7.860	
Effective	plate area	(m²)	3.0	5.0	8.0	10.0	13.0	
Total	plate area	(m²)	5.0	10.0	15.0	20.0	25.0	
Interceptor	width	(ш)	1.5	2.5	3.5	4.5	5.5	
Oil flow to	interceptor	(m²/h)	0.01250	0.01250	0.01250	0.01250	0.01250	
Plate	width	(m)	ł	7	e	4	5	

Plate	Oil flow to	Oxygen	Bioreactor	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
width	bioreactor	demand	volume	interceptor	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(m)	(m³/h)	(d.l/)	(m³)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
1	0.00599	1.66	76.4	136.42	16.14	120.28	323.73	3104.54	3428.27	3548.55
7	0.00178	0.49	22.7	201.64	26.59	175.05	126.12	920.90	1047.02	1222.07
e	0.00053	0.15	6.8	254.81	29.68	225.13	61.86	275.36	337.22	562.35
4	0.00021	0.06	2.7	301.33	30.48	270.86	35.69	108.40	144.10	414.95
S	0.00006	0.02	0.8	343.45	30.84	312.61	17.75	33.18	50.93	363.55

### Effect of plate angle on the integrated process of CPI and ICB: Input parameters

Influent characteristics		Interceptor dimensions	
Water flowrate	10 m³/h	Width	1.50 m
Water density	1000 kg/m³	Height	VARIABLE m
Water viscosity	0.001 Pa.s	Length	VARIABLE m
Oil concentration	1000 mg/l	Number of plates	£
Average oil density	800 kg/m³	Plate width	1 m
Average droplet size	60 µm	Plate length	т Т
Standard deviation	15 µm	Plate spacing	0.02 m
		Plate angle to horizontal	VARIABLE °
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

.

c n	r.
ئٹ	
F	
- <del>2</del>	
- 65	
ž	
14	ŀ
$\infty$	
5	
$\mathcal{O}$	
H	
Ч	
- <del>2</del>	
- 22	
-	
ĿЧ.	Ĺ
$\circ$	
<i>t</i> .	Į.
Ĭ	
0	
Ś	
S	
O O	
్ల	ŀ
0	I.
- 12	l.
	Ļ
- X	1
<u>ل</u>	
55	
H	ŀ
્વ	•
୍ର	
7	ł
.=	
ല	
- 53	ŀ
_ <b>_</b>	
0	l
<u> </u>	
5	
- F	
3	
e	
H	
2	
<u>í</u>	
ž	
0	
بب	ŀ
ုပ္ပ	
.0	1
£	
(T)	ŀ

Plate	Oil flow to	Interceptor	Interceptor	Total	Effective	Interceptor	Velocity	ĪŌ	Effluent oil	Interceptor
angle	interceptor	length	height	plate area	plate area	volume	through plates	recovery	concentration	efficiency
(。)	(m³/h)	(m)	(m²)	(m²)	(m²)	(m³)	(m/s)	(m³/h)	(I/gm)	(%)
60	0.01250	1.08	1.32	5	ю	2.144	0.0278	0.0065	479	52.1
40	0.01250	1.61	1.10	5 L	4	2.660	0.0278	0.0092	266	73.4
30	0.01250	1.81	0.96	5	4	2.602	0.0278	0.0099	205	79.5
20	0.01250	1.96	0.80	2	5	2.349	0.0278	0.0103	172	82.8
10	0.01250	2.05	0.63	5	5	1.939	0.0278	0.0105	158	84.2

Plate	Oil flow to	Oxvaen	Bioreactor	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
angle	bioreactor	demand	volume	interceptor	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
	(m³/h)	(d.h)	(m³)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
60	0.00599	1.66	76.4	136.42	16.14	120.28	323.73	3104.54	3428.27	3548.55
40	0.00332	0.92	42.4	140.68	22.76	117.93	182.43	1721.55	1903.98	2021.90
30	0.00256	0.71	32.6	140.22	24.65	115.57	156.35	1325.43	1481.78	1597.35
20	0.00215	0.60	27.4	138.16	25.66	112.50	141.12	1114.11	1255.23	1367.73
10	0.00197	0.55	25.2	134.63	26.10	108.52	134.16	1022.56	1156.72	1265.24

### Effect of plate spacing on the integrated process of CPI and ICB: Input parameters

Influent characteristics		Interceptor dimensions	
Water flowrate	10 m³/h	Width	1.50 m
Water density	1000 kg/m <sup>3</sup>	Height	1.32 m
Water viscosity	0.001 Pa.s	Length	VARIABLE m
Oil concentration	1000 mg/l	Number of plates	ۍ ۲
Average oil density	800 kg/m³	Plate width	<del>г</del>
Average droplet size	60 µm	Plate length	E F
Standard deviation	15 µm	Plate spacing	VARIABLE m
		Plate angle to horizontal	° 09
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

lts	
esu	
 8.	
ICI	
and	
<u>P</u>	
ofC	
ess	
Droc	
ed I	
grat	
inte	
the	
on	
ing	
spac	
ates	
fpl	
ct o	
ffe	•
H-H	

Plate	Oil flow to	Interceptor	Total	Effective	Interceptor	Velocity	Reynolds	ΪÖ	Effluent oil	Interceptor
spacing	interceptor	length	plate area	plate area	volume	through plates	Number	recovery	concentration	efficiency
(m)	(m³/h)	(m)	(m²)	(m²)	(m³)	(m/s)		(m³/h)	(I/gm)	(%)
0.020	0.01250	1.08	5.0	3.0	2.144	0.0278	556	0.0065	479	52.1
0.025	0.01250	1.10	5.0	3.0	2.183	0.0222	556	0.0066	475	52.5
0.030	0.01250	1.12	5.0	3.0	2.223	0.0185	556	0.0066	471	52.9
0.040	0.01250	1.16	5.0	3.0	2.302	0.0139	556	0.0067	462	53.7
0.050	0.01250	1.20	5.0	3.0	2.382	0.0111	556	0.0068	454	54.6

Plate	Oil flow to	Oxygen	Bioreactor	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
spacing	bioreactor	demand	volume	interceptor	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(m)	(m³/h)	(d/l.h)	(m³)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
0.020	0.00599	1.66	76.4	136.42	16.14	120.28	323.73	3104.54	3428.27	3548.55
0.025	0.00593	1.65	75.8	136.76	16.27	120.49	322.02	3077.26	3399.28	3519.78
0.030	0.00588	1.63	75.1	137.10	16.40	120.70	320.31	3049.98	3370.29	3490.99
0.040	0.00578	1.60	73.7	137.77	16.66	121.11	316.86	2995.43	3312.29	3433.40
0.050	0.00567	1.58	72.4	138.43	16.92	121.51	313.38	2940.88	3254.26	3375.77

Effect of plate length on the integrated process of CPI and ICB: Input parameters

Influent characteristics		Interceptor dimensions	
Water flowrate	10 m³/h	Width	1.50 m
Water density	1000 kg/m³	Height	VARIABLE m
Water viscosity	0.001 Pa.s	Length	VARIABLE m
Oil concentration	1000 mg/l	Number of plates	S
Average oil density	800 kg/m³	Plate width	1 H
Average droplet size	60 µm	Plate length	VARIABLE m
Standard deviation	15 µm	Plate spacing	0.02 m
		Plate angle to horizontal	° 09
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

: Results
ICB
and
CPI
b
process
integrated
the
uo u
engt
plate
tof
Effect

Plate	Oil flow to	Interceptor	Interceptor	Total	Effective	Interceptor	Velocity	0 II	Effluent oil	Interceptor
length	interceptor	length	height	plate area	plate area	volume	through plates	recovery	concentration	efficiency
(m)	(m³/h)	(m)	(E)	(m²)	(m²)	(m³)	(m/s)	(m³/h)	(mg/l)	(%)
-	0.01250	1.08	1.32	5.0	3.0	2.144	0.0278	0.0065	479	52.1
ы	0.01250	2.08	2.19	10.0	5.0	6.830	0.0278	0.0106	148	85.2
ო	0.01250	3.08	3.06	15.0	8.0	14.115	0.0278	0.0119	47	95.3
4	0.01250	4.08	3.92	20.0	10.0	23.998	0.0278	0.0123	18	98.2
5	0.01250	5.08	4.79	25.0	13.0	36.479	0.0278	0.0124	5	99.5

Plate	Oil flow to	Oxygen	Bioreactor	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
length	bioreactor	demand	volume	interceptor	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
(m)	(m³/h)	(d.l/b)	(m³)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
Ŧ	0.00599	1.66	76.4	136.42	16.14	120.28	323.73	3104.54	3428.27	3548.55
7	0.00185	0.51	23.6	221.38	26.41	194.97	129.16	958.88	1088.04	1283.02
ę	0.00058	0.16	7.4	298.46	29.55	268.91	65.37	302.33	367.70	636.60
4	0.00023	90.0	2.9	371.78	30.44	341.35	37.33	116.99	154.32	495.67
S	0.00007	0.02	0.8	442.92	30.83	412.08	18.16	34.47	52.63	464.71

## Effect of number of plates on the integrated process of CPI and ICB: Input parameters

Influent characteristics		Interceptor dimensions	
Water flowrate	10 m³/h	Width	1.50 m
Water density	1000 kg/m³	Height	1.32 m
Water viscosity	0.001 Pa.s	Length	VARIABLE m
Oil concentration	1000 mg/l	Number of plates	VARIABLE
Average oil density	800 kg/m³	Plate width	÷
Average droplet size	60 µm	Plate length	1 E
Standard deviation	15 µm	Plate spacing	0.02 m
		Plate angle to horizontal	° 09
Basis of costing			
Cost of oil	0.62 £/m³		
Equipment cost model	CS <sup>n</sup>		
Daily hours of operation	16 h/day		
Annual days of operation	250 days/yr		
Period of operation	25 yr		

Effect of number of plates on the integrated process of CPI and ICB: Input parameters

t oil Interceptor	ation efficiency	(%)	52.1	85.8	95.7	98.3	99.5	
Effluent	concentra	(mg/)	479	142	43	17	5	
ō	recovery	(m³/h)	0.0065	0.0107	0.0120	0.0123	0.0124	
Reynolds	Number		556	278	185	139	111	
Velocity	through plates	(m/s)	0.0278	0.0139	0.0093	0.0069	0.0056	
Interceptor	volume	(m³)	2.144	2.342	2.541	2.739	2.938	
Effective	plate area	(m²)	3.0	5.0	8.0	10.0	13.0	
Total	piate area	(m²)	5.0	10.0	15.0	20.0	25.0	
Interceptor	length	(m)	1.08	1.18	1.28	1.38	1.48	
Oil flow to	interceptor	(u/²/n)	0.01250	0.01250	0.01250	0.01250	0.01250	
Number of	plates		S	10	15	20	25	

Number of	Oil flow to	Oxygen	Bioreactor	Cost of	Value of	Cost of	Cost of	Cost of	Cost of	Cost of
plates	bioreactor	demand	volume	interceptor	recovered oil	separation	bioreactor	electricity	bioprocess	int. process
	(m³/h)	(g/l.h)	(m³)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)	(£/yr)
5	0.00599	1.66	76.4	136.42	16.14	120.28	323.73	3104.54	3428.27	3548.55
10	0.00178	0.49	22.7	192.29	26.59	165.70	126.12	920.90	1047.02	1212.72
15	0.00053	0.15	6.8	237.77	29.68	208.09	61.86	275.36	337.22	545.32
20	0.00021	0.06	2.7	277.61	30.48	247.13	35.69	108.40	144.10	. 391.23
25	0.00006	0.02	0.8	313.72	30.84	282.88	17.75	33.18	50.93	333.81